

# SENATE JOURNAL

# STATE OF ILLINOIS

# NINETY-THIRD GENERAL ASSEMBLY

# 58TH LEGISLATIVE DAY

**Perfunctory Session** 

THURSDAY, OCTOBER 23, 2003

12:05 O'CLOCK P.M.

NO. 58

# SENATE Daily Journal Index 58th Legislative Day

	Action Introduction of Senate Bills No'd. 2090-2103 Joint Action Motion Filed Joint Action Motions Filed Message from the President			
	Motion in Writing			
Bill Number SR 0274	Legislative Action Committee on Rules	Page(s)		

The Senate met pursuant to adjournment. Senator Welch, presiding. Silent prayer was observed.

#### REPORTS RECEIVED

The Secretary placed before the Senate the following reports:

Report on Village of Sidney – Phase II, Flood Hazard Mitigation Project submitted by the Illinois Department of Natural Resources, Office of Water Resources pursuant to Flood Control Act of 1945.

Quarterly report on Adult and Juvenile Facilities submitted by the Illinois Department of Corrections.

Report on Bonded Indebtedness and Long Term Obligations submitted by the Office of the Comptroller.

2002 Annual Report submitted by the University of Illinois at Chicago.

Investigational Cancer Treatments Coverage Report submitted by the Illinois Department of Insurance pursuant to P.A. 91-0406.

2002 Annual Report submitted by the Illinois Building Commission.

Fifteenth Annual Toxic Chemical Report submitted by the Illinois Environmental Protection Agency.

Construction Progress Report for the period ending December 31, 2002 submitted by the Metropolitan Pier and Exposition Authority pursuant to the Metroplitan Pier and Exposition Authority Act as amended (70 ILCS 210/1).

Construction Progress Report for the period ending January 31, 2003 submitted by the Metropolitan Pier and Exposition Authority pursuant to the Metroplitan Pier and Exposition Authority Act as amended (70 ILCS 210/1).

Construction Progress Report for the period ending February 28, 2003 submitted by the Metropolitan Pier and Exposition Authority pursuant to the Metroplitan Pier and Exposition Authority Act as amended (70 ILCS 210/1).

2002 Annual Report submitted by the Illinois Labor Relations Board.

Statement of Receipts and Expenditures for July 1, 2001 through June 30, 2002 submitted by the Northeastern Illinois Planning Commission.

Annual Report on School Breakfast Incentives submitted pursuant to P.A. 91-0843 by The State Board of Education.

2002 Annual Report submitted by the Chicago Gary Regional Airport Authority.

Illinois Emergency Food and Shelter Program Report submitted by the Illinois Department of Human Services.

2002 Report on Illinois Child Care submitted by the Illinois Department of Human Services.

Financial Statements Report for the year ending December 31, 2002 submitted by the Illinois Thoroughbred Breeders and Owners Foundation.

Financial Statements submitted by the Metropolitan Pier and Exposition Authority pursuant to 70 ILCS 210/1.

Annual Report of the Office of Consumer Health Insurance submitted pursuant to 505ILCS 19/20 by the Illinois Department of Insurance.

2002 Executive Summary Report submitted by the Office of the Comptroller.

FY 2004 Legislative Capital Plan Analysis Report submitted by the Illinois Economic and Fiscal Commission.

Report on the Rock Island County Phase II Flood Hazard Mitigation Project submitted by the Illinois Department of Natural Resources pursuant to the Flood Control Act of 1945.

Traditional Budgetary Financial Report submitted by the Office of the Comptroller.

2002 Annual Report on the Status of Transition Services for Secondary Students with Disabilities in Illinois submitted by the Interagency Coordinating Council.

2002 Annual Report for Centers for Independent Living submitted by the Illinois Department of Human Services.

Report on Emerging Investment Managers for Fiscal Year 2003 submitted by the Illinois State Board of Investment under the Illinois Pension Code amended and adopted effective January 25, 1993.

Fiscal Year 2003 Report on Before and After School Program in Illinois Public Schools submitted by the Illinois State Board of Education as required by Section 2.311b of the School Code..

2002 Annual Report and Financial Summaries submitted by the Illinois Comprehensive Health Insurance Plan.

2002 Annual Report submitted by the Quad Cities Regional Economic Development.

2001 Annual Report submitted by the State Rehabilitation Council of Illinois.

Financial, Compliance, and Program Audit for the Village of Robbins' Use of Municipal Economic Development Funds submitted by the Illinois Office of the Auditor General pursuant to Public Act 90-813.

Financial Statements submitted by the Metropolitan Pier and Exposition Authority for the twelve months ended June 30, 2003 pursuant to the Metropolitan Pier and Exposition Authority Act as amended (70 ILCS 210/1).

Fiscal Year 2004 Economic and Revenue Estimate submitted by the Illinois Economic and Fiscal Commission.

Third Quarterly Procurement Report regarding Minority and Women Owned Business Participation submitted by the Metropolitan Pier and Exposition Authority pursuant to the Metropolitan Pier and Expostion Authority Act as amended (70 ILCS 210/1).

2003 Report on the Use of Emerging Investment Managers submitted by the State Universities Retirement System of Illinois as required by Public Act 87-1265.

Annual Report for Fiscal Year ending June 30, 2003 submitted by Illinois Department of Transportation in accordance with Section 5.201.16 of the Illinois Highway Code.

2001 – 2002 Biennial Report submitted by the Illinois Community College Board.

Construction Progress Report for period ending March 2003 submitted by the Metropolitan Pier and Exposition Authority pursuant to the Metropolitan Pier and Exposition Authority Act as amended (70 ILCS 210/1).

#### COMMUNICATIONS

# OFFICE OF THE SECRETARY OF STATE JESSE WHITE – Secretary of State

#### CERTIFICATION

RE: Senate Bill 0015 of the 93<sup>rd</sup> General Assembly

Pursuant to 15 ILCS 305/10, this bill having remained with the Governor 60 calendar days after it was presented to him, and the Governor having failed to return this bill to the General Assembly during its session, and having failed to file it in my office, with his objections, within such 60 calendar days, it has thereby become law.

Dated: August 11, 2003

Signature: JESSE WHITE, Secretary of State

# OFFICE OF THE SECRETARY OF STATE JESSE WHITE – Secretary of State

### CERTIFICATION

RE: Senate Bill 0064 of the 93rd General Assembly

Pursuant to 15 ILCS 305/10, this bill having remained with the Governor 60 calendar days after it was presented to him, and the Governor having failed to return this bill to the General Assembly during its session, and having failed to file it in my office, with his objections, within such 60 calendar days, it has thereby become law.

Dated: August 11, 2003

Signature: JESSE WHITE, Secretary of State

# OFFICE OF THE SECRETARY OF STATE **JESSE WHITE** – Secretary of State

# CERTIFICATION

RE: Senate Bill 0078 of the 93rd General Assembly

Pursuant to 15 ILCS 305/10, this bill having remained with the Governor 60 calendar days after it was presented to him, and the Governor having failed to return this bill to the General Assembly during its session, and having failed to file it in my office, with his objections, within such 60 calendar days, it has thereby become law.

Dated: August 11, 2003

Signature: JESSE WHITE, Secretary of State

# OFFICE OF THE SECRETARY OF STATE **JESSE WHITE** – Secretary of State

#### CERTIFICATION

RE: Senate Bill 0265 of the 93<sup>rd</sup> General Assembly

Pursuant to 15 ILCS 305/10, this bill having remained with the Governor 60 calendar days after it was presented to him, and the Governor having failed to return this bill to the General Assembly during its session, and having failed to file it in my office, with his objections, within such 60 calendar days, it has thereby become law.

Dated: August 11, 2003

Signature: JESSE WHITE, Secretary of State

# OFFICE OF THE SECRETARY OF STATE **JESSE WHITE** – Secretary of State

#### CERTIFICATION

RE: Senate Bill 0884 of the 93rd General Assembly

Pursuant to 15 ILCS 305/10, this bill having remained with the Governor 60 calendar days after it was presented to him, and the Governor having failed to return this bill to the General Assembly during its session, and having failed to file it in my office, with his objections, within such 60 calendar days, it has thereby become law.

Dated: August 11, 2003

Signature: JESSE WHITE, Secretary of State

#### REPORTS RECEIVED

#### ILLINOIS STATE BOARD OF EDUCATION

#### MEMORANDUM

**TO:** The Honorable Emil Jones, Senate President

The Honorable Frank C. Watson, Senate Minority Leader The Honorable Michael J. Madigan, Speaker of the House The Honorable Tom Cross, House Minority Leader

FROM: Robert E. Schiller

State Superintendent of Education

DATE: September 30, 2003

RE: Waivers of School Code Mandates: Fall 2003 Summary Report

As required by Section 2-3.25g of the School Code (105 ILCS 5/2-3.25g), the following report provides summaries of requests for waivers of School Code mandates being transmitted to the Illinois General Assembly for its consideration. Also included are summaries of requests for waivers and modifications acted on by the State Board of Education and of applications that have been returned to school districts or other eligible applicants.

If you have any questions or comments, please contact Respicio F. Vazquez, General Counsel, at 217/782-8535.

cc: The Honorable Rod R. Blagojevich, Governor Brad Bolin, Assistant Clerk of the House Linda Hawker, Secretary of the Senate Legislative Research Unit State Government Report Center

# **Executive Summary**

The following report outlines waivers of School Code mandates that school districts, regional offices of education, or special education or vocational education cooperatives have requested since the last report, which was transmitted in May 2003. Pursuant to Section 2-3.25g of the School Code (105 ILCS 5/2-3.25g), these requests must be sent to the General Assembly for its consideration before October 1, 2003.

The report is organized by subject area and by school district, regional office, or special education or vocational education cooperative. The General Assembly may disapprove the

report in whole or in part within 30 calendar days after each chamber next convenes once the report is filed. This is done by a joint resolution. If either chamber fails to reject a waiver request, then that request is deemed granted.

Section I summarizes the 21 requests received for waivers of School Code mandates for consideration by the General Assembly, which are presented alphabetically by topic area. The largest number of applications received seeks waivers from Section 27-23 of the School Code regarding driver education (seven requests), followed by three petitions each pertaining to content of evaluation plans and physical education. Two petitions each were received for non-resident tuition, parent-teacher conferences, and school improvement/inservice training, and one petition each was received for limitation of administrative costs and substitute teachers.

The request dealing with substitute teacher certificates seeks authorization to employ substitutes for longer than the 90 days allowed under Section 21-9 of the School Code (see page 4 of the report). Since the petition submitted does not propose to restrict substitutes who also hold a teaching certificate to 120 days of employment nor limit the waiver request to the next school year (both of which are requirements imposed by P.A. 92-184, effective July 27, 2001), the State Board of Education must continue to send it to the General Assembly for action.

Additionally, the General Assembly will be asked to consider an appeal of a State Board of Education decision to deny a request wishing to waive regulatory requirements pertaining to rules governing reimbursement of transportation costs. The appeal is summarized in Section II.

This document also contains three other sections beyond what is required under Section 2-3.25g of the School Code. Section III lists the modifications or waivers of State Board of Education rules and modifications of School Code mandates upon which the State Board has acted. The State Board denied one and approved 136. Of the approvals, 125 address legal school holidays; five involve physical education and block scheduling; three address instructional time and the Prairie State Achievement Examination; and one each addresses course requirements for bilingual education, driver education, and substitute teachers.

Section IV describes the 27 requests that have been returned to or withdrawn by the petitioning entities. Section V shows all the requests submitted, organized by Senate and House district.

In addition, the requests received are summarized by subject area in a table following this Executive Summary. Complete copies of the waiver requests for the General Assembly's consideration have been made available to legislative staff.

This is the seventeenth report submitted pursuant to Section 2-3.25g of the School Code, which requires that the State Board of Education compile and submit requests for waivers of School Code mandates to the General Assembly before May 1 and October 1 of each year.

# Summary of Applications for Waivers and Modifications Volume 17 – Fall 2003

Topic	Approved	Denied by I SBE	Transmitted to GA	Withdrawn or Returned	
Content of Evaluation I	Plans 0	0	3	0	
Course Requirements	1	0	0	0	
Driver Education	1	0	7	1	
Legal School Holidays	125	0	0	23	

Limitation of Administrati Costs	ive 0	0	1	0	
Non-Resident Tuition	0	0	2	0	
Parent-Teacher Confere	nces 0	0	2	0	
Physical Education	5	0	3	1	
PSAE – Instructional Tim	ne 3	0	0	1	
School Improvement/ Inservice Training	0	0	2	1	
Substitute Teachers	1	0	1	0	
Transportation Reimburs	1	0	0		
Transportation Reimburs (Appeal)	sement 0	0	1	0	

TOTAL NUMBER OF APPLICATIONS:

136

### SECTION I

1

22

186

27

### **Applications Transmitted to the General Assembly**

# **Content of Evaluation Plans**

**Petition Summary** 

Galesburg CUSD 205 - Knox (SD 37/HD 74) / Expiration: 2007-08 school year

**WM100-2886 (renewal) – Waiver of School Code** (Section 24A-5) request to allow the district to replace the current rating scale of "excellent, satisfactory, and unsatisfactory" for the teacher evaluation process with a professional growth plan that incorporates a professional assistance component that allows for remediation of teachers with unsatisfactory performance, as required under Section 24A-5.

Deer Creek-Mackinaw CUSD 701 - Tazewell (SD 44/HD 87) / Expiration: 2008-09 school year

**WM100-3012 – Waiver of School Code** (Section 24A-5) request to allow the district to replace the current rating scale of "excellent, satisfactory, and unsatisfactory" for the teacher evaluation process with a professional growth plan that incorporates a professional assistance component that allows for remediation of teachers with unsatisfactory performance, as required under Section 24A-5. If approved, this waiver would take effect in the 2004-05 school year.

Evanston CCSD 65 - Cook (SD 9/HD 18) / Expiration: 2008-09 school year

**WM100-3013 (renewal) – Waiver of School Code** (Section 24A-5) request to allow the district to replace the current rating scale of "excellent, satisfactory, and unsatisfactory" for the teacher evaluation process with "meets district standards of excellence, needs to improve to meet district standards, and unsatisfactory." If approved, this waiver would take effect in the 2004-05 school year.

#### **Driver Education**

Lake Zurich SD 95 – Lake (SD 26/HD 51) / Expiration: 2007-08 school year WM100-2968 (renewal) – Waiver of School Code (Section 27-23) request to allow the district [October 23, 2003]

to charge a reasonable fee not to exceed \$300 of students who participate in driver education courses.

Elmhurst CUSD 205 – DuPage (SD 21/HD 41) / Expiration: 2007-08 school year WM100-2980 (renewal) – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$200 of students who participate in driver education courses.

Antioch CHSD 117 – Lake (SD 31/HD 61) / Expiration: 2008-09 school year WM100-2997-2 (renewal) – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$300 of students who participate in driver education courses. If approved, this waiver takes effect in the 2004-05 school year.

Oak Lawn CHSD 218 – Cook (SD 18/HD 36) / Expiration: 2008-09 school year WM100-3015 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$250 of students who participate in driver education courses. If approved, this waiver takes effect in the 2004-05 school year.

Hononegah CHSD 207 – Winnebago (SD 34/HD 68) / Expiration: 2008-09 school year WM100-3019 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$200 of students who participate in driver education courses. If approved, this waiver takes effect in the 2004-05 school year.

Harvard CUSD 50 – McHenry (SD 32/HD 63) / Expiration: 2007-08 school year WM100-3026 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$200 of students who participate in driver education courses.

Oak Park-River Forest HSD 200 – Cook (SD 39/HD 78) / Expiration: 2007-08 school year WM100-3041 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$350 of students who participate in driver education courses.

### Limitation of Administrative Cost

Zion-Benton THSD 126 – Lake (SD 31/HD 61) / Expiration: 2003-04 school year WM100-3028-2 – Waiver of School Code (Section 17-1.5) request to allow the district to waive the limitation of administrative costs due to circumstances beyond its control. For the last two school years, the district has employed an interim superintendent. A permanent superintendent has been hired for the 2003-04 school year with a salary and benefit package that is higher than that paid to the interim superintendent, causing the district to exceed the 5 percent limitation.

# **Non-Resident Tuition**

Century CUSD 100 – Pulaski (SD 59/HD 118) / Expiration: 2008-09 school year WM100-2961 – Waiver of School Code (Section 10-20.12a) request to allow the district to charge non-resident pupil tuition in an amount that is less than 100 percent of the preceding year's per capita tuition costs to those students whose parents are employees of the district. If approved, this waiver would take effect in the 2004-05 school year.

Rankin SD 98 – Tazewell (SD 46/HD 91) / Expiration: 2008-09 school year WM100-3036-1 (renewal) – Waiver of School Code (Section 10-20.12a) request to allow the district to charge non-resident pupil tuition in an amount that is less than 100 percent of the preceding year's per capita tuition costs to students who wish to enroll in the district. The district maintains that a lower tuition rate will help it attract additional students necessary to offer a full curriculum without combining grade levels. If approved, this waiver would take effect in the 2004-05 school year.

# Parent-Teacher Conferences

South Beloit CUSD 320 – Winnebago (SD 34/HD 68) / Expiration: 2007-08 school year WM100-2924 – Waiver of School Code (Section 18-8.05(F)(2)(d)(1)) request to allow the district the option of scheduling twice during the school year two, four-hour parent-teacher conferences in the evening following a full day of student attendance. The four evening sessions will be counted as two of the 176 days of pupil attendance required by Section 10-19.

Roselle SD 12 – DuPage (SD 23/HD 45) / Expiration: 2008-09 school year WM100-3007-2 – Waiver of School Code (Section 18-8.05(F)(2)(d)(1)) request to allow the district to schedule the equivalent of two and a half days of parent-teacher conferences in a single school year. The district plans a full day of parent-teacher conferences (from 1 to 8:30)

single school year. The district plans a full day of parent-teacher conferences (from 1 to 8:30 p.m.) combined with a half day of inservice training in the fall; a half day of conferences to be held from 8 a.m. until noon in the fall; and a full day of conferences to be held from 1 to 8:30 p.m. in the spring. Students will not be in attendance on these days, and the district does not intend to count these days among the 176 days of pupil attendance required by Section 10-19. If approved, this waiver would take effect in the 2004-05 school year.

# Physical Education

Wauconda CUSD 118 - Lake (SD 30/HD 59) / Expiration: 2008-09 school year

**WM100-3011-3** (renewal) – Waiver of School Code (Section 27-6) request to allow the district to excuse students in grades 9 through 12 from daily physical education for ongoing participation in cheerleading and pom-pons. This request will allow these students more time during the school day to take additional academic courses. If approved, this waiver would take effect in the 2004-05 school year.

Zion-Benton THSD 126 - Lake (SD 31/HD 61) / Expiration: 2008-09 school year

**WM100-3028-1 (renewal) – Waiver of School Code** (Section 27-6) request to allow the district to excuse approximately 80 ninth-grade students who enroll in the "school-within-aschool Technology Academy" from the daily physical education requirement in order to take additional coursework required by the program and other elective courses that otherwise would not be available. Students could choose to take physical education as an elective course. If approved, this waiver takes effect in the 2004-05 school year.

Grayslake CHSD 127 - Lake (SD 31/HD 62) / Expiration: 2008-09 school year

**WM100-3038 (renewal) – Waiver of School Code** (Section 27-6) request to allow the district to excuse students in grades 11 and 12 from daily physical education for ongoing participation in cheerleading and pom-pons if they are enrolled in six academic classes. If approved, this waiver would take effect in the 2004-05 school year.

#### School Improvement/Inservice Training

Dwight THSD 230 – Livingston (SD 53/HD 105) / Expiration: 2005-06 school year WM100-2901-1 (renewal) – Waiver of School Code (Section 18-8.05(F)(2)(d)(2)) request to allow the district to dismiss students at 11:30 a.m. for three days the first week of the school year for the purpose of holding school improvement activities, and to accumulate sufficient

student attendance time beyond the five-clock-hour requirement within the next six school days to apply towards these days.

Kaneland CUSD 302- Kane, DeKalb (SD 25/HD 50) / Expiration: 2008-09 school year WM100-2987 (renewal) — Waiver of School Code (Section 18-8.05(F)(2)(d)(2)) request to allow the district to hold two full-day teacher inservice sessions instead of four half days, and to count the days among the 176 days of pupil attendance required by Section 10-19. The district will accumulate sufficient time beyond the five-clock-hour requirement to apply towards these days. If approved, this waiver would take effect in the 2004-2005 school year.

#### Substitute Teachers

Oak Lawn CHSD 218 – Cook (SD 18/HD 36) / Expiration: 2008-09 school year WM100-3005 (renewal) – Waiver of School Code (Section 21-9) request to allow the district

to employ substitute teachers for more than 90 days in any one school year. The most-qualified substitutes will be employed; each must meet the approval of the building principal in order to continue his or her employment. If approved, this waiver takes effect in the 2004-05 school year.

### SECTION II

# Appeals of Applications Denied by the Illinois State Board of Education

#### Transportation Reimbursement

Oak Park ESD 97 - Cook (SD 4/HD 7) / Expiration: 2007-08 school year

**WM100-3032** (appeal) – Waiver of Administrative Rules (23 III. Adm. Code 120.90(e)). The district is appealing the State Board of Education's denial of its petition requesting to be allowed to treat separately for the purposes of calculating its reimbursement for pupil transportation the category rather than the contractor since the district employs only one contractor to provide both regular and special education transportation. The district states that using the proration mechanism provided in the rules will reduce the district's annual transportation reimbursement from the state by more than \$60,000.

The rule that is the subject of this waiver is similar to the regulatory requirement that districts that own and operate their own transportation services prorate total transportation costs across all categories of transportation services, based on the ratio of miles traveled in each category to total system miles (Section 120.90(d) of the rules). The same proration formula is used for a school district that chooses one contractor to provide all of its transportation services.

The Oak Park board is using the waiver process to request that the programs and claims for special education (which is reimbursed at a higher rate) and regular education services remain separate as if there were two contractors. While there are two separate <u>contracts</u>. State Board staff believe that the district did not contract with a "company that provides only one type of transportation service;" rather, there is <u>only one contractor</u> for all transportation services. Therefore, that contractor should prorate costs across all transportation services provided under all contracts held with the school district.

The State Board of Education denied this request because the district failed to show that its request would meet the intent of the mandate (i.e., cost containment) in a more effective, efficient or economical manner. While the district may receive a larger payment from the state if this request is approved, the intent of the rule has not been addressed. Furthermore, approval of this request could create a cost incentive for other districts to contract for transportation services unrelated to any real efficiencies in managing such programs. Finally, when the state is unable to fully fund transportation claims, the additional cost of this waiver will spread to all other school districts.

## **SECTION III**

### Applications Acted on by the Illinois State Board of Education

# Applications Denied by the Illinois State Board of Education

# **Transportation Reimbursement**

Oak Park ESD 97 - Cook (SD 4/HD 7) / Expiration: 2007-08 school year

**WM300-2954 – Waiver of Administrative Rules** (23 III. Adm. Code 120.90(e)). The district requested to be allowed to treat separately for the purposes of calculating its reimbursement for pupil transportation the category rather than the contractor since the district employs only one contractor to provide both regular and special education transportation. The State Board of Education denied this request because the district failed to show that its request would meet the intent of the mandate in a more effective, efficient or economical manner.

# Applications Approved by the State Board of Education

### Course Requirements

Bensenville SD 2 – DuPage (SD 23/HD 46) / Expiration: 2003-04 school year WM300-2938 (renewal) – Modification of School Code (Sections 14C-2, 14C-2.1, and 14C-3) allows the district to offer an English immersion curriculum, with instruction in English, rather than a bilingual program that provides instruction in the student's home language.

#### Driver Education

Evergreen Park CHSD 231 – Cook (SD 18/HD 36) / Expiration: 2007-08 school year WM300-2874 – Waiver of Administrative Rule (23 III. Adm. Code 252.20(b)(1)) allows the district to provide behind-the-wheel instruction before and after school only, and on Saturdays, if additional sessions are needed.

# Holidays

Polo CUSD 222 – Ogle, Lee, Whiteside (SD 45/HD 90) / Expiration: 2007-08 school year WM300-2872 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Marquardt SD 15 – DuPage (SD 23/HD 45) / Expiration: 2007-08 school year WM300-2873 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Warren THSD 121 - Lake (SD 31/HD 62) / Expiration: 2007-08 school year

**WM300-2875 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, members of the armed forces killed in war, Columbus Day, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Cahokia CUSD 187 – St. Clair (SD 57/HD 114) / Expiration: 2003-04 school year WM300-2877 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Jasper County CUSD 1 – Jasper (SD 54/HD 108) / Expiration: 2007-08 school year WM300-2878 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Lick Creek SD 16 – Union (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2879 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Rutland CCSD 230 – LaSalle (SD 38/HD 76) / Expiration: 2007-08 school year WM300-2880 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Troy SD 30-C — Will (SD 42/HD 84) / Expiration: 2003-04 school year WM300-2881 (renewal) — Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than [October 23, 2003]

observing a school holiday in his honor.

Kinnikinnick CCSD 131 – Winnebago (SD 34/HD 68) / Expiration: 2007-08 school year WM300-2882 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Deland-Weldon CUSD 57 – Piatt (SD 55/HD 110) / Expiration: 2007-08 school year WM300-2884 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Wayne City CUSD 100 – Wayne (SD 54/HD 108) / Expiration: 2007-08 school year WM300-2885 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Armstrong-Ellis CSD 61 – Vermilion (SD 52/HD 104) / Expiration: 2007-08 school year WM300-2889 – Modification of School Code (Section 24-2) allows the district to hold a teacher institute on the legal holiday honoring Casimir Pulaski. The institute program is subject to prior approval from the Regional Office of Education. Instruction pertaining to the contributions of the honored individual will be provided through instructional activities rather than observing the legal school holiday.

Donovan CUSD 3 – Iroquois (SD 40/HD 79) / Expiration: 2007-08 school year WM300-2890 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Anna CCSD 37 – Union (SD 59/HD 118) / Expiration: 2007-08 school year WM300-2891 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Mascoutah CUSD 19 – St. Clair (SD 57/HD 114) / Expiration: 2007-08 school year WM300-2892 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Oakland CUSD 5 – Coles (SD 55/HD 110) / Expiration: 2007-08 school year WM300-2893 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Armstrong THSD 225 – Vermilion (SD 52/HD 104) / Expiration: 2007-08 school year WM300-2894 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Midwest Central CUSD 191 – Mason (SD 47/HD 94) / Expiration: 2007-08 school year WM300-2895 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Peoria Heights CUSD 325 – Peoria (SD 46/HD 92) / Expiration: 2003-04 school year WM300-2897 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Woodland CCSD 50 - Lake (SD 31/HD 62) / Expiration: 2003-04 school year

**WM300-2898 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Lake Bluff SD 65 – Lake (SD 29/HD 58) / Expiration: 2003-04 school year

**WM300-2899 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Cowden-Herrick CUSD 3A – Shelby (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2900 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Dwight THSD 230 – Livingston (SD 53/HD 105) / Expiration: 2005-06 school year WM300-2901-2 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Northwestern CUSD 2 – Macoupin (SD 49/HD 97) / Expiration: 2007-08 school year WM300-2903 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Warrensburg-Latham CUSD 11 – Macon (SD 44/HD 87) / Expiration: 2007-08 school year WM300-2904 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Tinley Park CCSD 146 – Cook (SD 19/HD 37) / Expiration: 2003-04 school year WM300-2905 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Berwyn South SD 100 – Cook (SD 12/HD 23) / Expiration: 2003-04 school year WM300-2906 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Avon CUSD 176 – Fulton (SD 47/HD 94) / Expiration: 2007-08 school year WM300-2908 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Burbank SD 111 – Cook (SD 11/HD 22) / Expiration: 2007-08 school year WM300-2910 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Sullivan CUSD 300 – Moultrie (SD 51/HD 101) / Expiration: 2007-08 school year WM300-2912 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Pembroke CCSD 259 – Kankakee (SD 40/HD 79) / Expiration: 2007-08 school year WM300-2914 – Modification of School Code (Section 24-2) allows the district to recognize

the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Alden-Hebron SD 19 – McHenry (SD 32/HD 63) / Expiration: 2007-08 school year WM300-2915 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Mundelein CHSD 120 – Lake (SD 26/HD 51) / Expiration: 2006-07 school year WM300-2916 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Grayslake CHSD 127 – Lake (SD 31/HD 62) / Expiration: 2003-04 school year WM300-2917 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Joliet THSD 204 – Will (SD 42/HD 84) / Expiration: 2003-04 school year WM300-2919 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Braceville SD 75 – Grundy (SD 38/HD 75) / Expiration: 2003-04 school year WM300-2920 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Rock Island SD 41 – Rock Island (SD 36/HD 72) / Expiration: 2003-04 school year WM300-2921 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Rooks Creek CCSD 425 – Livingston (SD 53/HD 106) / Expiration: 2007-08 school year WM300-2922 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Farmington Central CUSD 265 – Peoria (SD 37/HD 73) / Expiration: 2007-08 school year WM300-2923 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Antioch CCSD 34 – Lake (SD 31/HD 61) / Expiration: 2007-08 school year WM300-2926 – Modification of School Code (Section 24-2) allows the district to hold school or a teacher institute on the legal holidays honoring Dr. Martin Luther King, Jr., Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars. The institute programs are subject to prior approval from the Regional Office of Education. Instruction pertaining to the contributions of the honored individuals will be provided through instructional activities rather than observing the legal school holidays.

Prairie Grove CSD 46 – McHenry (SD 26/HD 52) / Expiration: 2007-08 school year WM300-2927-1 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor.

Prairie Grove CSD 46 - McHenry (SD 26/HD 52) / Expiration: 2007-08 school year

**WM300-2927-2 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in his honor.

Oak Lawn CHSD 229 – Cook (SD 18/HD 36) / Expiration: 2007-08 school year WM300-2928 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Pontiac SD 429 – Livingston (SD 53/HD 106) / Expiration: 2007-08 school year WM300-2929 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Rantoul THSD 193 – Champaign (SD 52/HD 104) / Expiration: 2007-08 school year WM300-2930 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

St. Clair Safe School – St. Clair (SD 57/HD 113) / Expiration: 2007-08 school year WM300-2931 – Modification of School Code (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

Highland Park THSD 113 – Lake (SD 29/HD 58) / Expiration: 2005-06 school year WM300-2932 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

St. Anne CHSD 302 – Kankakee (SD 40/HD 79) / Expiration: 2007-08 school year WM300-2933 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Burnham SD 154-5 – Cook (SD 17/HD 34) / Expiration: 2007-08 school year WM300-2934 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Tamaroa SD 5 – Perry (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2935 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Mulberry Grove CUSD 1 – Bond, Fayette (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2936 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Gardner CCSD 72C – Grundy (SD 38/HD 75) / Expiration: 2007-08 school year WM300-2937 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

*Union SD 81 – Will* (SD 43/HD 86) / **Expiration: 2003-04 school year WM300-2939 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Monticello CUSD 25 - Piatt (SD 55/HD 110) / Expiration: 2007-08 school year

**WM300-2943 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Mt. Zion CUSD 3 – Macon, Moultrie (SD 51/HD 101) / Expiration: 2007-08 school year WM300-2944 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Sparta CUSD 140 – Randolph (SD 58/HD 116) / Expiration: 2005-06 school year WM300-2945 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

East Dubuque USD 119 – Jo Daviess (SD 45/HD 89) / Expiration: 2007-08 school year WM300-2946 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Altamont CUSD 10 – Effingham (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2947 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Joppa-Maple Grove UD 38 – Massac (SD 59/HD 118) / Expiration: 2007-08 school year WM300-2948 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Elverado CUSD 196 – Jackson (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2949 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Lyons ESD 103 - Cook (SD 11/HD 21) / Expiration: 2003-04 school year

**WM300-2950 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

SPEED Special Education Joint Agreement 802 – Cook (SD 40/HD 80) / Expiration: 2007-08 school year

**WM300-2951 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Bellwood SD 88 - Cook (SD 4/HD 7) / Expiration: 2007-08 school year

**WM300-2952 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Dolton SD 148 - Cook (SD 15/HD 30) / Expiration: 2007-08 school year

**WM300-2953 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Marshall CUSD C-2 - Clark (SD 55/HD 109) / Expiration: 2007-08 school year

**WM300-2955 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Cary CCSD 26 – McHenry (SD 32/HD 64) / Expiration: 2007-08 school year WM300-2956 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Stewardson-Strasburg CUD 5A – Shelby (SD 55/HD 109) / Expiration: 2007-08 school year WM300-2957 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Dieterich CUSD 30 – Effingham (SD 54/HD 108) / Expiration: 2007-08 school year WM300-2958 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Saratoga CCSD 60C – Grundy (SD 38/HD 75) / Expiration: 2007-08 school year WM300-2959 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Hinsdale CCSD 181– DuPage (SD 41/HD 82) / Expiration: 2008-09 school year WM300-2960 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Century CUSD 100 – Pulaski (SD 59/HD 118) / Expiration: 2007-08 school year WM300-2962 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Neponset CCSD 307 – Bureau (SD 37/HD 74) / Expiration: 2006-07 school year WM300-2963 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Robinson CUSD 2 – Crawford (SD 55/HD 109) / Expiration: 2007-08 school year WM300-2964 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Allendale CCSD 17 – Wabash (SD 55/HD 109) / Expiration: 2004-05 school year WM300-2965 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

East Coloma SD 12 – Whiteside (SD 45/HD 90) / Expiration: 2007-08 school year WM300-2967 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Cornell CHSD 70 - Livingston (SD 53/HD 106) / Expiration: 2007-08 school year

**WM300-2969 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Willow Grove SD 46 – Clinton (SD 54/HD 107) / Expiration: 2007-08 school year WM300-2970 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Gardner-South Wilmington THSD 73 – Grundy (SD 38/HD 75) / Expiration: 2007-08 school year

**WM300-2972 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Woodland CUSD 5 – Livingston (SD 53/HD 106) / Expiration: 2007-08 school year WM300-2973 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Norwood SD 63 – Peoria (SD 46/HD 92) / Expiration: 2008-09 school year WM300-2975 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. The modification will take effect in the 2004-05 school year.

Potomac CUSD 10 – Vermilion (SD 52/HD 104) / Expiration: 2003-04 school year WM300-2976 – Modification of School Code (Section 24-2) allows the district to hold a teacher institute on the legal holiday honoring Casimir Pulaski. The institute program is subject to prior approval from the Regional Office of Education. Instruction pertaining to the contributions of the honored individual will be provided through instructional activities rather than observing the legal school holiday.

Knoxville CUSD 202 – Knox (SD 37/HD 74) / Expiration: 2007-08 school year WM300-2977 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Berwyn North SD 98 – Cook (SD 4/HD 8) / Expiration: 2003-04 school year WM300-2978 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Nippersink SD 2 – McHenry (SD 32/HD 63) / Expiration: 2006-07 school year WM300-2981 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Wolf Branch SD 113 – St. Clair (SD 57/HD 114) / Expiration: 2007-08 school year WM300-2982 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Carlinville CUSD 1 – Macoupin (SD 49/HD 98) / Expiration: 2007-08 school year WM300-2984 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

*Princeville CUSD 326 – Peoria* (SD 37/HD 73) / **Expiration: 2007-08 school year WM300-2985 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Kewanee CUSD 229 – Henry (SD 37/HD 74) / Expiration: 2007-08 school year WM300-2986 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Ottawa THSD 140 – LaSalle (SD 38/HD 76) / Expiration: 2007-08 school year WM300-2988 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

New Berlin CUSD 16 – Sangamon (SD 50/HD 100) / Expiration: 2007-08 school year WM300-2989 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Oregon CUSD 220 – Ogle (SD 45/HD 90) / Expiration: 2005-06 school year WM300-2990 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Riverside Brookfield THSD 206 – Cook (SD 21/HD 41) / Expiration: 2008-09 school year WM300-2992 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Niantic-Harristown CUSD 6 – Macon, Christian (SD 44/HD 87) / Expiration: 2007-08 school vear

**WM300-2994 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Carol Stream CCSD 93 – DuPage (SD 23/HD 45) / Expiration: 2007-08 school year WM300-2995 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Lena-Winslow CUSD 202 – Stephenson (SD 45/HD 89) / Expiration: 2007-08 school year WM300-2996 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Christopher Columbus through instructional activities rather than observing a school holiday in his honor.

Antioch CHSD 117 – Lake (SD 31/HD 61) / Expiration: 2008-09 school year WM300-2997-1 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05

Pleasantdale SD 107 – Cook (SD 41/HD 82) / Expiration: 2007-08 school year

WM300-2998-1 (renewal) – Modification of School Code (Section 24-2) allows the district to

[October 23, 2003]

school year.

recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor.

Richland SD 88A - Will (SD 43/HD 85) / Expiration: 2008-09 school year

**WM300-2999 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Flanagan CUSD 44 – Livingston (SD 53/HD 106) / Expiration: 2005-06 school year WM300-3001 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Geff CCSD 14 – Wayne (SD 54/HD 108) / Expiration: 2008-09 school year WM300-3004 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor. This modification takes effect in the 2004-05 school year.

Limestone Walters CCSD 316 – Peoria (SD 37/HD 73) / Expiration: 2008-09 school year WM300-3006 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Roselle SD 12 - DuPage (SD 23/HD 45) / Expiration: 2008-09 school year

**WM300-3007-1 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Riverside SD 96 - Cook (SD 21/HD 41) / Expiration: 2008-09 school year

**WM300-3008 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor. This modification takes effect in the 2004-05 school year.

Odell CCSD 435 – Livingston (SD 53/HD 105) / Expiration: 2007-08 school year WM300-3009 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Nokomis CUSD 22 – Montgomery (SD 49/HD 98) / Expiration: 2008-09 school year WM300-3010 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Wauconda CUSD 118 – Lake (SD 30/HD 59) / Expiration: 2008-09 school year WM300-3011-1 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor. This modification takes effect in the 2004-05 school year.

Arlington Heights CCSD 59 - Cook (SD 33/HD 66) / Expiration: 2008-09 school year WM300-3014 (renewal) - Modification of School Code (Section 24-2) allows the district to

recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor. This modification takes effect in the 2004-05 school year.

Virginia CUSD 64 - Cass (SD 47/HD 93) / Expiration: 2007-08 school year

**WM300-3017 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Riverview CCSD 2 - Woodford (SD 37/HD 73) / Expiration: 2008-09 school year

**WM300-3018 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

 $\textit{Triopia CUSD 27-Morgan} \ (SD\ 49/HD\ 97) \ / \ \textbf{Expiration:} \ \ \textbf{2008-09 school year}$ 

**WM300-3020 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

South Wilmington CSD 74 – Grundy (SD 38/HD 75) / Expiration: 2007-08 school year WM300-3022 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Hollis CSD 328 - Peoria (SD 46/HD 91) / Expiration: 2008-09 school year

**WM300-3024 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Oblong CUSD 4 – Crawford (SD 55/HD 109) / Expiration: 2003-04 school year WM300-3025 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Darien SD 61 - DuPage (SD 41/HD 82) / Expiration: 2008-09 school year

WM300-3027 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Safe Schools - Savanna Site - Carroll (SD 36/HD 71) / Expiration: 2007-08 school year WM300-3029 - Modification of School Code (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

Safe Schools - Freeport Site - Stephenson (SD 45/HD 89) / Expiration: 2007-08 school year WM300-3030 - Modification of School Code (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

Cumberland CUSD 77 – Cumberland (SD 55/HD 109) / Expiration: 2007-08 school year WM300-3031 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a

school holiday in his honor.

Adams/Pike County Regional Safe Schools – Adams, Pike (SD 47/HD 93) / Expiration: 2007-08 school year

**WM300-3033 – Modification of School Code** (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

Ashton CUSD 275 - Lee (SD 45/HD 90) / Expiration: 2007-08 school year

**WM300-3034-1** (renewal) - Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Massac UD 1 - Massac (SD 59/HD 118) / Expiration: 2003-04 school year

**WM300-3035 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Rankin SD 98 - Tazewell (SD 46/HD 91) / Expiration: 2008-09 school year

**WM300-3036-2 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

 $\textit{Park Forest-Chicago Heights SD 163-Cook} \ (SD \ 19/HD \ 38) \ / \ \textbf{Expiration:} \ \ \textbf{2003-04 school year}$ 

**WM300-3037 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Hinsdale THSD 86 - DuPage (SD 24/HD 47) / Expiration: 2008-09 school year

**WM300-3042 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Jo Daviess-Carroll Area Vocational Center – Jo Daviess, Carroll (SD 45/HD 89) / Expiration: 2007-08 school year

**WM300-3043 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Sandridge SD 172 - Cook (SD 40/HD 80) / Expiration: 2005-06 school year

**WM300-3044-2** (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

# **Physical Education**

Vienna HSD 133 – Johnson (SD 59/HD 118) / Expiration: 2007-08 school year

**WM300-2974 (renewal) – Modification of School Code** (Section 27-6) allows the district to permit students in grades 9 through 12 to participate in physical education every day for one semester only for 80 to 85 minutes each session rather than daily due to a 4-block schedule.

Bureau Valley CUSD 340 – Bureau (SD 37/HD 74) / Expiration: 2008-09 school year WM300-2993 (renewal) – Modification of School Code (Section 27-6) allows the district to permit students in grades 9 through 12 to participate in physical education every other day for

83 minutes each session rather than daily due to an 8-block schedule. This modification takes effect in the 2004-05 school year.

Lemont-Bromberek CSD 113A – Cook (SD 41/HD 82) / Expiration: 2008-09 school year WM300-3016 (renewal) – Modification of School Code (Section 27-6) allows the district to permit students in kindergarten through grade 5 to participate in physical education every other day for 40 minutes each session rather than daily for 20 minutes each session. This modification takes effect in the 2004-05 school year.

Palatine THSD 211 - Cook (SD 28/HD 56) / Expiration: 2008-09 school year

**WM300-3021 (renewal) – Modification of School Code** (Section 27-6) allows the district to permit students in grades 9 through 12 to participate in physical education every other day for 70 to 75 minutes each session rather than daily due to an 8-block schedule. This modification takes effect in the 2004-05 school year.

Ashton CUSD 275 – Lee (SD 45/HD 90) / Expiration: 2007-08 school year WM300-3034-2 (renewal) - Modification of School Code (Section 27-6) allows the district to permit students in grades 7 through 12 to participate in physical education every other day for 82 to 85 minutes each session rather than daily due to an 8-block schedule.

# Prairie State Achievement Examination - Instructional Time

Northfield THSD 225 - Cook (SD 9/HD 17) / Expiration: 2006-07 school year

**WM300-2913 (renewal) – Modification of School Code** (Section 18-8.05(F)(2)(d)) allows the district to hold school for students in grades 9 through 12 for three hours on the first day of the April administration of the Prairie State Achievement Examination. The district will accumulate sufficient student attendance time beyond the five-clock-hour instructional day to be applied to these days so that they can be counted among the 176 days of pupil attendance required by Section 10-19.

St. Charles CUSD 303 - Kane (SD 25/HD 49) / Expiration: 2006-07 school year

**WM300-2918 (renewal) – Modification of School Code** (Section 18-8.05(F)(2)(d)) allows the district to dismiss students in grade 11 following the administration of the Prairie State Achievement Examination for two days in April and to not hold school on those days for students in grade 12. The district will accumulate sufficient student attendance time beyond the five-clock-hour instructional day to be applied to these days so that they can be counted among the 176 days of pupil attendance required by Section 10-19.

Wauconda CUSD 118 - Lake (SD 30/HD 59) / Expiration: 2007-08 school year

**WM300-3011-2** (renewal) – Modification of School Code (Section 18-8.05(F)(2)(d)) allows the district to dismiss students in grade 11 two days in April following the administration of the Prairie State Achievement Examination. Students in grades 9, 10, and 12 will attend school for a half day on the first day of the examinations and have no school on the second day. The district will accumulate sufficient student attendance time beyond the five-clock-hour instructional day to be applied to these days so that they can be counted among the 176 days of pupil attendance required by Section 10-19.

# Substitute Teachers

Sandridge SD 172 - Cook (SD 40/HD 80) / Expiration: 2003-04 school year

**WM300-3044-1** (renewal) – Modification of School Code (Section 21-9) allows the district to employ a substitute teacher up to 130 paid school days for a single school year instead of 90 paid days. The district had to let one of its substitute teachers go due to budget constraints; the remaining substitute, who also works part-time as a teacher's aide, will work up to 130 days if it is necessary.

#### **SECTION IV**

# **Applications Returned to Applicants**

Listed below are several categories of requests that have been returned to school districts or other entities. Some of these applicants sought permission for actions that were already permissible under the law or rules. Other requests were returned because they were ineligible under the law (e.g., mandates not found in the School Code, applicant is not eligible to apply, application incomplete).

#### NO WAIVER NEEDED

#### Holidays

Vienna SD 55 – Johnson (SD 59/HD 118) / Expiration: 2006-07 school year

**WM300-2907 (renewal) – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district currently has an approved modification for this time period.

Red Brick Alternative School – Monroe, Randolph (SD 58/HD 116) / Expiration: 2006-07 school year

**WM300-2979 (renewal) – Modification of School Code** (Section 24-2). The regional office of education requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln for alternative schools established under Article 13A of the School Code. The regional office currently has an approved modification for this time period.

Pleasantdale SD 107 – Cook (SD 41/HD 82) / Expiration: 2007-08 school year WM300-2998-2 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring all veterans of foreign wars. The district currently has an approved modification for this time period.

#### **INELIGIBLE**

#### **Driver Education**

Robinson CUSD 2 – Crawford (SD 55/HD 109) / Expiration: 2007-08 school year WM300-3045 – Modification of Administrative Rule (23 III. Adm. Code 252.20(b)(1) and (c)(4) and (5)). The district requested to be allowed to offer the classroom and behind-thewheel portions of driver education during the summer for a two-week period, two days a week for three hours a day. The district failed to provide proper notice of the public hearing held to consider the request.

#### Holidays

Mulberry Grove CUSD 1 – Bond, Fayette (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2876 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

Altamont CUSD 10 – Effingham (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2883 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

St. Clair Safe School – St. Clair (SD 57/HD 113) / Expiration: 2007-08 school year WM300-2887 – Modification of School Code (Section 24-2). The regional office of education requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski for alternative schools established under Article 13A of the School Code. The regional office failed to provide proper notice of the public hearing held to consider the request.

Stewardson-Strasburg CUD 5A - Shelby (SD 55/HD 109) / Expiration: 2007-08 school year

**WM300-2888 – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

East Dubuque USD 119 – Jo Daviess (SD 45/HD 89) / Expiration: 2007-08 school year WM300-2896 – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Dwight Common SD 232 – Livingston (SD 53/HD 105) / Expiration: 2005-06 school year WM300-2902-2 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Elverado CUSD 196 – Jackson (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2909 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus. The district failed to provide proper notice of the public hearing held to consider the request.

Livingston County Special Services Unit – Livingston (SD 53/HD 106) / Expiration: 2003-04 school year

**WM300-2911 – Modification of School Code** (Section 24-2). The special education cooperative requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The cooperative failed to hold a public hearing to consider the request.

Allendale CCSD 17 – Wabash (SD 55/HD 109) / Expiration: 2004-05 school year WM300-2925 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

Willow Grove SD 46 – Clinton (SD 54/HD 107) / Expiration: 2007-08 school year WM300-2940 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to hold a public hearing to consider the request.

Illiopolis CUSD 12 – Sangamon (SD 44/HD 87) / Expiration: 2007-08 school year WM300-2941 – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

South Holland SD 151 – Cook (SD 15/HD 29) / Expiration: 2007-08 school year WM300-2942 – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Richland SD 88A – Will (SD 43/HD 85) / Expiration: 2007-08 school year WM300-2966 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on certain unspecified legal holidays. The district failed to provide proper notice of the public hearing held to consider the request.

Flanagan CUSD 4 – Livingston (SD 53/HD 106) / Expiration: 2005-06 school year WM300-2971 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring all veterans of foreign wars. The district failed to provide proper notice of the public hearing held to consider the request.

R.O.W.V.A. CUSD 208 - Knox, Henry (SD 37/HD 74) / Expiration: 2008-09 school year

**WM300-2983 (renewal) – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Adams/Pike County Regional Safe Schools – Adams, Pike (SD 47/HD 93) / Expiration: 2007-08 school year

**WM300-2991 – Modification of School Code** (Section 24-2). The regional office requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski for alternative schools established under Article 13A. The regional office failed to hold a public hearing to consider the request.

Jo Daviess-Carroll Area Vocational Center – Jo Daviess, Carroll (SD 45/HD 89) / Expiration: 2007-08 school year

**WM300-3002 – Modification of School Code** (Section 24-2). The center requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The center failed to provide proper notice of the public hearing held to consider the request.

Abingdon CUSD 217 – Knox, Warren (SD 37/HD 74) / Expiration: 2005-06 school year WM300-3003 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

Mount Olive CUSD 5 – Macoupin (SD 49/HD 98) / Expiration: 2008-09 school year WM300-3023 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to hold a public hearing to consider the request.

Wood Dale SD 7 – DuPage (SD 23/HD 46) / Expiration: 2007-08 school year WM300-3040 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

# Physical Education

Grant Park CUSD 6 – Kankakee (SD 40/HD 79) / Expiration: 2007-08 school year WM300-3000 – Modification of School Code (Section 27-6). The district requested to be allowed to hold physical education classes every other day for 84 minutes a session rather than daily. The district failed to provide proper notice of the public hearing held to consider the request.

# Prairie State Achievement Examination - Instructional Time

LaSalle-Peru THSD 120 – LaSalle (SD 38/HD 76) / Expiration: 2003-04 school year WM300-3039 (renewal) – Modification of School Code (Section 18-8.05(F)(2)(d)). The district requested to be allowed to adjust the length of the school day for some students on the days that the Prairie State Achievement Examination is administered. The district failed to hold a public hearing to consider the request.

# School Improvement/Inservice Training

Dwight Common SD 232 – Livingston (SD 53/HD 105) / Expiration: 2005-06 school year WM100-2902-1 (renewal) – Waiver of School Code (Section 18-8.05(F)(2)(d)(2)). The district requested to be allowed to dismiss students at 11:30 a.m. for three days the first week of the school year for the purpose of holding school improvement activities, and to accumulate sufficient student attendance time beyond the five-clock-hour requirement within the next six school days to apply towards these days. The district failed to provide proper notice of the public hearing held to consider the request.

#### **SECTION V**

# **Applications by Senate and House Districts**

All requests received during this waiver cycle are presented numerically by Senate and House district, and then alphabetically by school district or eligible applicant. The "action" to be taken or already taken for each request is noted; that is, requests for waivers upon which the General Assembly must act are noted as "waivers", modifications already acted upon by the State Board of Education are noted as "modifications", and requests that were returned for a variety of reasons are listed under their respective legislative district.

### SD 4/HD 7

Bellwood SD 88 - Cook (SD 4/HD 7) / Expiration: 2007-08 school year

**WM300-2952 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Oak Park ESD 97 - Cook (SD 4/HD 7) / Expiration: 2007-08 school year

**WM300-2954 – Waiver of Administrative Rules** (23 III. Adm. Code 120.90(e)). The district requested to be allowed to treat separately for the purposes of calculating its reimbursement for pupil transportation the category rather than the contractor since the district employs only one contractor to provide both regular and special education transportation. The State Board of Education denied this request because the district failed to show that its request would meet the intent of the mandate in a more effective, efficient or economical manner.

Oak Park ESD 97 - Cook (SD 4/HD 7) / Expiration: 2007-08 school year

**WM100-3032** (appeal) – Waiver of Administrative Rules (23 III. Adm. Code 120.90(e)). The district is appealing the State Board of Education's denial of its petition requesting to be allowed to treat separately for the purposes of calculating its reimbursement for pupil transportation the category rather than the contractor since the district employs only one contractor to provide both regular and special education transportation. The district states that using the proration mechanism provided in the rules will reduce the district's annual transportation reimbursement from the state by more than \$60,000.

The rule that is the subject of this waiver is similar to the regulatory requirement that districts that own and operate their own transportation services prorate total transportation costs across all categories of transportation services, based on the ratio of miles traveled in each category to total system miles (Section 120.90(d) of the rules). The same proration formula is used for a school district that chooses one contractor to provide all of its transportation services.

The Oak Park board is using the waiver process to request that the programs and claims for special education (which is reimbursed at a higher rate) and regular education services remain separate as if there were two contractors. While there are two separate contracts, State Board staff believe that the district did not contract with a "company that provides only one type of transportation service;" rather, there is only one contractor for all transportation services. Therefore, that contractor should prorate costs across all transportation services provided under all contracts held with the school district.

The State Board of Education denied this request because the district failed to show that its request would meet the intent of the mandate (i.e., cost containment) in a more effective, efficient or economical manner. While the district may receive a larger payment from the state if this request is approved, the intent of the rule has not been addressed. Furthermore, approval of this request could create a cost incentive for other districts to contract for transportation services unrelated to any real efficiencies in managing such programs. Finally, when the state is unable to fully fund transportation claims, the additional cost of this waiver will spread to all other school districts.

#### SD 4/HD 8

Berwyn North SD 98 - Cook (SD 4/HD 8) / Expiration: 2003-04 school year

**WM300-2978 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

#### SD 9/HD 17

Northfield THSD 225 - Cook (SD 9/HD 17) / Expiration: 2006-07 school year

**WM300-2913 (renewal) – Modification of School Code** (Section 18-8.05(F)(2)(d)) allows the district to hold school for students in grades 9 through 12 for three hours on the first day of the April administration of the Prairie State Achievement Examination. The district will accumulate sufficient student attendance time beyond the five-clock-hour instructional day to be applied to these days so that they can be counted among the 176 days of pupil attendance required by Section 10-19.

### SD 9/HD 18

Evanston CCSD 65 - Cook (SD 9/HD 18) / Expiration: 2008-09 school year

**WM100-3013 (renewal) – Waiver of School Code** (Section 24A-5) request to allow the district to replace the current rating scale of "excellent, satisfactory, and unsatisfactory" for the teacher evaluation process with "meets district standards of excellence, needs to improve to meet district standards, and unsatisfactory." If approved, this waiver would take effect in the 2004-05 school year.

### SD 11/HD 21

Lyons ESD 103 - Cook (SD 11/HD 21) / Expiration: 2003-04 school year

**WM300-2950 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

#### SD 11/HD 22

Burbank SD 111 - Cook (SD 11/HD 22) / Expiration: 2007-08 school year

**WM300-2910 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

# SD 12/HD 23

Berwyn South SD 100 - Cook (SD 12/HD 23) / Expiration: 2003-04 school year

**WM300-2906 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

# SD 15/HD 29

South Holland SD 151 - Cook (SD 15/HD 29) / Expiration: 2007-08 school year

**WM300-2942 – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

### SD 15/HD 30

Dolton SD 148 - Cook (SD 15/HD 30) / Expiration: 2007-08 school year

**WM300-2953 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 17/HD 34

Burnham SD 154-5 - Cook (SD 17/HD 34) / Expiration: 2007-08 school year

**WM300-2934 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 18/HD 36

Evergreen Park CHSD 231 – Cook (SD 18/HD 36) / Expiration: 2007-08 school year WM300-2874 – Waiver of Administrative Rule (23 III. Adm. Code 252.20(b)(1)) allows the district to provide behind-the-wheel instruction before and after school only, and on Saturdays, if additional sessions are needed.

Oak Lawn CHSD 229 – Cook (SD 18/HD 36) / Expiration: 2007-08 school year WM300-2928 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Oak Lawn CHSD 218 – Cook (SD 18/HD 36) / Expiration: 2008-09 school year WM100-3005 (renewal) – Waiver of School Code (Section 21-9) request to allow the district to employ substitute teachers for more than 90 days in any one school year. The most-qualified substitutes will be employed; each must meet the approval of the building principal in order to continue his or her employment. If approved, this waiver takes effect in the 2004-05 school year.

Oak Lawn CHSD 218 – Cook (SD 18/HD 36) / Expiration: 2008-09 school year WM100-3015 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$250 of students who participate in driver education courses. If approved, this waiver takes effect in the 2004-05 school year.

# SD 19/HD 37

Tinley Park CCSD 146 – Cook (SD 19/HD 37) / Expiration: 2003-04 school year WM300-2905 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 19/HD 38

Park Forest-Chicago Heights SD 163 - Cook (SD 19/HD 38) / Expiration: 2003-04 school year

**WM300-3037 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 21/HD 41

Elmhurst CUSD 205 – DuPage (SD 21/HD 41) / Expiration: 2007-08 school year WM100-2980 (renewal) – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$200 of students who participate in driver education courses.

Riverside SD 96 – Cook (SD 21/HD 41) / Expiration: 2008-09 school year WM300-3008 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather

than observing a school holiday in their honor. This modification takes effect in the 2004-05 school year.

Riverside Brookfield THSD 206 – Cook (SD 21/HD 41) / Expiration: 2008-09 school year WM300-2992 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

#### SD 23/HD 45

Carol Stream CCSD 93 – DuPage (SD 23/HD 45) / Expiration: 2007-08 school year WM300-2995 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Marquardt SD 15 – DuPage (SD 23/HD 45) / Expiration: 2007-08 school year WM300-2873 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Roselle SD 12 – DuPage (SD 23/HD 45) / Expiration: 2008-09 school year WM300-3007-1 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Roselle SD 12 – DuPage (SD 23/HD 45) / Expiration: 2008-09 school year WM100-3007-2 – Waiver of School Code (Section 18-8.05(F)(2)(d)(1)) request to allow the district to schedule the equivalent of two and a half days of parent-teacher conferences in a

district to schedule the equivalent of two and a half days of parent-teacher conferences in a single school year. The district plans a full day of parent-teacher conferences (from 1 to 8:30 p.m.) combined with a half day of inservice training in the fall; a half day of conferences to be held from 8 a.m. until noon in the fall; and a full day of conferences to be held from 1 to 8:30 p.m. in the spring. Students will not be in attendance on these days, and the district does not intend to count these days among the 176 days of pupil attendance required by Section 10-19. If approved, this waiver would take effect in the 2004-05 school year.

# SD 23/HD 46

Bensenville SD 2 – DuPage (SD 23/HD 46) / Expiration: 2003-04 school year WM300-2938 (renewal) – Modification of School Code (Sections 14C-2, 14C-2.1, and 14C-3) allows the district to offer an English immersion curriculum, with instruction in English, rather than a bilingual program that provides instruction in the student's home language.

Wood Dale SD 7 – DuPage (SD 23/HD 46) / Expiration: 2007-08 school year WM300-3040 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

### SD 24/HD 47

Hinsdale THSD 86 – DuPage (SD 24/HD 47) / Expiration: 2008-09 school year WM300-3042 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

#### SD 25/HD 49

St. Charles CUSD 303 - Kane (SD 25/HD 49) / Expiration: 2006-07 school year

WM300-2918 (renewal) – Modification of School Code (Section 18-8.05(F)(2)(d)) allows the district to dismiss students in grade 11 following the administration of the Prairie State Achievement Examination for two days in April and to not hold school on those days for students in grade 12. The district will accumulate sufficient student attendance time beyond the five-clock-hour instructional day to be applied to these days so that they can be counted among the 176 days of pupil attendance required by Section 10-19.

#### SD 25/HD 50

Kaneland CUSD 302 – Kane, DeKalb (SD 25/HD 50) / Expiration: 2008-09 school year WM100-2987 (renewal) – Waiver of School Code (Section 18-8.05(F)(2)(d)(2)) request to allow the district to hold two full-day teacher inservice sessions instead of four half days, and to count the days among the 176 days of pupil attendance required by Section 10-19. The district will accumulate sufficient time beyond the five-clock-hour requirement to apply towards these days. If approved, this waiver would take effect in the 2004-2005 school year.

### SD 26/HD 51

Lake Zurich SD 95 – Lake (SD 26/HD 51) / Expiration: 2007-08 school year WM100-2968 (renewal) – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$300 of students who participate in driver education courses.

Mundelein CHSD 120 – Lake (SD 26/HD 51) / Expiration: 2006-07 school year WM300-2916 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

#### SD 26/HD 52

Prairie Grove CSD 46 – McHenry (SD 26/HD 52) / Expiration: 2007-08 school year WM300-2927-1 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor.

Prairie Grove CSD 46 – McHenry (SD 26/HD 52) / Expiration: 2007-08 school year WM300-2927-2 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in his honor.

#### SD 28/HD 56

Palatine THSD 211 – Cook (SD 28/HD 56) / Expiration: 2008-09 school year WM300-3021 (renewal) – Modification of School Code (Section 27-6) allows the district to permit students in grades 9 through 12 to participate in physical education every other day for 70 to 75 minutes each session rather than daily due to an 8-block schedule. This modification takes effect in the 2004-05 school year.

#### SD 29/HD 58

Highland Park THSD 113 – Lake (SD 29/HD 58) / Expiration: 2005-06 school year WM300-2932 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Lake Bluff SD 65 - Lake (SD 29/HD 58) / Expiration: 2003-04 school year

**WM300-2899 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

#### SD 30/HD 59

Wauconda CUSD 118 - Lake (SD 30/HD 59) / Expiration: 2008-09 school year

**WM300-3011-1 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor. This modification takes effect in the 2004-05 school year.

Wauconda CUSD 118 - Lake (SD 30/HD 59) / Expiration: 2007-08 school year

**WM300-3011-2** (renewal) – Modification of School Code (Section 18-8.05(F)(2)(d)) allows the district to dismiss students in grade 11 two days in April following the administration of the Prairie State Achievement Examination. Students in grades 9, 10, and 12 will attend school for a half day on the first day of the examinations and have no school on the second day. The district will accumulate sufficient student attendance time beyond the five-clock-hour instructional day to be applied to these days so that they can be counted among the 176 days of pupil attendance required by Section 10-19.

Wauconda CUSD 118 - Lake (SD 30/HD 59) / Expiration: 2008-09 school year

**WM100-3011-3** (renewal) – Waiver of School Code (Section 27-6) request to allow the district to excuse students in grades 9 through 12 from daily physical education for ongoing participation in cheerleading and pom-pons. This request will allow these students more time during the school day to take additional academic courses. If approved, this waiver would take effect in the 2004-05 school year.

#### SD 31/HD 61

Antioch CCSD 34 - Lake (SD 31/HD 61) / Expiration: 2007-08 school year

**WM300-2926 – Modification of School Code** (Section 24-2) allows the district to hold school or a teacher institute on the legal holidays honoring Dr. Martin Luther King, Jr., Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars. The institute programs are subject to prior approval from the Regional Office of Education. Instruction pertaining to the contributions of the honored individuals will be provided through instructional activities rather than observing the legal school holidays.

Antioch CHSD 117 - Lake (SD 31/HD 61) / Expiration: 2008-09 school year

**WM300-2997-1 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Antioch CHSD 117 - Lake (SD 31/HD 61) / Expiration: 2008-09 school year

**WM100-2997-2** (renewal) – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$300 of students who participate in driver education courses. If approved, this waiver takes effect in the 2004-05 school year.

Zion-Benton THSD 126 - Lake (SD 31/HD 61) / Expiration: 2008-09 school year

**WM100-3028-1 (renewal) – Waiver of School Code** (Section 27-6) request to allow the district to excuse approximately 80 ninth-grade students who enroll in the "school-within-aschool Technology Academy" from the daily physical education requirement in order to take additional coursework required by the program and other elective courses that otherwise would not be available. Students could choose to take physical education as an elective course. If approved, this waiver takes effect in the 2004-05 school year.

Zion-Benton THSD 126 - Lake (SD 31/HD 61) / Expiration: 2003-04 school year

**WM100-3028-2 – Waiver of School Code** (Section 17-1.5) request to allow the district to waive the limitation of administrative costs due to circumstances beyond its control. For the last two school years, the district has employed an interim superintendent. A permanent superintendent has been hired for the 2003-04 school year with a salary and benefit package that is higher than that paid to the interim superintendent, causing the district to exceed the 5 percent limitation.

#### SD 31/HD 62

Grayslake CHSD 127 - Lake (SD 31/HD 62) / Expiration: 2003-04 school year

**WM300-2917 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Grayslake CHSD 127 - Lake (SD 31/HD 62) / Expiration: 2008-09 school year

**WM100-3038 (renewal) – Waiver of School Code** (Section 27-6) request to allow the district to excuse students in grades 11 and 12 from daily physical education for ongoing participation in cheerleading and pom-pons if they are enrolled in six academic classes. If approved, this waiver would take effect in the 2004-05 school year.

Warren THSD 121 - Lake (SD 31/HD 62) / Expiration: 2007-08 school year

**WM300-2875 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, members of the armed forces killed in war, Columbus Day, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Woodland CCSD 50 - Lake (SD 31/HD 62) / Expiration: 2003-04 school year

**WM300-2898 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 32/HD 63

Alden-Hebron SD 19 – McHenry (SD 32/HD 63) / Expiration: 2007-08 school year WM300-2915 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Harvard CUSD 50 – McHenry (SD 32/HD 63) / Expiration: 2007-08 school year WM100-3026 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$200 of students who participate in driver education courses.

Nippersink SD 2 - McHenry (SD 32/HD 63) / Expiration: 2006-07 school year

**WM300-2981 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

#### SD 32/HD 64

Cary CCSD 26 - McHenry (SD 32/HD 64) / Expiration: 2007-08 school year

WM300-2956 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

# SD 33/HD 66

Arlington Heights CCSD 59 – Cook (SD 33/HD 66) / Expiration: 2008-09 school year WM300-3014 (renewal) – Modification of School Code (Section 24-2) allows the district to

recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor. This modification takes effect in the 2004-05 school year.

# SD 34/HD 68

Hononegah CHSD 207 – Winnebago (SD 34/HD 68) / Expiration: 2008-09 school year WM100-3019 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$200 of students who participate in driver education courses. If approved, this waiver takes effect in the 2004-05 school year.

Kinnikinnick CCSD 131 – Winnebago (SD 34/HD 68) / Expiration: 2007-08 school year WM300-2882 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

South Beloit CUSD 320 – Winnebago (SD 34/HD 68) / Expiration: 2007-08 school year WM100-2924 – Waiver of School Code (Section 18-8.05(F)(2)(d)(1)) request to allow the district the option of scheduling twice during the school year two, four-hour parent-teacher conferences in the evening following a full day of student attendance. The four evening sessions will be counted as two of the 176 days of pupil attendance required by Section 10-19.

### SD 36/HD 71

Safe Schools - Savanna Site - Carroll (SD 36/HD 71) / Expiration: 2007-08 school year WM300-3029 - Modification of School Code (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

#### SD 36/HD 72

Rock Island SD 41 – Rock Island (SD 36/HD 72) / Expiration: 2003-04 school year WM300-2921 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 37/HD 73

Farmington Central CUSD 265 – Peoria (SD 37/HD 73) / Expiration: 2007-08 school year WM300-2923 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Limestone Walters CCSD 316 – Peoria (SD 37/HD 73) / Expiration: 2008-09 school year WM300-3006 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

*Princeville CUSD 326 – Peoria* (SD 37/HD 73) / **Expiration: 2007-08 school year WM300-2985 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Riverview CCSD 2 – Woodford (SD 37/HD 73) / Expiration: 2008-09 school year WM300-3018 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional

activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

### SD 37/HD 74

Abingdon CUSD 217 – Knox, Warren (SD 37/HD 74) / Expiration: 2005-06 school year WM300-3003 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

Bureau Valley CUSD 340 – Bureau (SD 37/HD 74) / Expiration: 2008-09 school year WM300-2993 (renewal) – Modification of School Code (Section 27-6) allows the district to permit students in grades 9 through 12 to participate in physical education every other day for 83 minutes each session rather than daily due to an 8-block schedule. This modification takes effect in the 2004-05 school year.

Galesburg CUSD 205 – Knox (SD 37/HD 74) / Expiration: 2007-08 school year WM100-2886 (renewal) – Waiver of School Code (Section 24A-5) request to allow the district to replace the current rating scale of "excellent, satisfactory, and unsatisfactory" for the teacher evaluation process with a professional growth plan that incorporates a professional assistance component that allows for remediation of teachers with unsatisfactory performance, as required under Section 24A-5.

Kewanee CUSD 229 – Henry (SD 37/HD 74) / Expiration: 2007-08 school year WM300-2986 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Knoxville CUSD 202 – Knox (SD 37/HD 74) / Expiration: 2007-08 school year WM300-2977 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Neponset CCSD 307 – Bureau (SD 37/HD 74) / Expiration: 2006-07 school year WM300-2963 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

R.O.W.V.A. CUSD 208 – Knox, Henry (SD 37/HD 74) / Expiration: 2008-09 school year WM300-2983 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

#### SD 38/HD 75

Braceville SD 75 – Grundy (SD 38/HD 75) / Expiration: 2003-04 school year WM300-2920 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Gardner CCSD 72C – Grundy (SD 38/HD 75) / Expiration: 2007-08 school year WM300-2937 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

 $\textit{Gardner-South Wilmington THSD 73-Grundy} \ (\text{SD 38/HD 75}) \ / \ \textbf{Expiration:} \ \ \textbf{2007-08 school year}$ 

WM300-2972 - Modification of School Code (Section 24-2) allows the district to recognize

the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Saratoga CCSD 60C – Grundy (SD 38/HD 75) / Expiration: 2007-08 school year WM300-2959 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

South Wilmington CSD 74 – Grundy (SD 38/HD 75) / Expiration: 2007-08 school year WM300-3022 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 38/HD 76

LaSalle-Peru THSD 120 – LaSalle (SD 38/HD 76) / Expiration: 2003-04 school year WM300-3039 (renewal) – Modification of School Code (Section 18-8.05(F)(2)(d)). The district requested to be allowed to adjust the length of the school day for some students on the days that the Prairie State Achievement Examination is administered. The district failed to hold a public hearing to consider the request.

Ottawa THSD 140 – LaSalle (SD 38/HD 76) / Expiration: 2007-08 school year WM300-2988 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Rutland CCSD 230 – LaSalle (SD 38/HD 76) / Expiration: 2007-08 school year WM300-2880 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

#### SD 39/HD 78

Oak Park-River Forest HSD 200 – Cook (SD 39/HD 78) / Expiration: 2007-08 school year WM100-3041 – Waiver of School Code (Section 27-23) request to allow the district to charge a reasonable fee not to exceed \$350 of students who participate in driver education courses.

### SD 40/HD 79

Donovan CUSD 3 – Iroquois (SD 40/HD 79) / Expiration: 2007-08 school year WM300-2890 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Grant Park CUSD 6 – Kankakee (SD 40/HD 79) / Expiration: 2007-08 school year WM300-3000 – Modification of School Code (Section 27-6). The district requested to be allowed to hold physical education classes every other day for 84 minutes a session rather than daily. The district failed to provide proper notice of the public hearing held to consider the request.

Pembroke CCSD 259 – Kankakee (SD 40/HD 79) / Expiration: 2007-08 school year WM300-2914 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

St. Anne CHSD 302 – Kankakee (SD 40/HD 79) / Expiration: 2007-08 school year WM300-2933 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a

school holiday in his honor.

#### SD 40/HD 80

Sandridge SD 172 - Cook (SD 40/HD 80) / Expiration: 2003-04 school year

**WM300-3044-1 (renewal) – Modification of School Code** (Section 21-9) allows the district to employ a substitute teacher up to 130 paid school days for a single school year instead of 90 paid days. The district had to let one of its substitute teachers go due to budget constraints; the remaining substitute, who also works part-time as a teacher's aide, will work up to 130 days if it is necessary.

Sandridge SD 172 - Cook (SD 40/HD 80) / Expiration: 2005-06 school year

**WM300-3044-2** (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

SPEED Special Education Joint Agreement 802 - Cook (SD 40/HD 80) / Expiration: 2007-08 school year

**WM300-2951 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

### SD 41/HD 82

Darien SD 61 - DuPage (SD 41/HD 82) / Expiration: 2008-09 school year

**WM300-3027 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Hinsdale CCSD 181- DuPage (SD 41/HD 82) / Expiration: 2008-09 school year

**WM300-2960 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Lemont-Bromberek CSD 113A – Cook (SD 41/HD 82) / Expiration: 2008-09 school year WM300-3016 (renewal) – Modification of School Code (Section 27-6) allows the district to permit students in kindergarten through grade 5 to participate in physical education every other day for 40 minutes each session rather than daily for 20 minutes each session. This modification takes effect in the 2004-05 school year.

Pleasantdale SD 107 - Cook (SD 41/HD 82) / Expiration: 2007-08 school year

**WM300-2998-1 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor.

Pleasantdale SD 107 – Cook (SD 41/HD 82) / Expiration: 2007-08 school year

**WM300-2998-2 (renewal) – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring all veterans of foreign wars. The district currently has an approved modification for this time period.

### SD 42/HD 84

Joliet THSD 204 - Will (SD 42/HD 84) / Expiration: 2003-04 school year

WM300-2919 - Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a

school holiday in his honor.

*Troy SD 30-C – Will* (SD 42/HD 84) / **Expiration: 2003-04 school year** 

**WM300-2881 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 43/HD 85

Richland SD 88A - Will (SD 43/HD 85) / Expiration: 2007-08 school year

**WM300-2966 (renewal) – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on certain unspecified legal holidays. The district failed to provide proper notice of the public hearing held to consider the request.

Richland SD 88A - Will (SD 43/HD 85) / Expiration: 2008-09 school year

**WM300-2999 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

#### SD 43/HD 86

Union SD 81 - Will (SD 43/HD 86) / Expiration: 2003-04 school year

**WM300-2939 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 44/HD 87

Deer Creek-Mackinaw CUSD 701 - Tazewell (SD 44/HD 87) / Expiration: 2008-09 school vear

**WM100-3012 – Waiver of School Code** (Section 24A-5) request to allow the district to replace the current rating scale of "excellent, satisfactory, and unsatisfactory" for the teacher evaluation process with a professional growth plan that incorporates a professional assistance component that allows for remediation of teachers with unsatisfactory performance, as required under Section 24A-5. If approved, this waiver would take effect in the 2004-05 school year.

Illiopolis CUSD 12 - Sangamon (SD 44/HD 87) / Expiration: 2007-08 school year

**WM300-2941 – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Niantic-Harristown CUSD 6 – Macon, Christian (SD 44/HD 87) / Expiration: 2007-08 school year

**WM300-2994 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Warrensburg-Latham CUSD 11 – Macon (SD 44/HD 87) / Expiration: 2007-08 school year WM300-2904 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

### SD 45/HD 89

East Dubuque USD 119 – Jo Daviess (SD 45/HD 89) / Expiration: 2007-08 school year WM300-2896 – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to

provide proper notice of the public hearing held to consider the request.

East Dubuque USD 119 – Jo Daviess (SD 45/HD 89) / Expiration: 2007-08 school year WM300-2946 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Jo Daviess-Carroll Area Vocational Center – Jo Daviess, Carroll (SD 45/HD 89) / Expiration: 2007-08 school year

**WM300-3002 – Modification of School Code** (Section 24-2). The center requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The center failed to provide proper notice of the public hearing held to consider the request.

Jo Daviess-Carroll Area Vocational Center – Jo Daviess, Carroll (SD 45/HD 89) / Expiration: 2007-08 school year

**WM300-3043 – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Lena-Winslow CUSD 202 – Stephenson (SD 45/HD 89) / Expiration: 2007-08 school year WM300-2996 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Christopher Columbus through instructional activities rather than observing a school holiday in his honor.

Safe Schools - Freeport Site - Stephenson (SD 45/HD 89) / Expiration: 2007-08 school year WM300-3030 - Modification of School Code (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

#### SD 45/HD 90

East Coloma SD 12 – Whiteside (SD 45/HD 90) / Expiration: 2007-08 school year WM300-2967 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Oregon CUSD 220 – Ogle (SD 45/HD 90) / Expiration: 2005-06 school year

**WM300-2990 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Polo CUSD 222 – Ogle, Lee, Whiteside (SD 45/HD 90) / Expiration: 2007-08 school year WM300-2872 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Ashton CUSD 275 - Lee (SD 45/HD 90) / Expiration: 2007-08 school year

**WM300-3034-1 (renewal) - Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Ashton CUSD 275 - Lee (SD 45/HD 90) / Expiration: 2007-08 school year

**WM300-3034-2 (renewal) - Modification of School Code** (Section 27-6) allows the district to permit students in grades 7 through 12 to participate in physical education every other day for 82 to 85 minutes each session rather than daily due to an 8-block schedule.

### SD 46/HD 91

Hollis CSD 328 - Peoria (SD 46/HD 91) / Expiration: 2008-09 school year

**WM300-3024 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

Rankin SD 98 - Tazewell (SD 46/HD 91) / Expiration: 2008-09 school year

**WM100-3036-1 (renewal) – Waiver of School Code** (Section 10-20.12a) request to allow the district to charge non-resident pupil tuition in an amount that is less than 100 percent of the preceding year's per capita tuition costs to students who wish to enroll in the district. The district maintains that a lower tuition rate will help it attract additional students necessary to offer a full curriculum without combining grade levels. If approved, this waiver would take effect in the 2004-05 school year.

Rankin SD 98 - Tazewell (SD 46/HD 91) / Expiration: 2008-09 school year

**WM300-3036-2 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

### SD 46/HD 92

Norwood SD 63 – Peoria (SD 46/HD 92) / Expiration: 2008-09 school year

**WM300-2975 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor. The modification will take effect in the 2004-05 school year.

Peoria Heights CUSD 325 – Peoria (SD 46/HD 92) / Expiration: 2003-04 school year

**WM300-2897 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

### SD 47/HD 93

Adams/Pike County Regional Safe Schools – Adams, Pike (SD 47/HD 93) / Expiration: 2007-08 school year

**WM300-2991 – Modification of School Code** (Section 24-2). The regional office requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski for alternative schools established under Article 13A. The regional office failed to hold a public hearing to consider the request.

Adams/Pike County Regional Safe Schools – Adams, Pike (SD 47/HD 93) / Expiration: 2007-08 school year

**WM300-3033 – Modification of School Code** (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

Virginia CUSD 64 - Cass (SD 47/HD 93) / Expiration: 2007-08 school year

WM300-3017 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

# SD 47/HD 94

Avon CUSD 176 - Fulton (SD 47/HD 94) / Expiration: 2007-08 school year

**WM300-2908 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Midwest Central CUSD 191 – Mason (SD 47/HD 94) / Expiration: 2007-08 school year WM300-2895 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

#### SD 49/HD 97

Northwestern CUSD 2 – Macoupin (SD 49/HD 97) / Expiration: 2007-08 school year WM300-2903 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

*Triopia CUSD 27 – Morgan (SD 49/HD 97) /* **Expiration: 2008-09 school year WM300-3020 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

### SD 49/HD 98

Carlinville CUSD 1 – Macoupin (SD 49/HD 98) / Expiration: 2007-08 school year WM300-2984 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Mount Olive CUSD 5 – Macoupin (SD 49/HD 98) / Expiration: 2008-09 school year WM300-3023 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to hold a public hearing to consider the request.

Nokomis CUSD 22 – Montgomery (SD 49/HD 98) / Expiration: 2008-09 school year WM300-3010 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor. This modification takes effect in the 2004-05 school year.

### SD 50/HD 100

New Berlin CUSD 16 – Sangamon (SD 50/HD 100) / Expiration: 2007-08 school year WM300-2989 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 51/HD 101

Mt. Zion CUSD 3 – Macon, Moultrie (SD 51/HD 101) / Expiration: 2007-08 school year WM300-2944 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

 $\begin{tabular}{ll} Sullivan CUSD 300-Moultrie (SD 51/HD 101) / Expiration: {\bf 2007-08 school year} \\ {\bf WM300-2912 (renewal) - Modification of School Code} (Section 24-2) allows the district to {\bf 1000 code} (Section 24-2) allows the district to {\bf 1000 code} (Section 24-2) allows the district to {\bf 1000 code} (Section 24-2) allows the {\bf 1000 code} (Section 24$ 

recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 51/HD 102

Altamont CUSD 10 – Effingham (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2883 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Altamont CUSD 10 – Effingham (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2947 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Cowden-Herrick CUSD 3A – Shelby (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2900 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Mulberry Grove CUSD 1 – Bond, Fayette (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2876 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

Mulberry Grove CUSD 1 – Bond, Fayette (SD 51/HD 102) / Expiration: 2007-08 school year WM300-2936 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

### SD 52/HD 104

Armstrong THSD 225 – Vermilion (SD 52/HD 104) / Expiration: 2007-08 school year WM300-2894 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, Christopher Columbus, and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Armstrong-Ellis CSD 61 – Vermilion (SD 52/HD 104) / Expiration: 2007-08 school year WM300-2889 – Modification of School Code (Section 24-2) allows the district to hold a teacher institute on the legal holiday honoring Casimir Pulaski. The institute program is subject to prior approval from the Regional Office of Education. Instruction pertaining to the contributions of the honored individual will be provided through instructional activities rather than observing the legal school holiday.

Potomac CUSD 10 – Vermilion (SD 52/HD 104) / Expiration: 2003-04 school year WM300-2976 – Modification of School Code (Section 24-2) allows the district to hold a teacher institute on the legal holiday honoring Casimir Pulaski. The institute program is subject to prior approval from the Regional Office of Education. Instruction pertaining to the contributions of the honored individual will be provided through instructional activities rather than observing the legal school holiday.

Rantoul THSD 193 – Champaign (SD 52/HD 104) / Expiration: 2007-08 school year WM300-2930 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 53/HD 105

Dwight THSD 230 - Livingston (SD 53/HD 105) / Expiration: 2005-06 school year

**WM100-2901-1** (renewal) – Waiver of School Code (Section 18-8.05(F)(2)(d)(2)) request to allow the district to dismiss students at 11:30 a.m. for three days the first week of the school year for the purpose of holding school improvement activities, and to accumulate sufficient student attendance time beyond the five-clock-hour requirement within the next six school days to apply towards these days.

Dwight THSD 230 – Livingston (SD 53/HD 105) / Expiration: 2005-06 school year WM300-2901-2 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Dwight Common SD 232 – Livingston (SD 53/HD 105) / Expiration: 2005-06 school year WM100-2902-1 (renewal) – Waiver of School Code (Section 18-8.05(F)(2)(d)(2)). The district requested to be allowed to dismiss students at 11:30 a.m. for three days the first week of the school year for the purpose of holding school improvement activities, and to accumulate sufficient student attendance time beyond the five-clock-hour requirement within the next six school days to apply towards these days. The district failed to provide proper notice of the public hearing held to consider the request.

Dwight Common SD 232 – Livingston (SD 53/HD 105) / Expiration: 2005-06 school year WM300-2902-2 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Odell CCSD 435 – Livingston (SD 53/HD 105) / Expiration: 2007-08 school year WM300-3009 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 53/HD 106

Cornell CHSD 70 – Livingston (SD 53/HD 106) / Expiration: 2007-08 school year WM300-2969 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Flanagan CUSD 4 – Livingston (SD 53/HD 106) / Expiration: 2005-06 school year WM300-2971 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring all veterans of foreign wars. The district failed to provide proper notice of the public hearing held to consider the request.

Flanagan CUSD 44 – Livingston (SD 53/HD 106) / Expiration: 2005-06 school year WM300-3001 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of all veterans of foreign wars through instructional activities rather than observing a school holiday in their honor.

Livingston County Special Services Unit – Livingston (SD 53/HD 106) / Expiration: 2003-04 school year

**WM300-2911 – Modification of School Code** (Section 24-2). The special education cooperative requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The cooperative failed to hold a public hearing to consider the request.

Pontiac SD 429 – Livingston (SD 53/HD 106) / Expiration: 2007-08 school year WM300-2929 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Rooks Creek CCSD 425 – Livingston (SD 53/HD 106) / Expiration: 2007-08 school year WM300-2922 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Woodland CUSD 5 – Livingston (SD 53/HD 106) / Expiration: 2007-08 school year WM300-2973 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

# SD 54/HD 107

Willow Grove SD 46 – Clinton (SD 54/HD 107) / Expiration: 2007-08 school year WM300-2940 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to hold a public hearing to consider the request.

Willow Grove SD 46 – Clinton (SD 54/HD 107) / Expiration: 2007-08 school year WM300-2970 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 54/HD 108

Dieterich CUSD 30 – Effingham (SD 54/HD 108) / Expiration: 2007-08 school year WM300-2958 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Geff CCSD 14 – Wayne (SD 54/HD 108) / Expiration: 2008-09 school year WM300-3004 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski through instructional activities rather than observing a school holiday in his honor. This modification takes effect in the 2004-05 school year.

Jasper County CUSD 1 – Jasper (SD 54/HD 108) / Expiration: 2007-08 school year WM300-2878 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Wayne City CUSD 100 – Wayne (SD 54/HD 108) / Expiration: 2007-08 school year WM300-2885 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 55/HD 109

Allendale CCSD 17 – Wabash (SD 55/HD 109) / Expiration: 2004-05 school year WM300-2925 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski. The district failed to provide proper notice of the public hearing held to consider the request.

Allendale CCSD 17 – Wabash (SD 55/HD 109) / Expiration: 2004-05 school year WM300-2965 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Cumberland CUSD 77 – Cumberland (SD 55/HD 109) / Expiration: 2007-08 school year WM300-3031 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Marshall CUSD C-2 – Clark (SD 55/HD 109) / Expiration: 2007-08 school year WM300-2955 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Oblong CUSD 4 – Crawford (SD 55/HD 109) / Expiration: 2003-04 school year WM300-3025 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Robinson CUSD 2 – Crawford (SD 55/HD 109) / Expiration: 2007-08 school year WM300-2964 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Robinson CUSD 2 – Crawford (SD 55/HD 109) / Expiration: 2007-08 school year WM300-3045 – Modification of Administrative Rule (23 III. Adm. Code 252.20(b)(1) and (c)(4) and (5)). The district requested to be allowed to offer the classroom and behind-thewheel portions of driver education during the summer for a two-week period, two days a week for three hours a day. The district failed to provide proper notice of the public hearing held to consider the request.

Stewardson-Strasburg CUD 5A – Shelby (SD 55/HD 109) / Expiration: 2007-08 school year WM300-2888 – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district failed to provide proper notice of the public hearing held to consider the request.

Stewardson-Strasburg CUD 5A – Shelby (SD 55/HD 109) / Expiration: 2007-08 school year WM300-2957 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 55/HD 110

Deland-Weldon CUSD 57 – Piatt (SD 55/HD 110) / Expiration: 2007-08 school year WM300-2884 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Monticello CUSD 25 – Piatt (SD 55/HD 110) / Expiration: 2007-08 school year WM300-2943 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Casimir Pulaski and all veterans of foreign wars through instructional activities rather than observing school holidays in their honor.

Oakland CUSD 5 – Coles (SD 55/HD 110) / Expiration: 2007-08 school year WM300-2893 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

#### SD 57/HD 113

St. Clair Safe School - St. Clair (SD 57/HD 113) / Expiration: 2007-08 school year

**WM300-2887 – Modification of School Code** (Section 24-2). The regional office of education requested to be allowed to hold school on the legal holidays honoring Abraham Lincoln and Casimir Pulaski for alternative schools established under Article 13A of the School Code. The regional office failed to provide proper notice of the public hearing held to consider the request.

St. Clair Safe School – St. Clair (SD 57/HD 113) / Expiration: 2007-08 school year WM300-2931 – Modification of School Code (Section 24-2) allows the regional office of education to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor for alternative schools established under Article 13A of the School Code.

### SD 57/HD 114

Cahokia CUSD 187 – St. Clair (SD 57/HD 114) / Expiration: 2003-04 school year WM300-2877 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Mascoutah CUSD 19 – St. Clair (SD 57/HD 114) / Expiration: 2007-08 school year WM300-2892 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Wolf Branch SD 113 – St. Clair (SD 57/HD 114) / Expiration: 2007-08 school year WM300-2982 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

#### SD 58/HD 115

Elverado CUSD 196 – Jackson (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2909 (renewal) – Modification of School Code (Section 24-2). The district requested to be allowed to hold school on the legal holidays honoring Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus. The district failed to provide proper notice of the public hearing held to consider the request.

Elverado CUSD 196 – Jackson (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2949 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Dr. Martin Luther King, Jr., Abraham Lincoln, Casimir Pulaski, and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Lick Creek SD 16 – Union (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2879 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Tamaroa SD 5 – Perry (SD 58/HD 115) / Expiration: 2007-08 school year WM300-2935 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 58/HD 116

Red Brick Alternative School – Monroe, Randolph (SD 58/HD 116) / Expiration: 2006-07 school year

**WM300-2979 (renewal) – Modification of School Code** (Section 24-2). The regional office of education requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln for alternative schools established under Article 13A of the School Code. The regional

office currently has an approved modification for this time period.

Sparta CUSD 140 – Randolph (SD 58/HD 116) / Expiration: 2005-06 school year WM300-2945 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

### SD 59/HD 118

Anna CCSD 37 - Union (SD 59/HD 118) / Expiration: 2007-08 school year

**WM300-2891 (renewal) – Modification of School Code** (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Century CUSD 100 - Pulaski (SD 59/HD 118) / Expiration: 2008-09 school year

**WM100-2961 – Waiver of School Code** (Section 10-20.12a) request to allow the district to charge non-resident pupil tuition in an amount that is less than 100 percent of the preceding year's per capita tuition costs to those students whose parents are employees of the district. If approved, this waiver would take effect in the 2004-05 school year.

Century CUSD 100 – Pulaski (SD 59/HD 118) / Expiration: 2007-08 school year WM300-2962 – Modification of School Code (Section 24-2) allows the district to recognize

the contributions of Casimir Pulaski and Christopher Columbus through instructional activities rather than observing school holidays in their honor.

Joppa-Maple Grove UD 38 – Massac (SD 59/HD 118) / Expiration: 2007-08 school year WM300-2948 – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln through instructional activities rather than observing a school holiday in his honor.

Massac UD 1 – Massac (SD 59/HD 118) / Expiration: 2003-04 school year WM300-3035 (renewal) – Modification of School Code (Section 24-2) allows the district to recognize the contributions of Abraham Lincoln and Casimir Pulaski through instructional activities rather than observing school holidays in their honor.

Vienna HSD 133 – Johnson (SD 59/HD 118) / Expiration: 2007-08 school year

**WM300-2974 (renewal) – Modification of School Code** (Section 27-6) allows the district to permit students in grades 9 through 12 to participate in physical education every day for one semester only for 80 to 85 minutes each session rather than daily due to a 4-block schedule.

Vienna SD 55 – Johnson (SD 59/HD 118) / Expiration: 2006-07 school year

**WM300-2907 (renewal) – Modification of School Code** (Section 24-2). The district requested to be allowed to hold school on the legal holiday honoring Abraham Lincoln. The district currently has an approved modification for this time period.

#### COMMUNICATIONS

OFFICE OF THE SECRETARY OF STATE **JESSE WHITE** – Secretary of State

October 23, 2003

To the Honorable President of the Senate:

Sir:

In compliance with the provisions of the Constitution of the State of Illinois, I am forwarding herewith the enclosed Senate Bill, from the 93<sup>rd</sup> General Assembly as vetoed by the Governor together with his objections.

### SENATE BILL

21	330	618
70	371	633
83	382	1067
100	408	1085
171	460	1101
191	496	1133
192	564	1176
257	594	1353
318	606	1765
	611	1881

Respectfully,

JESSE WHITE Secretary of State

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 25, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby return Senate Bill 21, entitled "AN ACT in relation to vehicles." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 21 contains identical language as Public Act 93-0080, effective July 2, 2003.

For this reason, I hereby veto and return Senate Bill 21.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 14, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Currently, school districts can seek reimbursement from the state for driver's education courses and the law requires six hours of behind the wheel practice. The school district may adopt a policy to permit a proficiency exam after three hours. Senate Bill 70 removes the discretion granted to the school district to craft its own policy. In addition, it will result in an increased financial burden on the school district and the state. During the current fiscal crisis, we should defer to the school district in

crafting its driver's education policy and inquire into the necessity of changing these individual policies before passing such a mandate. Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 70, entitled "AN ACT relating to education."

For this reason, I hereby veto and return Senate Bill 70.

Sincerely, Rod R. Blagojevich Governor

### State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

August 26, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Park districts and forest preserves are central to the life of many communities for recreation, renewal, and quality of life. Managing and maintaining these public areas are central to serving the community. SB 83 proposed one avenue for ensuring that funds are available, but it relies on increasing property taxes outside the limits set through tax caps and without the voter approval process required by them. Our analysis indicates that property taxes on a home valued at \$100,000 would increase by \$7 per year to \$52 per year, depending on the park district. In total, the Cook County Forest Preserve District provisions could cost \$8.0 million per year in property taxes. I do not support property tax increases outside the limits of the tax cap law.

I do, however, support the intent of this legislation and will therefore create a grant program, within the state's bonded capital program, that will allow eligible park districts and forest preserve districts to apply for funds to support capital improvements. The grant program will provide a match to amounts local park districts make available under the terms of the grant and will make grant funds available to eligible forest preserve districts. Grant recipients will have to document the need for the projects and the state funds, demonstrating that no other funds within their existing tax authority will support the proposed projects.

I commend the park districts and forest preserves for their support of our communities. These new grant programs demonstrate my support for the vital role they play.

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 83, entitled "AN ACT concerning taxes."

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

July 24, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 100, entitled "AN ACT concerning compensation of public officials."

Our state's judges work hard, and they are well compensated for their work. They receive good benefits and enjoy good working conditions. In a time when the economy continues to weaken, revenues continue to decline, more and more Illinois citizens find themselves out of work, and the need for resources in critical areas like education, health care and public safety continue to rise. Providing a pay increase to judges that would cost the taxpayers nearly \$4 million is inappropriate, ill timed, and unnecessary.

For this reason, I hereby veto and return Senate Bill 100.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

July 24, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 171, entitled "AN ACT in relation to local government." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 171 contains identical language as Public Act 93-0256, effective July 22, 2002.

For this reason, I hereby veto and return Senate Bill 171.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Senate Bill 191 extends reimbursement for special education services to school districts for children who were DCFS wards, have been adopted, and are still receiving services from DCFS. This is creating a new category of eligibility for full reimbursement of special education costs and is expected to cost the State tens of millions of dollars. Given the fiscal constraints of the State, we cannot justify providing additional reimbursement to school districts outside of the nearly \$145 million increase already approved for FY 04.

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 191, entitled "AN ACT in relation to schools."

Sincerely, Rod R. Blagojevich Governor

### State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Senate Bill 192 requires the State to reimburse school districts for administrative costs of coordinating the special education services for wards of the State who are educated in residential facilities. The administration has been committed to reducing administrative costs wherever possible and given the fiscal constraints the State is facing, we cannot justify an additional \$500,000 to \$1.0 million in reimbursement for administrative costs.

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 192, entitled "AN ACT in relating to education."

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 20, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 257, entitled "AN ACT in relation to deer hunting." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 257 contains identical language as Public Act 93-0554, effective August 20, 2003.

For this reason, I hereby veto and return Senate Bill 257.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706 Rod Blagojevich Governor

August 15, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

This Bill presents several concerns that require further review. Increasing access to the surplus lines market and unauthorized insurers may be an unwise option for consumers and claimants. Even though an insured may be aware of the risk associated with purchasing this insurance, a claimant injured by the insured has not knowingly taken on such a risk. These unauthorized insurers are not regulated by the State of Illinois. Therefore, their financial stability is unknown to our Department of Insurance. Our citizens rely upon insurance being able to pay claims during the worst of times; people are maimed, killed, and injured every day. Insurance is in place to help cover these terrible losses, insurmountable medical bills, and lost wages. It would be imprudent to expand use of these unauthorized insurance companies without further study of its impact. I recommend that the Department of Insurance be allowed additional time to study the potential impact of this bill. Therefore, pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 318, entitled "AN ACT concerning insurance."

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 20, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 330, entitled "AN ACT in relation to vehicles." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 330 contains identical language as Senate Bill 563. On July 29, 2003, I forwarded my recommendation for change to SB 563.

For this reason, I hereby veto and return Senate Bill 330.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 371, entitled "AN ACT in relation to public health." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 371 contains identical language as Public Act 93-0165, effective January 1, 2004.

For this reason, I hereby veto and return Senate Bill 371.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 5, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 382, entitled "AN ACT in relation to criminal law." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 382 contains identical language as Public Act 093-0451, effective August 7, 2003.

For this reason, I hereby veto and return Senate Bill 382.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 25, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 408, entitled "AN ACT concerning sanitary districts."

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

August 19, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 460, entitled "AN ACT concerning health care." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 460 contains identical language as Public Act 93-540, effective August 18, 2003.

For this reason, I hereby veto and return Senate Bill 460.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

July 30, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 496, entitled "AN ACT in relation to taxes." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 496 contains identical language as Public Act 93-0188, effective July 11, 2003.

For this reason, I hereby veto and return Senate Bill 496.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 8, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Though this Bill would extend the repeal dates of three education related funds, no actual appropriation was made for them. The State will pursue alternatives other than these funds to increase State funding for education. Therefore, pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 564, entitled "AN ACT concerning education".

Sincerely, Rod Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

August 25, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

While I support economic development initiatives, I am committed to not raising sales tax. Therefore, pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 594, entitled "AN ACT concerning municipalities."

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 11, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

I cannot at this time sign legislation that will allow for an increase in property taxes that has not been adequately proven to be necessary. Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 606, entitled "AN ACT concerning taxes."

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

July 30, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 611, entitled "AN ACT in electronic mail." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 611 contains identical language as Public Act 93-0199, effective January 1, 2004.

For this reason, I hereby veto and return Senate Bill 611.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

July 30, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 618, entitled "AN ACT regarding education." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 618 contains identical language as Public Act 93-0212, effective July 18, 2003.

For this reason, I hereby veto and return Senate Bill 618.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

July 30, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 633, entitled "AN ACT in relation to aging." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 633 contains identical language as Public Act 93-0246, effective July 22, 2003.

For this reason, I hereby veto and return Senate Bill 633.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 20, 2003

To the Honorable Members of the

Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 1067, entitled "AN ACT concerning senior citizens." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 1067 contains identical language as Public Act 93-0241, effective July 22, 2003.

For this reason, I hereby veto and return Senate Bill 1067.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

July 29, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 1085, entitled "AN ACT in relation to groundwater." While I support the study of the State's groundwater supply as comprehensive and long-term approach to water supply management in Illinois, due to the State's diminishing revenues, we do not have the money available to fund the study at this time

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 19, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 1101, entitled "AN ACT in relation to taxes." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 1101 is substantially similar to Public Act 93-0286, which I signed on July 22, 2003.

For this reason, I hereby veto and return Senate Bill 1101.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

July 29, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 1133, entitled "AN ACT concerning state government." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 1133 contains identical language as Public Act 93-0318, effective January 1, 2004.

For this reason, I hereby veto and return Senate Bill 1133.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

July 29, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 1176, entitled "AN ACT in relation to criminal law." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 1176 contains identical language as Public Act 93-0317, effective January 1, 2004.

For this reason, I hereby veto and return Senate Bill 1176.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 19, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto Senate Bill 1353, entitled "AN ACT in relation to transportation." I recognize and appreciate the hard work of all the sponsors in passing this legislation. However, Senate Bill 1353 contains language identical to Public Act 93-0109, effective July 8, 2003.

For this reason, I hereby veto and return Senate Bill 1353.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Senate Bill 1765 creates another Illinois commission. In order to remain consistent with efforts to reduce the multiple boards and commissions in Illinois, I hereby veto Senate Bill 1765. Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 1765 "AN ACT to create a commission to study tax expenditures in Illinois.

Sincerely, Rod R. Blagojevich

Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 25, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

This legislation promotes an important purpose: providing programs through our park districts and special recreation districts to serve persons with physical or mental disabilities. Yet, this legislation also poses a significant hurdle: increasing property taxes to pay for these programs outside the limits of tax caps and the voter approval process required by them. We have estimated that property taxes could increase by \$16 million in Chicago over several years. The challenge, which we have faced throughout state government, is to fund important and priority services and programs without asking the people of Illinois to pay more in taxes. I remain committed to this principle and cannot support legislation that asks for an increase in property taxes. I do believe that there are other ways these services can be funded, and I am directing the Office of Management and Budget to convene a meeting with representatives of the Special Recreation Associations and others affected to propose alternatives. I know that by working together we can provide these important opportunities for persons with disabilities.

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 1881, entitled "AN ACT concerning taxes."

Sincerely, Rod R. Blagojevich Governor

# OFFICE OF THE SECRETARY OF STATE JESSE WHITE – Secretary of State

October 23, 2003

To the Honorable President of the Senate:

Sir:

In compliance with the provisions of the Consitution of the State of Illinois, I am forwarding herewith the enclosed Senate Bills that are being returned by the Governor with specific recommendations for change.

#### SENATE BILLS

150	640
180	699
196	777
207	902
212	1321
216	1333
272	1364
319	1521
392	1523
472	1754
563	1757
629	1759
639	1804
	1848

Respectfully,

JESSE WHITE Secretary of State

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 25, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 150, entitled "AN ACT in relation to vehicles", with the following specific recommendations for change:

on page 1, line 5, by deleting "and changing Section 29-15"; and

by deleting line 26 on page 2 through line 13 on page 3.

I applaud the sponsors for the many good provisions contained in this bill that seek to improve the safe transportation of children in school districts throughout the state. However, in light of the current fiscal crisis, we simply cannot commit additional state funds to reimburse school districts for transporting children who use public transportation to get to school. Therefore, with the changes outlined above, Senate Bill 150 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 7, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(b) of the Illinois Constitution of 1970, I hereby veto and return Senate Bill 180, entitled "AN ACT concerning records", with the following specific recommendations for change:

on page 1, line 21 by deleting "who was adopted under the"; and

on page 1, by deleting lines 22 and 23; and

on page 1, line 24, by deleting "and"; and

on page 1, line 26, after "Act", by inserting the following: "and who was adopted under the laws of a jurisdiction or country other than the United States by an adopting parent who is a resident of this State".

With these changes, Senate Bill 180 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 22, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 196, entitled "AN ACT in relation to taxes", with the following specific recommendations for change:

on page 1, line 30, after "<u>increase</u>", by inserting "<u>property</u>"; and on page 2, line 1, by replacing "<u>tax rate</u>" with "<u>property tax levy</u>"; and on page 2, line 2, after "<u>more</u>", by inserting "<u>of the park district's</u>"; and on page 2, line 6 after "<u>increase</u>", by inserting "<u>either</u>"; and on page 2, line 6 after "<u>exceed</u>", by inserting "<u>or result in a reduction to</u>".

With these changes, Senate Bill 196 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 207 entitled "AN ACT concerning schools," with the following specific recommendation for change:

on page 2, by replacing lines 19 and 20 with the following:

"(b) The State Board of Education, from any moneys it may have available for this purpose, must implement and administer a grant program that".

Improving student achievement at schools on our academic watch list is a top priority and one we cannot afford to sacrifice. Given the fiscal climate, signing this bill in its current form is not feasible. However, we believe there is significant room for savings in the operations of the State Board of Education. To the extent that savings can be found in the State Board of Education, it would be fitting to apply those resources to the intent of this legislation. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 25, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Quick-take authority should be limited in scope, limited in time, limited to specific property, and confined to a public purpose. This bill contains provisions that have raised many questions as to the appropriate use of quick-take in regards to economic development. The provision regarding Palatine

fails in this regard but I am interested in any subsequent bill regarding Palatine if it is limited to specific property with clearly defined boundaries as opposed to an entire area in a redevelopment district.

In case of the SWIDA provision, each project subject to the quick-take authority must have specific, limited and clearly defined boundaries when it is submitted for approval from the Department of Transportation. Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 212, entitled "AN ACT in relation to local government", with the following specific recommendations for change:

On page 4, line 18, after "2005", by inserting ", subject to the written approval of the Illinois Secretary of Transportation based upon the overall economic development and fiscal plans and needs of the State of Illinois." And

by deleting line 23 on page 4 through line 22 on page 5.

With these changes, Senate Bill 212 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 14, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 216, entitled "AN ACT in relation to civil procedure", with the following specific recommendations for change:

on page 1, line 11, by replacing "<u>for station</u>" with the following: "<u>for the acquisition of rights-of-way and related property necessary for the construction and operation</u>"; and

on page 1, by deleting line 12; and

on page 1, line 13, by deleting "development initiatives in support".

With these changes, Senate Bill 216 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 19, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 272, entitled "AN ACT in relation to vehicles", with the following specific recommendations for change:

on page 5, line 13, by replacing "shall" with "may"; and

on page 5, by replacing line 15 with the following: "If the Commission orders the installation of such temporary stop signs, then the temporary stop signs shall remain in place until the".

With these changes, Senate Bill 272 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

### State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 22, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

While this legislation pursues honorable ends, I cannot sign into law measures that might have a chilling effect on the filing of legitimate complaints of abuse and neglect. Future legislation should be developed to better address the issue of the filing of false or malicious complaints. Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 319 entitled "AN ACT concerning abuse and neglect," with the following specific recommendations for change:

by deleting line 33 on page 2 through line 4 on page 3; and

on page 4, lines 3 and 4, by deleting "Nothing in this Section requires an investigation to include an onsite visit."; and

by replacing line 23 on page 4 through line 4 on page 5 with the following:

"Act has been or is being violated. The Department shall investigate all complaints alleging abuse or neglect within 7 days after the receipt of the complaint except that complaints of abuse or neglect which indicate that a resident's life or safety is in imminent danger shall be investigated within 24 hours after receipt of the complaint. All other"; and

on page 7, line 2, by replacing "2, 4," with "2"; and

by deleting line 6 on page 11 through line 19 on page 13, and

by deleting all of pages 14 through 17.

With these changes, Senate Bill 319 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 15, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 392, entitled "AN ACT in relation to taxes", with the following specific recommendation for change:

on page 1, line 30, by replacing "and \$750,000" with "and (notwithstanding any changes made to this subsection (c) by Public Act 93-32) \$750,000".

With this change, Senate Bill 392 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

July 29, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

This landmark legislation will provide vital reforms to the criminal justice system as it relates to capital punishment. The legislation overwhelmingly achieves these ends and I support all of its provisions except that provision assigning the decertification of police officers to the law enforcement training board. It will have the unintended and adverse impact of placing new constraints on the effective administration of justice, while denying due process rights to police officers.

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 472 entitled "AN ACT in relation to criminal law," with the following specific recommendation for change:

By deleting all lines between and including page 4, line 17, through page 11, line 25.

With this change, Senate Bill 472 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

July 29, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 563 entitled "AN ACT in relation to vehicles," with the following specific recommendation for change:

Among other things, Senate Bill 563 allows tow truck operators to have triple rear axle weights of 56,000 pounds on highways under control of State and local authorities. This is a significantly large weight that creates the potential for damage to highways throughout Illinois. While the Department has the resources to assess and protect against damage to its bridges and structures, local highway authorities do not have the same abilities.

In order to protect local highway infrastructure, I make the following specific recommendations for change:

on page 6, line 11, by replacing ", and" with "and"; and

on page 6, lines 12 and 13, by deleting " or 56,000 pounds on manufactured recovery units with a triple rear axle,"; and

on page 11, line 10, by replacing "axle or" with "axle, or"; and

on page 11, line 10, by inserting the following before the period: "or 56,000 pounds on manufactured recovery units with a triple rear axle".

With these specific recommendations for change, SB 563 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 11, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 629, entitled "AN ACT concerning prisons", with the following specific recommendations for change:

On page 2, by replacing lines 17 through 21 with the following: "from sales from commissary stores must be deposited into the General Revenue Fund."; and

by replacing line 33 on page 2 through line 6 on page 3 with the following: "non-tobacco products."; and

on page 3, line 7, by deleting "agreement".

With these changes, Senate Bill 629 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

August 25, 2003

To the Honorable Members of the Illinois Senate 93rd General Assembly

I appreciate the burden placed on sheriffs when a major mental health facility is closed. This amendatory veto is drafted to help alleviate the impact of that burden. Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 639, entitled "AN ACT concerning mental health", with the following specific recommendations for change:

on page 1, by replacing lines 11 through 25 with the following:

"take a respondent into custody and transport him to a mental health facility, or may make arrangements with another public or private entity including a licensed ambulance service to appropriately transport the respondent to the mental health facility. In the event it is determined by such facility that the respondent is in need of commitment or treatment at another mental health facility, the county sheriff shall transport the respondent to the appropriate mental health facility, or the county sheriff may make arrangements with another public or private entity including a licensed ambulance service to appropriately transport the respondent to the mental health facility. If, however, the respondent is found in any county where an entire State-operated mental health facility, operated by the Department, was closed between August 2002 and September 2002, then upon receipt of a petition and certification prepared pursuant to this Article, the county sheriff of the county in which the respondent is found shall take the respondent into custody and notify the Department. Thereafter, the Department shall make arrangements either directly or through agreements with other public or private entities to appropriately transport the respondent to the mental health facility. If it is determined that the respondent in such a county is in need of commitment or treatment at another mental health facility, the Department shall make arrangements either directly or through agreements with other public or private entities to appropriately transport the respondent to the mental health facility."; and

by replacing line 32 on page 2 through line 4 on page 3 with the following:

"such person is able to do so safely and humanely. When the Department indicates that it has transportation to the facility available, the order may authorize the Department to transport the recipient there. The court may order the sheriff of the county in which such proceedings are held to transport the recipient to the facility. If, however, the respondent is found in any county where an entire State-operated mental health facility, operated by the Department, was closed between August 2002 and September 2002, then upon receipt of a petition and certification prepared pursuant to this Article, the county sheriff of the county in which the respondent is found shall take the respondent into custody and notify the Department. Thereafter, the Department shall make arrangements either directly or through agreements with other public or private entities to appropriately transport the respondent to the mental

health facility. If it is determined that the respondent in such a county is in need of commitment or treatment at another mental health facility, the Department shall make arrangements either directly or through agreements with other public or private entities to appropriately transport the respondent to the mental health facility. When a"; and

on page 3, by replacing lines 11 and 12 with the following:

"or the Department is authorized to transport the recipient between facilities, or whether the <u>Department</u> or the county sheriff"; and

on page 3, by replacing lines 14 and 15 with the following:

"facilities. The party determined by the court to be responsible for the recipient's transportation sheriff may make arrangements with"; and

on page 5, by replacing lines 13 through 16 with the following:

"placement process, the sheriff shall be notified and shall appropriately transport the defendant to the designated facility. If, however, the defendant is located in any county where an entire State-operated mental health facility, operated by the Department of Human Services, was closed between August 2002 and September 2002, the Department of Human Services shall make arrangements either directly or through agreements with other public or private entities to appropriately transport the defendant to the designated"; and

on page 7, by replacing lines 17 through 20 with the following:

"Upon completion of the placement process the sheriff shall be notified and shall <u>transport the defendant</u> to the designated facility. If, however, the defendant is located in any county where an entire State-operated mental health facility, operated by the Department of Human Services, was closed between August 2002 and September 2002, the Department of Human Services shall make arrangements either directly or through agreements with other public or private entities to appropriately transport the".

With these changes, Senate Bill 639 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 25, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Quick-take authority should be limited in scope, limited in time, limited to specific property, and confined to a public purpose. The provision regarding Morton Grove has raised many questions as to the appropriate use of quick-take in regards to economic development. The provision regarding Morton Grove fails in this regard but I am interested in any subsequent bill regarding Morton Grove if it is limited to specific property with clearly defined boundaries as opposed to an entire area in a redevelopment district. Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby

return Senate Bill 640, entitled "AN ACT in relation to real property", with the following specific recommendations for change:

on page 1, line 6, by deleting "7-103.106".

by deleting page 3, line 24 through page 8, line 22.

With these changes, Senate Bill 640 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

### State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

September 3, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 699, entitled "AN ACT concerning electronic transmissions", with the following specific recommendations for change:

on page 2, by replacing lines 16 through18 with "conference, or other electronic means, all locations at which members of the public may attend must be disclosed in the agenda. An agenda for each regular meeting shall be"; and

on page 5, line 22, by deleting "special, emergency,"; and

on page 5, by replacing lines 26 and 27 with "place designated in the notice of the meeting, a public body may allow a member of that body to attend the meeting by electronic means if the"; and

on page 6, line 16, after "law", by inserting "other than a hearing conducted under Article 9 or 10 of the Election Code".

Several of these changes are required to give the necessary flexibility to the State Board of Elections to conduct their electoral board meetings and campaign disclosure hearings within the limited timeframe required by the Election Code. With these changes, Senate Bill 699 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

September 3, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

This groundbreaking legislation will make available to Illinois' citizens medical equipment proven to save lives. I support the intent of the legislation; however, there are several changes that I am recommending in order to ensure that the intent is fulfilled.

The language I am recommending provides a clearer definition of physical fitness facilities required to comply; the current bill would apply to venues beyond the intended scope, like churches and wellness centers. The recommended language also provides that these protections are available to citizens living in home rule units, like the City of Chicago. The recommended language provides for stronger enforcement by increasing the penalties for violations in an effort to encourage compliance.

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 777, entitled "AN ACT in relation to health, which may be known as the Colleen O'Sullivan Law", with the following specific recommendations for change:

on page 2, line 1, after "government", by inserting ", including a home rule unit,"; and

on page 2, by replacing lines 15 through 20 with the following:

"provides services or facilities focusing primarily on cardiovascular exertion as defined by Department rule.

- (b) "Physical fitness facility" does not include a facility serving less than a total of 100 individuals, as further defined by Department rule. In addition, the term does not include a"; and
- on page 2, by replacing line 26 with the following:
  - "(a) Before January 1, 2005, each person or entity, including a home rule unit, that"; and

on page 3, line 4, after "entity", by inserting ", including a home rule unit,"; and

on page 4, by deleting lines 1 through 12.

on page 4, by replacing lines 24 through 32 with the following:

"subsection (a) or (b) of Section 15, the Director may issue to the facility a written administrative warning without monetary penalty for the initial violation. The facility may reply to the Department with written comments concerning the facility's remedial response to the warning. For subsequent violations, the Director may impose a civil monetary penalty against the facility as follows:

- (1) At least \$1,500 but less than \$2,000 for a second violation.
- (2) At least \$2,000 for a third or subsequent violation.
- (b) The Director may impose a civil monetary penalty under this"; and

on page 5, below line 16, by inserting the following:

"(d) The fines shall be deposited into the Physical Fitness Facility Medical Emergency Preparedness Fund to be appropriated to the Department, together with any other amounts, for the costs of administering this Act."; and

on page 6, after line 20 by adding the following:

Section 55. Home rule. A home rule unit must comply with the requirements of this Act. A home rule unit may not regulate physical fitness facilities in a manner inconsistent with this Act. This Section is a limitation under subsection (i) of Section 6 of Article VII of the Illinois Constitution on the concurrent exercise by home rule units of powers and functions exercised by the State.

Section 75. The State Finance Act is amended by adding Section 5.595 as follows:

(30 ILCS 105/5.595 new)

Sec. 5.595. The Physical Fitness Facility Medical Emergency Preparedness Fund."; and

by deleting page 6, line 28 through page 46, line 6.

With these changes, Senate Bill 777 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 11, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 902, entitled "AN ACT relating to schools", with the following specific recommendations for change:

on page 1, by replacing lines 18 through 26 with the following: "shall provide that those schools that are on the State Board of Education Early Academic Warning List or the academic watch list under Section 2-3.25d that maintain grades kindergarten through 3 shall be given priority."; and

by replacing lines 30 and 31 on page 1 and lines 1 through 4 on page 2 with the following: "class size of no more than 20 pupils."

These changes are necessary to give flexibility to the State Board of Education to award grants to the most disadvantaged schools, based on a variety of factors. With these changes, Senate Bill 902 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

I am firmly committed to funding elementary and secondary education. The fiscal year 2004 budget provides over \$381 million new dollars directly to school districts, including a \$250 increase in the per student Foundation Level. SB 1321 targets additional resources to school districts affected by tax caps and struggling to make up for lost resources through property tax appeals. While we have provided millions of new dollars to school districts, we must recognize our budget constraints, and therefore, I am modifying SB 1321 to limit the allocation for state aid adjustments to \$20 million of the General State Aid appropriation within each fiscal year. This amount is consistent with the average allocated in the past four years. Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1321, entitled "AN ACT regarding schools", with the following specific recommendations for change:

on page 2, line 4, after "claim.", by inserting "From the total amount of general State aid to be provided to districts, adjustments as a result of recomputation under this Section together with adjustments under Section 2-3.84 shall not exceed \$20 million, in the aggregate for all districts under both Sections combined, of the general State aid appropriation in any fiscal year; if necessary, amounts shall be prorated among districts."; and

on page 2, line 16, after "calculation.", by inserting "From the total amount of general State aid to be provided to districts, adjustments under this Section together with adjustments as a result of recomputation under Section 2-3.33 shall not exceed \$20 million, in the aggregate for all districts under both Sections combined, of the general State aid appropriation in any fiscal year; if necessary, amounts shall be prorated among districts.".

With these changes, Senate Bill 1321 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

August 15, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

am committed to funding education for our school children, including special education, and the FY 04 budget reflects that commitment. While we were not able to fund every program, we significantly increased funding for education (by \$381 million for fiscal year 2004) and for special education programs overall (by 145 million for fiscal year 2004). In these difficult fiscal times, I believe that we must operate within the constraints set by that budget and not allow costs to be passed on to future years. Therefore, I am modifying SB 1333 so that at least for the next two years reimbursements for these special education costs must be paid solely from the year they are appropriated.

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1333, entitled "AN ACT concerning education", with the following specific recommendations for change:

on page 4, by replacing line 17 with the following:

"year thereafter through fiscal year 2002 and then in fiscal year 2006 and each fiscal year thereafter, if the amount"; and

on page 4, by replacing line 22 with the following:

"the next fiscal year, and the payments required to eliminate"; and

on page 4, by replacing lines 25 through 32 with the following:

"Notwithstanding any other provision of this Section or this Code, beginning with fiscal year 2003 and continuing through fiscal year 2005, total reimbursement under this Section in any fiscal year is limited to the amount appropriated for that purpose for that fiscal year, and if the amount appropriated for any fiscal year is less than the amount required for purposes of this Section, the insufficiency shall be apportioned pro rata among the school districts seeking reimbursement."; and

on page 10, by replacing lines 15 and 16 with the following:

"1994 and each fiscal year thereafter through fiscal year 2002 and then in fiscal year 2006 and each fiscal year thereafter, if the amount appropriated for any fiscal year is less"; and

on page 10, by replacing lines 20 and 21 with the following:

"reimbursed on August 30 of the next fiscal year, and the payments required to eliminate any insufficiency for prior"; and

on page 10, by replacing lines 23 through 31 with the following:

"for the current fiscal year. Notwithstanding any other provision of this Section or this Code, beginning with fiscal year 2003 and continuing through fiscal year 2005, total reimbursement under this Section in any fiscal year is limited to the amount appropriated for that purpose for that fiscal year, and if the amount appropriated for any fiscal year is less than the amount required for purposes of this Section, the insufficiency shall be apportioned pro rata among the school districts seeking reimbursement."

With these changes, Senate Bill 1333 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

September 3, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1364, entitled "AN ACT in relation to public aid", with the following specific recommendation for change:

on page 1, line 10, by replacing "disparate impact" with "applicability".

While I fully support the bill's intentions in making information about the TANF program available for distribution, the term "disparate impact" is a legal term of art that connotes more than the mere distribution of this information. With this change, Senate Bill 1364 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

August 15, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1521 entitled "AN ACT concerning recreational trails," with the following specific recommendation for change:

on page 3, by replacing lines 3 and 4 with the following:

"(3) Construction of motorized recreational trails on Department owned or managed land, except recreation areas acquired after the effective date of this amendatory Act of the 93rd General Assembly."

With this change, Senate Bill 1521 will have my approval. I respectfully request your concurrence

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1523, entitled "AN ACT concerning the Deaf and Hard of Hearing Commission", with the following specific recommendations for change:

on page 3, by deleting line 33; and

on page 3, line 34 by deleting "assure"; and

on page 4, by replacing lines 7 through 14 with the following:

"(5) Monitor State funded programs delivering services to persons who are deaf or hard of hearing to determine the extent that promised and mandated services are delivered."; and

on page 4, by replacing lines 25 through 33 with the following:

"(7) Promote cooperation among State and local agencies providing educational programs for deaf and hard of hearing individuals.".

With these changes, Senate Bill 1523 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich Governor

August 18, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1754, entitled "AN ACT creating the Western Illinois Economic Development Authority", with the following specific recommendations for change:

on page 6, line 2, by replacing "23" with "17"; and

on page 6, by replacing lines 9 through 14 with the following:

"(2) Public members. Fifteen members shall be appointed by the Governor with the advice and consent of the Senate. Each county within the territorial jurisdiction of the Authority shall be represented by at least one member. All"; and

on page 6, line 24, by replacing "Twelve" with "Eight"; and

on page 6, by replacing lines 26 and 27 with the following: "annually by the Board."; and

on page 6, by replacing line 30 with the following:

"Of the 15 original public members appointed pursuant to subsection (b), 3 shall serve until the third Monday in January, 2005; 3 shall serve until the third Monday in January, 2006; 3 shall serve until the third Monday in January, 2008; and 3 shall serve until the third Monday in January, 2008; and 3 shall serve until the third Monday in January, 2009. All successors to these original public members shall be appointed by the Governor with the advice and consent of the Senate, pursuant to subsection (b), and shall hold office for a term of 6 years commencing the third Monday in January of the year in which their term commences, except in the case of an appointment to fill a vacancy. Vacancies occurring among the public members shall be filled"; and

on page 7, by replacing lines 10 through 13 with the following: "malfeasance in office."; and

by replacing lines 31 through 34 on page 7 and lines 1 through 29 on page 8 with the following: "the Illinois Finance Authority, attorneys, appraisers, engineers, accountants, credit analysts, and other consultants if the Western Illinois Economic Development Authority deems it advisable.

Section 25. Duty. All official acts of the Authority shall require the approval of at least 8 members. It shall"; and

on page 9, by replacing line 13 with the following

"(3) to utilize services of the Illinois Finance Authority necessary to".

With these changes, Senate Bill 1754 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 12, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1757, entitled "AN ACT concerning budget stabilization", with the following specific recommendations for change:

on page 1, line 22, after "appropriations", by inserting ", and transfers or diversions as required by law,"; and

on page 1, line 27, after "appropriations", by inserting ", and transfers or diversions as required by law,";

on page 2, between lines 2 and 3, by inserting the following:

"(c) For purposes of this Act, "estimated general funds revenues" are defined as the total amount of revenues as presented in the Governor's State budget pursuant to Section 50-5 of the State Budget Law of the Civil Administrative Code of Illinois. These estimated general funds revenues include, for each budget year, all taxes, fees, and other revenues expected to be deposited into the State's general funds, including recurring transfers from other State funds into the general funds.

Year-over-year comparisons used to determine the percentage growth factor of estimated general funds revenues shall exclude the sum of the following: (i) expected revenues resulting from new taxes or fees or from tax or fee increases during the first year of the change, (ii) expected revenues resulting from one-time receipts or non-recurring transfers in, (iii) expected proceeds resulting from borrowing, and (iv) increases in federal grants that must be completely appropriated based on the terms of the grants."; and

on page 2, line 3, by replacing "and" with "."; and

on page 2, by replacing line 4 with the following:

"In furtherance of the State's objective for the Budget Stabilization Fund to have resources representing 5% of the State's annual general funds revenues:"; and

on page 2, line 6, after "appropriations", by inserting ", and transfers or diversions as required by law,"; and

on page 2, by replacing lines 11 and 12 with "to the Budget Stabilization Fund."; and

on page 2, line 14, after "appropriations", by inserting ", and transfers or diversions as required by law,"; and

on page 2, by replacing lines 19 and 20 with "the Budget Stabilization Fund."; and

on page 2, line 26, by replacing "4%" with "5%"; and

on page 2, line 27, after "year", by inserting "except as provided by subsection (d) of this Section"; and

on page 2, line 29, by replacing "4%" with "5%"; and

by replacing line 30 on page 2 through line 4 on page 3 with the following:

"that fiscal year, then additional transfers are not required unless there are outstanding liabilities under Section 25 of the State Finance Act from prior fiscal years. If there are such outstanding Section 25 liabilities, then the Comptroller shall continue to transfer 1/12 of the total amount identified for transfer to the Budget Stabilization Fund on the first day of each month of that fiscal year or as soon thereafter as possible to be reserved for those Section 25 liabilities. Nothing in this Act prohibits the General Assembly from appropriating additional moneys into the Budget Stabilization Fund.

(e) On or before August 31 of each fiscal year, the amount determined to be transferred to the Budget Stabilization Fund shall be reconciled to actual general funds revenues for the fiscal year. The final transfer for each fiscal year shall be adjusted so that the amount transferred is equal to the percentage specified in subsection (a) or (b) of Section 10 of this Act, as applicable, based on actual general funds revenues calculated consistently with subsection (c) of Section 10 of this Act for each fiscal year."; and

on page 3, by replacing line 6 with "Section 6z-51 as"; and

on page 3, by deleting lines 8 and 9; and

on page 3, line 20, after "meet", by inserting "cash flow"; and

by deleting lines 26 through 30 on page 3 and deleting lines 1 through 4 on page 4.

With these changes, Senate Bill 1757 will have my approval. I respectfully request your concurrence.

Sincerely, Rod Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 12, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1759, entitled "AN ACT concerning truth in budgeting", with the following specific recommendations for change:

on page 1, by replacing line 6 with the following: "Section 50-10 as follows:"; and

by deleting lines 7 through 31 on page 1, by deleting page 2, and by deleting lines 1 through 17 on page 3; and

on page 4, by replacing lines 10 through 15 with the following:

"For the fiscal year beginning July 1, 2005 and for each fiscal year thereafter, the budget shall identify liabilities incurred in a prior fiscal year under Section 25 of the State Finance Act and provide funding as allowable pursuant to Section 15 (d) of the Budget Stabilization Act, if applicable."; and

by deleting lines 17 through 32 on page 5, and by deleting pages 6, 7, 8, and 9.

With these changes, Senate Bill 1759 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 18, 2003

To the Honorable Members of the Illinois Senate

93rd General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1804, entitled "AN ACT concerning recreational trails", with the following specific recommendation for change:

on page 2, line 20, by replacing "85%" with "70%".

With this change, Senate Bill 1804 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

Rod Blagojevich GOVERNOR

August 19, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

Pursuant to Article IV, Section 9(e) of the Illinois Constitution of 1970, I hereby return Senate Bill 1848, entitled "AN ACT in relation to the Illinois State Toll Highway Authority", with the following specific recommendations for change:

on page 1, line 7, by deleting "9.65,"; and

on page 5, by deleting lines 21 through 26.

With these changes, Senate Bill 1848 will have my approval. I respectfully request your concurrence.

Sincerely, Rod R. Blagojevich Governor

# OFFICE OF THE SECRETARY OF STATE **JESSE WHITE** – Secretary of State

October 23, 2003

To the Honorable President of the Senate:

Sir:

In compliance with the provisions of the Constitution of the State of Illinois, I am forwarding herewith the enclosed Senate Bill which the Governor vetoed in part (Line Item Vetoes) and reduced.

SENATE BILL 1239 PUBLIC ACT 93-0587

Respectfully,

JESSE WHITE Secretary of State

# State of Illinois OFFICE OF THE GOVERNOR 207 State Capitol, Springfield, Illinois 62706

August 22, 2003

To the Honorable Members of the Illinois Senate 93<sup>rd</sup> General Assembly

This bill, while extremely important, is technically problematic given the multiple years of appropriations and funding sources and other issues. After exhaustive review, the most prudent option seems to be to sign the majority of the bill. The Administration will only release certain projects, over time, rather than trying to address all of the bill's issues at the outset through amendatory vetoes.

The Administration's intent with regard to member projects is to release a total of \$450 million as indicated in my budget released on April 9, including spending on projects that have already been released by this Administration. The total amount of member initiative projects listed in the bill is \$907,892,320. However, the bill contains many duplicates of member initiatives, meaning some items appear more than once in the bill. Obviously, projects will only be funded once, so that the total amount of spending authority in the bill for member projects is actually \$657,153,623. Only \$450 million total will be spent on projects released by this Administration.

The total amount of item vetoes and reductions totals \$555,806.937. The items vetoed and reduced are:

- \$54,844,803 for land acquisition at the Capitol Complex. In these fiscally difficult times, and
  at a time that the state has achieved a decrease in its workforce, it is not appropriate for the
  State to be expanding its property holdings for administrative purposes at this large a scale.
- \$47,514,810 in additional capital vetoes and reductions, primarily of repair and renovation projects on state facilities that, while worthy, can be put off until the state's fiscal situation improves, or that it has been deemed that the state can do without.
- \$35,000,000 of appropriations to transfer funding from the Anti-Pollution Fund to the Clean Water Trust Fund. Since legislation to create a program for expenditure of these funds, and bond authorization for this purpose, have not yet passed the legislature, this item will be held till it can be dealt with as a package.
- \$17,266,562 in projects that have completely spent out by the end of Fiscal Year 2003 and are completed.
- \$401,180,762 in technical writedowns to reflect the actual amount available on June 30 after all Fiscal Year 2003 spending.

Pursuant to Article IV, Section 9(d) of the Illinois Constitution of 1970, I hereby veto and return several appropriation items included in Senate Bill 1239, entitled "AN ACT making appropriations", having taken the actions set forth below.

### Item Vetoes

I hereby veto the following appropriations items:

Article		Section	Page	Lines	Amount Enacted	
	1	3	2	12-15	\$	440,000.00
	1	4	2	31 and	l	

3 1-3 \$ 150,000.00

	1		5	1-3	Ψ	130,000.00
	1	4	3	23-25	\$	1,640,000.00
	1	4	3	30-31	\$	290,000.00
	1	4	4	2-4	\$	1,660,000.00
	1	5	4	23-25	\$	180,000.00
	1	5	5	30-31	\$	720,000.00
	1	14	11	7-12	\$	35,000,000.00
	2	4	17	1	\$	1,100,000.00
	2	4.1	19	8-22	\$	54,844,803.00
	2	5	22	26-29	\$	3,500,000.00
	2	5	23	32-34	and	
	2	5	24	1	\$	61,141.00
Article		Section	Page	Lines		Amount Enacted
711 tiele	2	5	27	14	\$	200,000.00
	2	5a	33	29-31	\$	10,000.00
	2	6	40	15-17	\$	15,711.00
	2	6	41	4	\$	1,730,000.00
	2	6	41	11	\$	1,255,605.00
	2	7	51		and	-,,
	2	7	52	1-2	\$	750,000.00
	2	7	52	22-23	\$	30,447.00
	2	7a	54	13-16	\$	400,000.00
	2	8	56	7-9	\$	3,681,000.00
	2	8	56	15-17	\$	3,350,612.00
	2	8	56	18-20	\$	41,510.00
	2	8	56	21-24	\$	2,062,047.00
	2	8	56	25-27	\$	193,055.00
	2	8	56	28-29	\$	237,723.00
	2	8	59	26	\$	32,913.00
	2	8	60	26-27	\$	27,563.00
	2	8	62	2	\$	150,000.00
	2	8.3	66	1-3	\$	32,227.00
	2	15	78	30-31	and	
	2	15	79	1-2	\$	36,199.00
	2	27	92	26-30	\$	5,943,800.00
	2	30	104	31	\$	67,523.00
	4	50	113	10-18	\$	108,148.00
	5	12	118	9-10	\$	1,200,000.00
	5	32	150	17-24	\$	3,554.00
	5	6-5.44b	154	8-16	\$	8,192.00
	6	75	161	8-15	\$	295,960.00
	6	175	166	23-31	\$	800,000.00
						[October 23, 2003]

	6 320		174	23-31	\$	1,800,000.00
Article		Section	Page	Lines	A	amount Enacted
	6	490	183	19-26	\$	28,510.00
	6	515	184	28-32 and		
	6	515	185	1-3	\$	225,000.00
	6	965	208	17-25	\$	12,800.00
	6	985	209	20-28	\$	50,000.00
	6	1010	210	29-32 and		
	6	1010	211	1-5	\$	10,000.00
	6	1015	211	6-13	\$	2,000.00
	6	1065	213	27-32 and		
	6	1065	214	1-2	\$	125,000.00
	6	1075	214	11-18	\$	25,000.00
	6	1135	217	18-25	\$	883,300.00
	6	1150	218	9-17	\$	2,421,374.00
	6	1190	221	6-14	\$	414,000.00
	6	1255	225	4-11	\$	150,000.00
	6	1270	225	32 and		
	6	1270	226	1-7	\$	2,000,000.00
	6	1280	226	16-22	\$	250,000.00
	6	1315	228	26-33	\$	2,000,000.00
	6	1355	230	23-30	\$	1,500,000.00
	6	1425	235	6-13	\$	400,000.00
	6	1480	238	10-18	\$	450,000.00
Article		Section	Page	Lines	A	amount Enacted
	6	1570	243	1-9	\$	1,000,000.00
	6	1610	245	8-16	\$	1,000,000.00

# **Reduction Vetoes**

I hereby reduce the following appropriation items and approve each item in the amount set forth in the "Reduced Amount" column below:

Article	Section	Page	Lines	Amou	int Enacted	Amount	t Reduced
1	4	3	14	\$	7,540,000.00	\$	5,540,000.00
2	1	12	4	\$	1,250,000.00	\$	1,203,030.00
2	1	12	7	\$	1,664,966.00	\$	1,577,210.00
2	1	12	9	\$	255,385.00	\$	17,536.00
2	1	12	14	\$	168,208.00	\$	143,100.00
2	1	12	15	\$	23,038.00	\$	12,890.00
2	1	12	17	\$	7,332,500.00	\$	4,880,850.00
2	1	12	24	\$	1,027,457.00	\$	977,309.00
2	1	12	27	\$	1,077,657.00	\$	1,051,648.00

2	1	12	28	\$	213,031.00	\$	79,714.00
2	1	12	29	\$	1,097,936.00	\$	1,072,556.00
2	1	12	31	\$	1,183,570.00	\$	539,289.00
2	1	12	32	\$	206,493.00	\$	162,533.00
2	1	12	33	\$	284,666.00	\$	192,398.00
2	1	13	1	\$	72,458.00	\$	69,202.00
2	2	13	28	\$	923,883.00	\$	
2	2		1	\$	ŕ		440,468.00
		14			537,678.00	\$	469,036.00
2	2	14	5	\$	170,000.00	\$	24,846.00
2	2	14	7	\$	23,725.00	\$	23,575.00
2	2	14	16	\$	465,906.00	\$	307,810.00
2	2a	14	32	\$	175,726.00	\$	172,679.00
2	3	15	18	\$	2,146,217.00	\$	1,806,499.00
2	3	15	22	\$	39,680.00	\$	31,784.00
2	3	16	12	\$	678,824.00	\$	123,634.00
2	4	16	32	\$	1,290,000.00	\$	190,000.00
2	4	17	10	\$	260,000.00	\$	250,954.00
2	4	17	16	\$	989,092.00	\$	931,295.00
2	4	17	17	\$	46,357.00	\$	35,067.00
2	4	17	23	\$	439,354.00	\$	372,948.00
2	4	17	25	\$	150,563.00	\$	49,793.00
2	4	17	28	\$	109,110.00	\$	62,001.00
Article	Section	Page	Lines	Amou	nt Enacted	Amou	int Reduced
2	4	18	1	\$	538,917.00	\$	436,295.00
2	4	18	9	\$	1,903,119.00	\$	1,441,000.00
2	4	18	16	\$	1,174,544.00	\$	1,152,669.00
2	4	18	24	\$	166,686.00	\$	30,408.00
2	4	18	27	\$	922,322.00	\$	795,154.00
2	4	18	32	\$	107,872.00	\$	43,499.00
2	4	19	6	\$	102,803.00	\$	67,470.00
2	4a	20	14	\$	50,100.00	\$	31,700.00
2	4a	20	17	\$	1,009,000.00	\$	923,544.00
2	5	21	3	\$	2,611,237.00	\$	2,610,485.00
2	5	21	9	\$	498,487.00	\$	362,650.00
2	5	21	13	\$	1,565,614.00	\$	1,079,272.00
2	5	21	16	\$	1,161,424.00	\$	1,121,801.00
2	5	21	18	\$	111,026.00	\$	50,887.00
2	5	21	22	\$	1,725,265.00	\$	1,182,795.00
2	5	21	25	\$	402,499.00	\$	125,505.00
					-		

2	5	21	27	\$ 2,596,923.00	\$ 1,120,522.00
2	5	22	4	\$ 900,000.00	\$ 883,500.00
2	5	22	13	\$ 842,905.00	\$ 842,605.00
2	5	22	20	\$ 383,807.00	\$ 378,381.00
2	5	23	1	\$ 788,695.00	\$ 267,509.00
2	5	23	3	\$ 325,099.00	\$ 195,693.00
2	5	23	6	\$ 258,521.00	\$ 177,501.00
2	5	23	9	\$ 400,606.00	\$ 57,219.00
2	5	23	11	\$ 290,640.00	\$ 145,238.00
2	5	23	13	\$ 92,519.00	\$ 6,486.00
2	5	23	15	\$ 48,529.00	\$ 46,729.00
2	5	23	19	\$ 19,533.00	\$ 15,889.00
2	5	23	27	\$ 272,971.00	\$ 199,308.00
2	5	23	31	\$ 45,526.00	\$ 24,417.00
2	5	24	7	\$ 536,973.00	\$ 467,192.00
2	5	24	13	\$ 814,351.00	\$ 765,500.00
2	5	24	23	\$ 76,570.00	\$ 50,391.00
2	5	24	27	\$ 157,921.00	\$ 150,735.00
2	5	24	30	\$ 1,048,647.00	\$ 997,592.00
2	5	25	5	\$ 116,226.00	\$ 48,685.00
2	5	25	8	\$ 129,235.00	\$ 22,548.00
2	5	25	20	\$ 100,000.00	\$ 92,100.00
2	5	25	23	\$ 96,258.00	\$ 61,411.00
2	5	25	26	\$ 208,431.00	\$ 145,245.00
2	5	25	29	\$ 827,292.00	\$ 533,853.00
2	5	25	32	\$ 483,869.00	\$ 371,443.00
Article	Section	Page	Lines	int Enacted	nt Reduced
2	5	26	3	\$ 1,355,915.00	\$ 684,534.00
2	5	26	6	\$ 41,129.00	\$ 35,144.00
2	5	26	22	\$ 1,000,000.00	\$ 930,507.00
2	5	26	26	\$ 6,280,841.00	\$ 6,143,432.00
2	5	26	28	\$ 384,497.00	\$ 265,215.00
2	5	27	17	\$ 412,163.00	\$ 398,794.00
2	5	27	23	\$ 696,913.00	\$ 667,141.00
2	5	27	28	\$ 345,966.00	\$ 305,847.00
2	5	29	29	\$ 136,345.00	\$ 106,610.00
2	5	30	1	\$ 148,691.00	\$ 94,226.00
2	5	30	5	\$ 38,308.00	\$ 20,243.00
2	5	30	11	\$ 283,905.00	\$ 231,596.00

2	5	31	24	\$ 274,839.00	\$ 274,539.00
2	5	31	26	\$ 65,702.00	\$ 48,763.00
2	5	31	33	\$ 1,880,795.00	\$ 1,877,225.00
2	5.2	32	22	\$ 30,000.00	\$ 25,007.00
2	5.2	32	25	\$ 40,191.00	\$ 17,198.00
2	5a	33	18	\$ 119,707.00	\$ 76,950.00
2	5a	33	20	\$ 43,760.00	\$ 43,335.00
2	5a	33	22	\$ 400,805.00	\$ 283,970.00
2	5a	33	26	\$ 318,664.00	\$ 17,359.00
2	5a	33	28	\$ 178,669.00	\$ 101,560.00
2	5a	34	1	\$ 394,965.00	\$ 394,351.00
2	5a	34	3	\$ 94,747.00	\$ 33,162.00
2	5a	34	9	\$ 241,632.00	\$ 71,184.00
2	5a	34	16	\$ 60,284.00	\$ 8,732.00
2	5a	34	19	\$ 50,000.00	\$ 42,600.00
2	5a	34	21	\$ 247,104.00	\$ 188,499.00
2	6	35	4	\$ 108,512.00	\$ 101,567.00
2	6	35	5	\$ 1,040,247.00	\$ 418,103.00
2	6	35	14	\$ 936,720.00	\$ 747,443.00
2	6	35	16	\$ 402,551.00	\$ 207,056.00
2	6	35	18	\$ 2,222,390.00	\$ 1,289,733.00
2	6	35	26	\$ 10,248.00	\$ 3,633.00
2	6	35	33	\$ 536,349.00	\$ 525,010.00
2	6	36	2	\$ 273,457.00	\$ 268,189.00
2	6	36	4	\$ 742,352.00	\$ 738,928.00
2	6	36	12	\$ 132,100.00	\$ 128,020.00
2	6	36	13	\$ 1,714,699.00	\$ 1,173,988.00
2	6	36	17	\$ 7,144,240.00	\$ 6,707,226.00
2	6	36	22	\$ 2,846,485.00	\$ 1,402,132.00
2	6	36	25	\$ 32,169,898.00	\$ 31,203,805.00
Article	Section	Ü	Lines	int Enacted	nt Reduced
2	6	36	27	\$ 35,629.00	\$ 4,242.00
2	6	36	29	\$ 150,709.00	\$ 70,597.00
2	6	36	32	\$ 9,800,803.00	\$ 9,764,711.00
2	6	37	2	\$ 6,947,807.00	\$ 5,645,076.00
2	6	37	4	\$ 257,003.00	\$ 250,554.00
2	6	37	6	\$ 713,067.00	\$ 606,285.00
2	6	37	12	\$ 844,516.00	\$ 85,200.00
2	6	37	16	\$ 291,177.00	\$ 158,637.00

2	6	44	6	\$ 611,854.00	\$ 602,162.00
2	6	43	31	\$ 1,207,076.00	\$ 717,642.00
2	6	43	16	\$ 2,408,056.00	\$ 2,370,595.00
2	6	42	30	\$ 665,102.00	\$ 655,768.00
Article	Section		Lines	int Enacted	nt Reduced
2	6	42	21	\$ 1,218,863.00	\$ 954,218.00
2	6	42	16	\$ 115,263,258.00	\$ 110,789,175.00
2	6	42	7	\$ 3,753,922.00	\$ 3,336,169.00
2	6	41	34	\$ 2,036,156.00	\$ 2,003,201.00
2	6	41	23	\$ 1,481,199.00	\$ 842,340.00
2	6	41	15	\$ 604,074.00	\$ 463,979.00
2	6	41	8	\$ 3,869,144.00	\$ 2,613,539.00
2	6	41	2	\$ 3,580,000.00	\$ 1,850,000.00
2	6	40	26	\$ 1,638,504.00	\$ 1,430,654.00
2	6	40	21	\$ 146,900.00	\$ 92,202.00
2	6	40	19	\$ 6,659.00	\$ 5,908.00
2	6	40	13	\$ 13,697.00	\$ 6,544.00
2	6	40	10	\$ 211,569.00	\$ 201,371.00
2	6	40	8	\$ 1,696,289.00	\$ 1,447,379.00
2	6	40	5	\$ 243,859.00	\$ 215,641.00
2	6	39	29	\$ 5,171,093.00	\$ 750,000.00
2	6	39	18	\$ 100,887.00	\$ 62,118.00
2	6	39	16	\$ 4,632,406.00	\$ 1,768,860.00
2	6	39	12	\$ 2,945,014.00	\$ 2,808,232.00
2	6	39	10	\$ 1,343,707.00	\$ 862,997.00
2	6	39	7	\$ 935,247.00	\$ 870,636.00
2	6	39	6	\$ 2,840,594.00	\$ 2,831,344.00
2	6	39	5	\$ 752,713.00	\$ 402,187.00
2	6	38	34	\$ 462,933.00	\$ 360,500.00
2	6	38	32	\$ 3,116,446.00	\$ 2,390,293.00
2	6	38	25	\$ 160,469.00	\$ 122,081.00
2	6	38	21	\$ 1,030,264.00	\$ 909,561.00
2	6	38	17	\$ 348,212.00	\$ 264,412.00
2	6	38	11	\$ 207,248.00	\$ 112,078.00
2	6	38	4	\$ 58,702.00	\$ 58,201.00
2	6	37	33	\$ 336,168.00	\$ 87,836.00
2	6	37	31	\$ 764,742.00	\$ 567,745.00
2	6	37	22	\$ 9,132,919.00	\$ 7,714,977.00
2	6	37	18	\$ 903,355.00	\$ 595,073.00

Article	Section	Page I	Lines	Amou	nt Enacted	Amou	nt Reduced
2	8	55	9	\$	244,413.00	\$	227,147.00
2	8	55	8	\$	8,889,527.00	\$	8,007,500.00
2	8	55	6	\$	1,104,182.00	\$	1,066,850.00
2	7a	54	20	\$	50,037.00	\$	18,513.00
2	7a	54	18	\$	104,200.00	\$	55,400.00
2	7a	54	12	\$	165,413.00	\$	67,849.00
2	7.3	53	28	\$	202,797.00	\$	147,510.00
2	7.3	53	26	\$	189,325.00	\$	43,450.00
2	7.3	53	22	\$	219,764.00	\$	191,709.00
2	7	52	33	\$	239,594.00	\$	195,056.00
2	7	52	19	\$	1,542,588.00	\$	991,996.00
2	7	52	7	\$	10,096,449.00	\$	5,092,677.00
2	7	52	5	\$	40,931,805.00	\$	39,131,834.00
2	7	51	29	\$	106,602.00	\$	98,155.00
2	7	51	25	\$	316,875.00	\$	72,637.00
2	7	51	23	\$	236,814.00	\$	201,760.00
2	7	51	20	\$	45,623.00	\$	22,152.00
2	7	51	3	\$	384,732.00	\$	229,256.00
2	7	50	24	\$	41,803.00	\$	31,186.00
2	7	50	20	\$	209,030.00	\$	198,287.00
2	7	50	16	\$	238,787.00	\$	195,682.00
2	7	50	14	\$	859,146.00	\$	857,993.00
2	7	50	13	\$	465,228.00	\$	220,700.00
2	6.2	49	13	\$	166,136.00	\$	129,434.00
2	6.1	49	8	\$	456,445.00	\$	335,121.00
2	6	48	27	\$	31,724.00	\$	29,547.00
2	6	48	25	\$	83,291.00	\$	67,489.00
2	6	48	23	\$	328,756.00	\$	123,517.00
2	6	48	10	\$	176,273.00	\$	30,388.00
2	6	48	7	\$	64,283.00	\$	1,317.00
2	6	47	13	\$	1,263,679.00	\$	983,450.00
2	6	46	23	\$	2,082,256.00	\$	2,037,256.00
2	6	46	5	\$	385,354.00	\$	356,789.00
2	6	45	25	\$	265,598.00	\$	242,733.00
2	6	45	22	\$	4,826,546.00	\$	1,896,151.00
2	6	45	10	\$	330,550.00	\$	41,972.00
2	6	45	2	\$	208,988.00	\$	189,284.00
2	6	44	33	\$	72,117,826.00	\$	69,159,373.00

2	8	55	13	\$ 697,026.00	\$ 603,187.00
2	8	55	14	\$ 380,093.00	\$ 357,046.00
2	8	55	16	\$ 264,147.00	\$ 217,692.00
2	8	55	17	\$ 805,265.00	\$ 456,743.00
2	8	55	21	\$ 313,919.00	\$ 281,235.00
2	8	55	24	\$ 103,710.00	\$ 78,455.00
2	8	55	27	\$ 1,694,461.00	\$ 983,917.00
2	8	55	29	\$ 86,234.00	\$ 51,640.00
2	8	55	31	\$ 148,407.00	\$ 134,469.00
2	8	55	33	\$ 132,961.00	\$ 91,042.00
2	8	56	1	\$ 461,976.00	\$ 266,600.00
2	8	56	14	\$ 1,435,918.00	\$ 1,429,587.00
2	8	57	8	\$ 1,119,534.00	\$ 1,116,364.00
2	8	57	12	\$ 232,666.00	\$ 89,893.00
2	8	57	15	\$ 2,084,242.00	\$ 910,306.00
2	8	57	17	\$ 2,149,559.00	\$ 1,960,967.00
2	8	57	19	\$ 164,319.00	\$ 126,350.00
2	8	57	24	\$ 754,589.00	\$ 749,752.00
2	8	57	26	\$ 2,395,780.00	\$ 2,325,690.00
2	8	57	27	\$ 800,000.00	\$ 766,258.00
2	8	57	29	\$ 1,093,138.00	\$ 1,069,250.00
2	8	57	32	\$ 1,290,556.00	\$ 955,287.00
2	8	58	7	\$ 149,677.00	\$ 81,651.00
2	8	58	13	\$ 753,200.00	\$ 699,347.00
2	8	58	14	\$ 83,404.00	\$ 75,463.00
2	8	58	16	\$ 77,595.00	\$ 49,838.00
2	8	58	21	\$ 1,132,065.00	\$ 1,131,120.00
2	8	58	25	\$ 1,013,511.00	\$ 277,510.00
2	8	59	4	\$ 753,573.00	\$ 742,685.00
2	8	59	7	\$ 1,763,492.00	\$ 1,756,372.00
2	8	59	12	\$ 960,000.00	\$ 944,700.00
2	8	59	15	\$ 1,529,639.00	\$ 1,528,089.00
2	8	59	21	\$ 876,700.00	\$ 870,500.00
2	8	59	31	\$ 2,450,000.00	\$ 2,433,110.00
2	8	60	13	\$ 1,515,000.00	\$ 1,496,720.00
2	8	60	17	\$ 1,544,936.00	\$ 1,031,289.00
2	8	60	20	\$ 2,758,571.00	\$ 2,226,921.00
2	8	60	22	\$ 314,387.00	\$ 180,251.00
2	8	60	24	\$ 179,015.00	\$ 167,308.00

2	8	60	25	\$ 422,839.00	\$ 88,490.00
2	8	60	33	\$ 1,900,000.00	\$ 1,856,918.00
2	8	61	6	\$ 741,404.00	\$ 731,508.00
Article	Section		Lines	int Enacted	nt Reduced
2	8	61	14	\$ 75,456.00	\$ 74,856.00
2	8	61	19	\$ 5,536,309.00	\$ 5,436,119.00
2	8	62	1	\$ 2,368,807.00	\$ 1,934,221.00
2	8	62	34	\$ 417,471.00	\$ 393,994.00
2	8	63	15	\$ 61,444.00	\$ 43,262.00
2	8.1	64	32	\$ 71,111.00	\$ 19,433.00
2	8.1	65	7	\$ 196,010.00	\$ 91,012.00
2	8.1	65	9	\$ 226,451.00	\$ 224,019.00
2	8a	66	19	\$ 210,226.00	\$ 203,032.00
2	8a	66	27	\$ 410,506.00	\$ 395,040.00
2	8a	67	2	\$ 112,875.00	\$ 107,825.00
2	9	67	16	\$ 1,000,000.00	\$ 650,000.00
2	9	67	18	\$ 546,697.00	\$ 527,947.00
2	9	67	19	\$ 594,961.00	\$ 439,992.00
2	9	67	21	\$ 625,400.00	\$ 376,091.00
2	10	68	28	\$ 1,329,133.00	\$ 1,290,331.00
2	10	69	3	\$ 301,427.00	\$ 160,394.00
2	10	69	5	\$ 35,410.00	\$ 21,499.00
2	10	69	14	\$ 1,017,627.00	\$ 857,081.00
2	10	69	16	\$ 133,535.00	\$ 42,881.00
2	10	69	24	\$ 440,745.00	\$ 325,703.00
2	10	69	27	\$ 897,000.00	\$ 876,576.00
2	10	69	31	\$ 537,277.00	\$ 416,216.00
2	10	69	32	\$ 287,897.00	\$ 191,657.00
2	10	70	2	\$ 3,728,632.00	\$ 2,690,227.00
2	10	70	6	\$ 179,832.00	\$ 57,116.00
2	10	70	9	\$ 30,354.00	\$ 16,989.00
2	10	70	12	\$ 340,662.00	\$ 118,301.00
2	10	70	15	\$ 2,255,312.00	\$ 1,329,925.00
2	10	70	19	\$ 517,000.00	\$ 503,003.00
2	10	70	29	\$ 971,564.00	\$ 940,549.00
2	10	70	32	\$ 844,033.00	\$ 809,274.00
2	10	71	2	\$ 439,102.00	\$ 431,187.00
2	10	71	4	\$ 2,028,419.00	\$ 1,997,538.00
2	10	71	7	\$ 388,234.00	\$ 358,227.00

2								
2         10         71         26         \$ 1,687,375.00         \$ 1,681,150.00           2         10         71         30         \$ 482,984.00         \$ 482,081.00           2         10         72         2         \$ 569,392.00         \$ 252,046.00           2         10.1         72         21         \$ 84,416.00         \$ 83,176.00           2         112         73         9         \$ 2,779,179.00         \$ 2,470,124.00           Article         Section         Page Lines         Amount Enacted         Amount Reduced           2         12         73         13         \$ 3,780,350.00         \$ 3,695,486.00           2         12         73         15         \$ 323,290.00         \$ 118,236.00           2         12a         73         29         \$ 64,920.00         \$ 3,695,486.00           2         13         74         16         \$ 108,891.00         \$ 71,460.00           2         13         74         21         \$ 3,673,697.00         \$ 2,796,822.00           2         13         74         25         \$ 113,101.00         \$ 61,784.00           2         13         75	2	10	71	11	\$	2,571,910.00	\$	2,523,733.00
2         10         71         30         \$ 482,984.00         \$ 482,081.00           2         10         72         2         \$ 569,392.00         \$ 252,046.00           2         10.1         72         21         \$ 84,416.00         \$ 83,176.00           2         112         73         9         \$ 2,779,179.00         \$ 2,470,124.00           Article         Section         Page   Lines         Amount Enacted         Amount Reduced           2         12         73         13         \$ 3,780,350.00         \$ 3,695,486.00           2         12         73         15         \$ 332,990.00         \$ 118,236.00           2         12a         73         29         \$ 64,920.00         \$ 37,620.00           2         13         74         14         \$ 3,422,605.00         \$ 1,684,298.00           2         13         74         16         \$ 108,891.00         \$ 71,460.00           2         13         74         25         \$ 113,101.00         \$ 61,784.00           2         13         74         25         \$ 113,101.00         \$ 77,375.00           2         13         75         9         \$ 3,404,148.00         \$ 2,	2	10	71	15	\$	477,157.00	\$	463,676.00
2         10         72         2         S 569,392.00         S 252,046.00           2         10.1         72         21         S 84,416.00         S 83,176.00           2         12         73         9         S 2,779,179.00         S 2,470,124.00           Article         Section         Page Lines         Amount Enacted         Amount Reduced           2         12         73         15         S 323,290.00         S 18,236.00           2         12a         73         29         S 64,920.00         S 37,620.00           2         13         74         14         S 3,622,605.00         S 1,684,298.00           2         13         74         16         S 108,891.00         S 71,460.00           2         13         74         21         S 3,673,697.00         S 2,796,822.00           2         13         74         25         S 113,101.00         S 61,784.00           2         13         75         9         S ,440,148.00         S 2,965,872.00           2         13         75         10         S 142,492.00         S 71,094.00           2         14         76         15         S 325,000.00         S 315,2	2	10	71	26	\$	1,687,375.00	\$	1,681,150.00
Table	2	10	71	30	\$	482,984.00	\$	482,081.00
Article         Section         Page         Lines         Amount Enacted         Amount Reduced           2         12         73         13         \$ 3,780,350.00         \$ 3,695,486.00           2         12         73         15         \$ 323,290.00         \$ 118,236.00           2         12a         73         29         \$ 64,920.00         \$ 37,620.00           2         13         74         14         \$ 3,422,605.00         \$ 1,684,298.00           2         13         74         16         \$ 108,891.00         \$ 71,460.00           2         13         74         21         \$ 3,673,697.00         \$ 2,796,822.00           2         13         74         25         \$ 113,101.00         \$ 61,784.00           2         13         74         25         \$ 113,101.00         \$ 77,375.00           2         13         75         9         \$ 3,404,148.00         \$ 2,965,787.00           2         14         76         15         \$ 325,000.00         \$ 315,292.00           2         14         76         18         \$ 375,000.00         \$ 359,544.00           2         14         76         22         \$ 1,365,000.00 </td <td>2</td> <td>10</td> <td>72</td> <td>2</td> <td>\$</td> <td>569,392.00</td> <td>\$</td> <td>252,046.00</td>	2	10	72	2	\$	569,392.00	\$	252,046.00
Article         Section         Page Lines         Amount Enacted         Amount Reduced           2         12         73         13         \$ 3,780,350.00         \$ 3,695,486.00           2         12         73         15         \$ 323,290.00         \$ 118,236.00           2         12a         73         29         \$ 64,920.00         \$ 37,620.00           2         13         74         14         \$ 3,422,605.00         \$ 1,684,298.00           2         13         74         16         \$ 108,891.00         \$ 71,460.00           2         13         74         21         \$ 3,673,697.00         \$ 2,796,822.00           2         13         74         25         \$ 113,101.00         \$ 61,784.00           2         13         74         25         \$ 113,101.00         \$ 77,375.00           2         13         75         9         \$ 3,404,148.00         \$ 2,965,787.00           2         14         76         15         \$ 325,000.00         \$ 315,292.00           2         14         76         18         \$ 375,000.00         \$ 315,292.00           2         14         76         20         \$ 635,000.00         \$ 13	2	10.1	72	21	\$	84,416.00	\$	83,176.00
2         12         73         13         \$ 3,780,350.00         \$ 3,695,486.00           2         12         73         15         \$ 323,290.00         \$ 118,236.00           2         12a         73         29         \$ 64,920.00         \$ 37,620.00           2         13         74         14         \$ 3,422,605.00         \$ 1,684,298.00           2         13         74         16         \$ 108,891.00         \$ 71,460.00           2         13         74         21         \$ 3,673,697.00         \$ 2,796,822.00           2         13         74         25         \$ 113,101.00         \$ 61,784.00           2         13         74         25         \$ 113,101.00         \$ 61,784.00           2         13         75         9         \$ 3,404,148.00         \$ 2,965,787.00           2         13         75         10         \$ 142,492.00         \$ 71,094.00           2         14         76         15         \$ 325,000.00         \$ 315,292.00           2         14         76         18         \$ 375,000.00         \$ 315,292.00           2         14         76         20         \$ 635,000.00         \$ 13,455	2	12	73	9	\$	2,779,179.00	\$	2,470,124.00
2         12         73         15         \$ 323,290.00         \$ 118,236.00           2         12a         73         29         \$ 64,920.00         \$ 37,620.00           2         13         74         14         \$ 3,422,605.00         \$ 1,684,298.00           2         13         74         16         \$ 108,891.00         \$ 71,460.00           2         13         74         21         \$ 3,673,697.00         \$ 2,796,822.00           2         13         74         25         \$ 113,101.00         \$ 61,784.00           2         13         74         33         \$ 92,470.00         \$ 77,375.00           2         13         75         9         \$ 3,404,148.00         \$ 2,965,787.00           2         13         75         10         \$ 142,492.00         \$ 71,094.00           2         14         76         15         \$ 325,000.00         \$ 315,292.00           2         14         76         18         \$ 375,000.00         \$ 315,292.00           2         14         76         20         \$ 635,000.00         \$ 61,500.00           2         14         76         22         \$ 1,365,000.00         \$ 13,4,550.0	Article	Section	Page	Lines	Amou	ınt Enacted	Amou	nt Reduced
2       12a       73       29       \$ 64,920.00       \$ 37,620.00         2       13       74       14       \$ 3,422,605.00       \$ 1,684,298.00         2       13       74       16       \$ 108,891.00       \$ 71,460.00         2       13       74       21       \$ 3,673,697.00       \$ 2,796,822.00         2       13       74       25       \$ 113,101.00       \$ 61,784.00         2       13       74       33       \$ 92,470.00       \$ 77,375.00         2       13       75       9       \$ 3,404,148.00       \$ 2,965,787.00         2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 315,292.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 13,4550.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00 </td <td>2</td> <td>12</td> <td>73</td> <td>13</td> <td>\$</td> <td>3,780,350.00</td> <td>\$</td> <td>3,695,486.00</td>	2	12	73	13	\$	3,780,350.00	\$	3,695,486.00
2       13       74       14       \$ 3,422,605.00       \$ 1,684,298.00         2       13       74       16       \$ 108,891.00       \$ 71,460.00         2       13       74       21       \$ 3,673,697.00       \$ 2,796,822.00         2       13       74       25       \$ 113,101.00       \$ 61,784.00         2       13       74       33       \$ 92,470.00       \$ 77,375.00         2       13       75       9       \$ 3,404,148.00       \$ 2,965,787.00         2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 13,4550.00         2       14       76       22       \$ 1,365,000.00       \$ 17,6380.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       28       \$ 21,209.00       \$ 176,380.00	2	12	73	15	\$	323,290.00	\$	118,236.00
2       13       74       16       \$ 108,891.00       \$ 71,460.00         2       13       74       21       \$ 3,673,697.00       \$ 2,796,822.00         2       13       74       25       \$ 113,101.00       \$ 61,784.00         2       13       74       33       \$ 92,470.00       \$ 77,375.00         2       13       75       9       \$ 3,404,148.00       \$ 2,965,787.00         2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       28       \$ 21,246.00       \$ 176,380.00 <td>2</td> <td>12a</td> <td>73</td> <td>29</td> <td>\$</td> <td>64,920.00</td> <td>\$</td> <td>37,620.00</td>	2	12a	73	29	\$	64,920.00	\$	37,620.00
2       13       74       21       \$ 3,673,697.00       \$ 2,796,822.00         2       13       74       25       \$ 113,101.00       \$ 61,784.00         2       13       74       33       \$ 92,470.00       \$ 77,375.00         2       13       75       9       \$ 3,404,148.00       \$ 2,965,787.00         2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 314,550.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       22       \$ 1,365,000.00       \$ 1340,550.00         2       14       76       24       \$ 2,499,029.00       \$ 13,07,022.00         2       14       76       28       \$ 212,009.00       \$ 176,380.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,	2	13	74	14	\$	3,422,605.00	\$	1,684,298.00
2       13       74       25       \$ 113,101.00       \$ 61,784.00         2       13       74       33       \$ 92,470.00       \$ 77,375.00         2       13       75       9       \$ 3,404,148.00       \$ 2,965,787.00         2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       22       \$ 1,365,000.00       \$ 13,07,022.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       28       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.0	2	13	74	16	\$	108,891.00	\$	71,460.00
2       13       74       33       \$ 92,470.00       \$ 77,375.00         2       13       75       9       \$ 3,404,148.00       \$ 2,965,787.00         2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       28       \$ 212,009.00       \$ 176,380.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00 <td>2</td> <td>13</td> <td>74</td> <td>21</td> <td>\$</td> <td>3,673,697.00</td> <td>\$</td> <td>2,796,822.00</td>	2	13	74	21	\$	3,673,697.00	\$	2,796,822.00
2       13       75       9       \$ 3,404,148.00       \$ 2,965,787.00         2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       1       \$ 1,136,700.00       \$ 88,100.00<	2	13	74	25	\$	113,101.00	\$	61,784.00
2       13       75       10       \$ 142,492.00       \$ 71,094.00         2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       8       \$ 60,635.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,354,043.00       \$ 1,226,031.00	2	13	74	33	\$	92,470.00	\$	77,375.00
2       14       76       15       \$ 325,000.00       \$ 315,292.00         2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.0	2	13	75	9	\$	3,404,148.00	\$	2,965,787.00
2       14       76       18       \$ 375,000.00       \$ 359,544.00         2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552	2	13	75	10	\$	142,492.00	\$	71,094.00
2       14       76       20       \$ 635,000.00       \$ 61,500.00         2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 748,552.00         2       15       79       5       1,062,997.00       \$ 748,552.0	2	14	76	15	\$	325,000.00	\$	315,292.00
2       14       76       22       \$ 1,365,000.00       \$ 134,550.00         2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762	2	14	76	18	\$	375,000.00	\$	359,544.00
2       14       76       24       \$ 2,499,029.00       \$ 1,307,022.00         2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 748,552.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       20       \$ 26,416.00       \$ 20,103.0	2	14	76	20	\$	635,000.00	\$	61,500.00
2       14       76       28       \$ 21,246.00       \$ 19,786.00         2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00<	2	14	76	22	\$	1,365,000.00	\$	134,550.00
2       14       76       29       \$ 212,009.00       \$ 176,380.00         2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,96	2	14	76	24	\$	2,499,029.00	\$	1,307,022.00
2       14       77       8       \$ 60,635.00       \$ 50,380.00         2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14	76	28	\$	21,246.00	\$	19,786.00
2       14       77       11       \$ 2,888,223.00       \$ 2,774,574.00         2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14	76	29	\$	212,009.00	\$	176,380.00
2       14a       78       6       \$ 73,938.00       \$ 50,978.00         2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14	77	8	\$	60,635.00	\$	50,380.00
2       14a       78       7       \$ 39,536.00       \$ 11,145.00         2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14	77	11	\$	2,888,223.00	\$	2,774,574.00
2       14a       78       11       \$ 1,136,700.00       \$ 88,100.00         2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14a	78	6	\$	73,938.00	\$	50,978.00
2       14a       78       12       \$ 1,354,043.00       \$ 1,226,031.00         2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14a	78	7	\$	39,536.00	\$	11,145.00
2       14a       78       14       \$ 653,539.00       \$ 169,895.00         2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14a	78	11	\$	1,136,700.00	\$	88,100.00
2       15       78       25       \$ 444,131.00       \$ 413,422.00         2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14a	78	12	\$	1,354,043.00	\$	1,226,031.00
2       15       79       5       \$ 1,062,997.00       \$ 748,552.00         2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	14a	78	14	\$	653,539.00	\$	169,895.00
2       15       79       12       \$ 1,469,826.00       \$ 762,728.00         2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	15	78	25	\$	444,131.00	\$	413,422.00
2       15       79       20       \$ 26,416.00       \$ 20,103.00         2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	15	79	5	\$	1,062,997.00	\$	748,552.00
2       15       79       24       \$ 2,334,116.00       \$ 2,015,988.00         2       15       79       26       \$ 2,037,933.00       \$ 1,968,461.00	2	15	79	12	\$	1,469,826.00	\$	762,728.00
2 15 79 26 \$ 2,037,933.00 \$ 1,968,461.00	2	15	79	20	\$	26,416.00	\$	20,103.00
	2	15	79	24	\$	2,334,116.00	\$	2,015,988.00
2 15 79 34 \$ 3,986,162.00 \$ 3,907,066.00	2	15	79	26	\$	2,037,933.00	\$	1,968,461.00
	2	15	79	34	\$	3,986,162.00	\$	3,907,066.00

15								
15	2	15	80	3	\$	270,386.00	\$	233,086.00
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2	15	80	8	\$	7,293,123.00	\$	6,423,879.00
15	2	15	80	9	\$	1,300,774.00	\$	1,031,246.00
2         15         80         18         \$ 935,938.00         \$ 916,492.00           2         15         80         19         \$ 929,925.00         \$ 773,667.00           2         15         80         24         \$ 503,668.00         \$ 370,127.00           2         15         80         25         \$ 417,978.00         \$ 415,653.00           2         15         80         27         \$ 55,633.00         \$ 53,118.00           Article         Section         Page         Lines         Amount Eneted         Amount Reduced           2         15         80         30         \$ 325,415.00         \$ 312,917.00           2         15         80         32         \$ 141,385.00         \$ 137,105.00           2         15         80         32         \$ 141,385.00         \$ 107,999.00           2         15         81         6         125,319.00         \$ 117,999.00           2         15         81         9         2,455,832.00         \$ 107,999.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         15         108,753.00         \$ 108,677	2	15	80	11	\$	5,131,958.00	\$	3,532,030.00
2         15         80         19         \$ 929,925.00         \$ 773,667.00           2         15         80         24         \$ 503,668.00         \$ 370,127.00           2         15         80         25         \$ 417,978.00         \$ 415,653.00           2         15         80         27         \$ 55,633.00         \$ 33,118.00           2         15         80         29         \$ 461,018.00         \$ 448,666.00           Article         Section         Page Lines         Amount Enacted         Amount Reduced           2         15         80         30         \$ 325,415.00         \$ 312,917.00           2         15         80         32         \$ 141,385.00         \$ 137,105.00           2         15         81         6         \$ 125,319.00         \$ 107,999.00           2         15         81         9         \$ 2,455,832.00         \$ 1,973,733.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         15         \$ 108,675.00         \$ 108,677.00           2         16         81         19	2	15	80	16	\$	360,993.00	\$	315,869.00
2	2	15	80	18	\$	935,938.00	\$	916,492.00
Table   Tabl	2	15	80	19	\$	929,925.00	\$	773,667.00
2         15         80         27         \$ 55,633.00         \$ 448,666.00           Article         Section         Page         Lines         Amount Enacted         Amount Reduced           2         15         80         30         \$ 325,415.00         \$ 312,917.00           2         15         80         32         \$ 141,385.00         \$ 137,105.00           2         15         80         34         \$ 132,695.00         \$ 107,999.00           2         15         81         6         \$ 125,319.00         \$ 111,834.00           2         15         81         9         \$ 2,455,832.00         \$ 1,973,733.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         19         \$ 713,137.00         \$ 647,386.00           2         16         81         19         \$ 713,137.00         \$ 647,386.00           2         18         82         18         \$ 1,154,427.00         \$ 735,801.00           2         19         83         11         \$ 332,243.00 <t< td=""><td>2</td><td>15</td><td>80</td><td>24</td><td>\$</td><td>503,668.00</td><td>\$</td><td>370,127.00</td></t<>	2	15	80	24	\$	503,668.00	\$	370,127.00
Article         Section         Page Lines         Amount Enacted         Amount Enacted         Amount Educed           2         15         80         30         \$ 325,415.00         \$ 312,917.00           2         15         80         32         \$ 141,385.00         \$ 137,105.00           2         15         80         34         \$ 132,695.00         \$ 107,999.00           2         15         81         6         \$ 125,319.00         \$ 111,834.00           2         15         81         9         \$ 2,455,832.00         \$ 1,973,733.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         16         81         19         \$ 713,137.00         \$ 647,386.00           2         16         81         19         \$ 713,137.00         \$ 80,7224.00           2         18         82         13         \$ 809,928.00         \$ 873,801.00           2         19         83         11         \$ 332,243.00         \$ 15,599,846.00           2         19         83         13         \$ 249,566.00	2	15	80	25	\$	417,978.00	\$	415,653.00
Article         Section         Page         Lines         Amount Enacted         Amount Reduced           2         15         80         30         \$ 325,415.00         \$ 312,917.00           2         15         80         32         \$ 141,385.00         \$ 137,105.00           2         15         80         34         \$ 132,695.00         \$ 107,999.00           2         15         81         6         \$ 125,319.00         \$ 111,834.00           2         15         81         9         \$ 2,455,832.00         \$ 1,973,733.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         16         81         19         \$ 713,137.00         \$ 647,386.00           2         16         81         19         \$ 713,137.00         \$ 647,386.00           2         18         82         13         \$ 809,928.00         \$ 807,224.00           2         18         82         18         \$ 1,554,427.00         \$ 735,801.00           2         19         83         11         \$ 332,243.00         \$ 2,564,00           2         19         83         13         \$ 249,566.00 <t< td=""><td>2</td><td>15</td><td>80</td><td>27</td><td>\$</td><td>55,633.00</td><td>\$</td><td>53,118.00</td></t<>	2	15	80	27	\$	55,633.00	\$	53,118.00
2         15         80         30         \$ 325,415.00         \$ 312,917.00           2         15         80         32         \$ 141,385.00         \$ 137,105.00           2         15         80         34         \$ 132,695.00         \$ 107,999.00           2         15         81         6         \$ 125,319.00         \$ 111,834.00           2         15         81         9         \$ 2,455,832.00         \$ 1,973,733.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         15         \$ 108,753.00         \$ 108,677.00           2         16         81         19         \$ 713,137.00         \$ 647,386.00           2         18         82         13         \$ 809,928.00         \$ 807,224.00           2         18         82         18         \$ 1,154,427.00         \$ 735,801.00           2         19         83         4         \$ 1,838,248.00         \$ 1,599,846.00           2         19         83         13         \$ 249,566.00         \$ 180,539.00 <td>2</td> <td>15</td> <td>80</td> <td>29</td> <td>\$</td> <td>461,018.00</td> <td>\$</td> <td>448,666.00</td>	2	15	80	29	\$	461,018.00	\$	448,666.00
2         15         80         32         \$ 141,385.00         \$ 137,105.00           2         15         80         34         \$ 132,695.00         \$ 107,999.00           2         15         81         6         \$ 125,319.00         \$ 111,834.00           2         15         81         9         \$ 2,455,832.00         \$ 1,973,733.00           2         15         81         10         \$ 97,105.00         \$ 83,518.00           2         15         81         15         \$ 108,753.00         \$ 108,677.00           2         16         81         19         \$ 713,137.00         \$ 647,386.00           2         18         82         13         \$ 809,928.00         \$ 807,224.00           2         18         82         18         \$ 1,154,427.00         \$ 735,801.00           2         19         83         4         \$ 1,838,248.00         \$ 1,599,846.00           2         19         83         11         \$ 332,243.00         \$ 275,820.00           2         19         83         13         \$ 249,566.00         \$ 180,539.00           2         19         83         18         \$ 287,780.00         \$ 2,564,105.0	Article	Section	Page	Lines	Amou	ınt Enacted	Amou	int Reduced
2       15       80       34       \$ 132,695.00       \$ 107,999.00         2       15       81       6       \$ 125,319.00       \$ 111,834.00         2       15       81       9       \$ 2,455,832.00       \$ 1,973,733.00         2       15       81       10       \$ 97,105.00       \$ 83,518.00         2       15       81       15       \$ 108,753.00       \$ 108,677.00         2       16       81       19       \$ 713,137.00       \$ 647,386.00         2       18       82       13       \$ 809,928.00       \$ 807,224.00         2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.0	2	15	80	30	\$	325,415.00	\$	312,917.00
2       15       81       6       \$ 125,319.00       \$ 111,834.00         2       15       81       9       \$ 2,455,832.00       \$ 1,973,733.00         2       15       81       10       \$ 97,105.00       \$ 83,518.00         2       15       81       15       \$ 108,753.00       \$ 108,677.00         2       16       81       19       \$ 713,137.00       \$ 647,386.00         2       18       82       13       \$ 809,928.00       \$ 807,224.00         2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.0	2	15	80	32	\$	141,385.00	\$	137,105.00
2       15       81       9       \$ 2,455,832.00       \$ 1,973,733.00         2       15       81       10       \$ 97,105.00       \$ 83,518.00         2       15       81       15       \$ 108,753.00       \$ 108,677.00         2       16       81       19       \$ 713,137.00       \$ 647,386.00         2       18       82       13       \$ 809,928.00       \$ 807,224.00         2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       22       \$ 85,516.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00<	2	15	80	34	\$	132,695.00	\$	107,999.00
2       15       81       10       \$ 97,105.00       \$ 83,518.00         2       15       81       15       \$ 108,753.00       \$ 108,677.00         2       16       81       19       \$ 713,137.00       \$ 647,386.00         2       18       82       13       \$ 809,928.00       \$ 807,224.00         2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       84,567.00       \$ 69,130.00	2	15	81	6	\$	125,319.00	\$	111,834.00
2       15       81       15       \$ 108,753.00       \$ 108,677.00         2       16       81       19       \$ 713,137.00       \$ 647,386.00         2       18       82       13       \$ 809,928.00       \$ 807,224.00         2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00	2	15	81	9	\$	2,455,832.00	\$	1,973,733.00
2       16       81       19       \$ 713,137.00       \$ 647,386.00         2       18       82       13       \$ 809,928.00       \$ 807,224.00         2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.	2	15	81	10	\$	97,105.00	\$	83,518.00
2       18       82       13       \$ 809,928.00       \$ 807,224.00         2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       8       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00	2	15	81	15	\$	108,753.00	\$	108,677.00
2       18       82       18       \$ 1,154,427.00       \$ 735,801.00         2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00 </td <td>2</td> <td>16</td> <td>81</td> <td>19</td> <td>\$</td> <td>713,137.00</td> <td>\$</td> <td>647,386.00</td>	2	16	81	19	\$	713,137.00	\$	647,386.00
2       19       83       4       \$ 1,838,248.00       \$ 1,599,846.00         2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00 <td>2</td> <td>18</td> <td>82</td> <td>13</td> <td>\$</td> <td>809,928.00</td> <td>\$</td> <td>807,224.00</td>	2	18	82	13	\$	809,928.00	\$	807,224.00
2       19       83       11       \$ 332,243.00       \$ 275,820.00         2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00	2	18	82	18	\$	1,154,427.00	\$	735,801.00
2       19       83       13       \$ 249,566.00       \$ 180,539.00         2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00 <td>2</td> <td>19</td> <td>83</td> <td>4</td> <td>\$</td> <td>1,838,248.00</td> <td>\$</td> <td>1,599,846.00</td>	2	19	83	4	\$	1,838,248.00	\$	1,599,846.00
2       19       83       16       \$ 3,208,975.00       \$ 2,564,105.00         2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	19	83	11	\$	332,243.00	\$	275,820.00
2       19       83       18       \$ 287,780.00       \$ 276,854.00         2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	19	83	13	\$	249,566.00	\$	180,539.00
2       19       83       22       \$ 85,516.00       \$ 49,880.00         2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	19	83	16	\$	3,208,975.00	\$	2,564,105.00
2       19       83       23       \$ 267,251.00       \$ 248,981.00         2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	19	83	18	\$	287,780.00	\$	276,854.00
2       19       83       27       \$ 18,353.00       \$ 13,414.00         2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	19	83	22	\$	85,516.00	\$	49,880.00
2       20       84       8       \$ 84,567.00       \$ 69,130.00         2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	19	83	23	\$	267,251.00	\$	248,981.00
2       20       84       15       \$ 3,824,380.00       \$ 3,708,888.00         2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	19	83	27	\$	18,353.00	\$	13,414.00
2       20       84       26       \$ 812,287.00       \$ 273,534.00         2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	20	84	8	\$	84,567.00	\$	69,130.00
2       20       84       31       \$ 1,495,897.00       \$ 748,599.00         2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	20	84	15	\$	3,824,380.00	\$	3,708,888.00
2       20       85       1       \$ 110,947.00       \$ 25,087.00         2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	20	84	26	\$	812,287.00	\$	273,534.00
2       20       85       8       \$ 183,304.00       \$ 15,958.00         2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	20	84	31	\$	1,495,897.00	\$	748,599.00
2       20       85       14       \$ 307,018.00       \$ 266,766.00         2       20       85       19       \$ 3,055,395.00       \$ 1,524,558.00	2	20	85	1	\$	110,947.00	\$	25,087.00
2 20 85 19 \$ 3,055,395.00 \$ 1,524,558.00	2	20	85	8	\$	183,304.00	\$	15,958.00
	2	20	85	14	\$	307,018.00	\$	266,766.00
2 20 85 24 \$ 165,889.00 \$ 67,962.00	2	20	85	19	\$	3,055,395.00	\$	1,524,558.00
	2	20	85	24	\$	165,889.00	\$	67,962.00

2	20	85	32	\$	7,162,092.00	\$	3,496,642.00
2	20	86	2	\$	187,637.00	\$	180,802.00
2	20	86	5	\$	3,774,538.00	\$	2,979,157.00
2	20	86	8	\$	6,629,480.00	\$	6,602,987.00
2	20	86	12	\$	38,709.00	\$	25,555.00
2	20	86	22	\$	425,867.00	\$	191,879.00
2	20	86	28	\$	56,635.00	\$	55,104.00
2	20	87	6	\$	279,382.00	\$	174,328.00
2	20	87	8	\$	87,740.00	\$	65,911.00
2	20	87	14	\$	923,249.00	\$	862,385.00
2	20	87	24	\$	388,692.00	\$	85,664.00
2	20	87	28	\$	76,956.00	\$	62,206.00
2	20	87	33	\$	8,981,505.00	\$	6,018,768.00
2	20	88	5	\$	26,911.00	\$	20,644.00
2	20	88	6	\$	13,631.00	\$	12,672.00
Article	Section	Page	Lines	Amou	ınt Enacted	Amou	ınt Reduced
2	20	88	8	\$	183,574.00	\$	177,904.00
2	20	88	12	\$	23,914.00	\$	23,195.00
2	20	88	17	\$	498,464.00	\$	143,560.00
2	20	88	22	\$	1,602,314.00	\$	1,133,436.00
2	20	88	26	\$	1,818,769.00	\$	1,737,698.00
2	20	88	28	\$	271,548.00	\$	238,753.00
2	20	89	10	\$	2,287,891.00	\$	1,976,661.00
2	20	89	21	\$	5,716,960.00	\$	5,691,847.00
2	20	89	30	\$	4,392,678.00	\$	4,353,800.00
2	20	90	7	\$	463,437.00	\$	431,012.00
2	21	90	9	\$	27,229.00	\$	7,468.00
2	22	90	17	\$	2,118,996.00	\$	2,083,479.00
2	23	90	30	\$	2,197,658.00	\$	2,085,947.00
2	24	91	13	\$	2,886,283.00	\$	2,856,772.00
2	25	91	27	\$	840,707.00	\$	760,193.00
2	27	93	2	\$	1,996,393.00	\$	108,843.00
2	28	94	6	\$	189,700.00	\$	132,852.00
2	28	94	13	\$	1,625,000.00	\$	1,549,721.00
2	28	94	33	\$	11,027,708.00	\$	9,499,651.00
2	28	95	2	\$	162,297.00	\$	160,545.00
2	28	95	9	\$	615,150.00	\$	315,031.00
2	28	95	11	\$	2,620,558.00	\$	2,589,416.00
2	28	95	15	\$	3,206,203.00	\$	2,393,750.00

2	28	95	25	\$	12,165,873.00	\$	6,195,200.00
2	28	95	28	\$	144,198.00	\$	15,311.00
2	28	95	29	\$	696,002.00	\$	688,162.00
2	28	95	33	\$	688,328.00	\$	382,484.00
2	28	96	1	\$	406,051.00	\$	221,681.00
2	28	96	3	\$	1,697,136.00	\$	1,114,777.00
2	28	96	5	\$	1,962,111.00	\$	1,550,737.00
2	28	96	18	\$	4,795,388.00	\$	3,019,768.00
2	28	96	19	\$	201,730.00	\$	191,127.00
2	28	96	21	\$	349,315.00	\$	342,029.00
2	28	96	22	\$	236,389.00	\$	197,609.00
2	28	96	24	\$	107,785.00	\$	79,906.00
2	28	96	26	\$	110,585.00	\$	20,639.00
2	28	96	28	\$	1,011,854.00	\$	165,696.00
2	28	96	30	\$	1,119,681.00	\$	996,136.00
2	28	97	8	\$	4,380,251.00	\$	2,394,813.00
2	28	97	10	\$	57,244.00	\$	52,904.00
2	28		10	\$		\$	· ·
2		97		\$	293,730.00		275,416.00
2	28	97	13	Э	67,649.00	\$	39,423.00
Autiala	Continu	Dage	linoo	A	nt Engated	A	nt Dadward
Article	Section		Lines		nt Enacted		nt Reduced
2	28	97	15	\$	27,397.00	\$	21,257.00
2 2	28 28	97 97	15 16	\$ \$	27,397.00 1,168,384.00	\$ \$	21,257.00 393,157.00
2 2 2	28 28 28	97 97 97	15 16 18	\$ \$ \$	27,397.00 1,168,384.00 1,170,565.00	\$ \$ \$	21,257.00 393,157.00 986,130.00
2 2 2 2	28 28 28 28	97 97 97 97	15 16 18 30	\$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00	\$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00
2 2 2 2 2	28 28 28 28 28	97 97 97 97 97	15 16 18 30 32	\$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00	\$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00
2 2 2 2 2 2	28 28 28 28 28 28	97 97 97 97 97 97	15 16 18 30 32 34	\$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00	\$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00
2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28	97 97 97 97 97 97 98	15 16 18 30 32 34 2	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00
2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28	97 97 97 97 97 97 97 98	15 16 18 30 32 34 2	\$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28	97 97 97 97 97 97 98 98	15 16 18 30 32 34 2 4 6	\$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28	97 97 97 97 97 97 98 98 98	15 16 18 30 32 34 2 4 6	\$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28	97 97 97 97 97 97 98 98 98	15 16 18 30 32 34 2 4 6 18	\$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28 28	97 97 97 97 97 97 98 98 98 98	15 16 18 30 32 34 2 4 6 18	\$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00 206,099.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00 199,051.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28 28 28	97 97 97 97 97 97 98 98 98 98	15 16 18 30 32 34 2 4 6 18 19 20 24	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00 206,099.00 268,274.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00 199,051.00 266,524.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28 28 28 28	97 97 97 97 97 97 98 98 98 98 98	15 16 18 30 32 34 2 4 6 18 19 20 24 25	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00 206,099.00 268,274.00 43,241.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00 199,051.00 266,524.00 24,413.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28 28 28 28 2	97 97 97 97 97 97 98 98 98 98 98 98	15 16 18 30 32 34 2 4 6 18 19 20 24 25 26	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00 206,099.00 268,274.00 43,241.00 288,813.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00 199,051.00 266,524.00 24,413.00 273,820.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28 28 28 28 2	97 97 97 97 97 97 98 98 98 98 98 98	15 16 18 30 32 34 2 4 6 18 19 20 24 25 26 5	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00 206,099.00 268,274.00 43,241.00 288,813.00 128,324.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00 199,051.00 266,524.00 24,413.00 273,820.00 121,599.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28 28 28 28 2	97 97 97 97 97 97 98 98 98 98 98 98 98	15 16 18 30 32 34 2 4 6 18 19 20 24 25 26 5	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00 206,099.00 268,274.00 43,241.00 288,813.00 128,324.00 199,575.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00 199,051.00 266,524.00 24,413.00 273,820.00 121,599.00 157,968.00
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 28 28 28 28 28 28 28 28 28 28 28 28 2	97 97 97 97 97 97 98 98 98 98 98 98	15 16 18 30 32 34 2 4 6 18 19 20 24 25 26 5	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	27,397.00 1,168,384.00 1,170,565.00 2,993,389.00 406,925.00 31,523.00 55,009.00 1,494,908.00 968,947.00 1,656,899.00 161,197.00 206,099.00 268,274.00 43,241.00 288,813.00 128,324.00	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	21,257.00 393,157.00 986,130.00 2,715,038.00 348,406.00 22,188.00 48,838.00 1,413,780.00 845,649.00 1,085,903.00 123,295.00 199,051.00 266,524.00 24,413.00 273,820.00 121,599.00

2	20	100	22	Φ.	5 000 000 00	Ф.	4.002.104.00
2 2	30 30	100 101	32 7	\$ \$	5,000,000.00 15,385,611.00	\$	4,982,104.00
2	30	101	9	\$		\$	15,021,338.00
2	30	101	11	\$	1,800,000.00 9,206,524.00	\$ \$	1,792,800.00 9,127,174.00
2	30	101	17	\$	15,883,402.00	\$	10,946,401.00
2	30	101	21	\$	8,664,333.00	\$	8,623,383.00
2	30	101	25			\$	
2	30	101	27	\$ \$	1,375,209.00 771,165.00	\$	1,052,466.00 679,179.00
2	30	101	32	\$	39,722,302.00	\$	39,705,500.00
2	30	101	2	\$			
2	30	102	5		1,923,194.00	\$	1,621,559.00
2	30	102	7	\$ \$	1,905,908.00	\$	1,824,490.00
				\$	478,759.00 232,996.00	\$	386,432.00
2 2	30 30	102 102	9 15	\$		\$ \$	164,611.00
2	30	102	19		17,798,923.00		15,979,731.00
				\$	877,073.00	\$	865,796.00
2 2	30 30	102 102	23 27	\$	973,016.00 130,318.00	\$	354,731.00
				\$		\$	7,464.00
2	30	103	4	\$	561,257.00	\$	527,176.00
2	30	103	8	\$	12,901,888.00	\$	10,814,323.00
2	30	103		\$	2,071,731.00	\$	1,355,468.00
2	30	103	17	\$	3,759,907.00	\$	3,718,870.00
2	30	103	22	\$	2,216,093.00	\$	1,871,089.00
2	30	103	24	\$	3,723,862.00	\$	2,345,730.00
Article	Section	O	Lines		int Enacted		int Reduced
2	30	103	30	\$	126,117.00	\$	84,571.00
2	30	104	1	\$	983,737.00	\$	768,745.00
2	30	104	8	\$	3,778,208.00	\$	3,777,651.00
2	30	104	10	\$	2,318,257.00	\$	1,853,450.00
2	30	104	12	\$	1,357,118.00	\$	1,265,161.00
2	30	104	16	\$	14,378,125.00	\$	14,314,690.00
2	30	104	25	\$	1,842,371.00	\$	1,523,552.00
2	30	104	29	\$	6,816,196.00	\$	4,924,540.00
2	30	105	8	\$	1,050,125.00	\$	1,029,729.00
2	30	105	11	\$	97,720.00	\$	24,511.00
2	30	105	20	\$	18,338,039.00	\$	12,368,319.00
2	30	105	30	\$	3,461,461.00	\$	3,326,338.00
2	30	105	33	\$	3,238,136.00	\$	1,906,902.00
2	30	106	3	\$	1,615,139.00	\$	1,571,369.00
2	30	106	15	\$	27,000,000.00	\$	24,713,878.00

2	30	106	18	\$ 899,327.00	\$ 130,165.00
2	30	106	20	\$ 3,243,346.00	\$ 3,224,471.00
2	30	106	23	\$ 1,190,856.00	\$ 868,928.00
2	30	106	26	\$ 369,844.00	\$ 179,276.00
2	30	106	32	\$ 149,872.00	\$ 75,852.00
2	30	107	14	\$ 11,957,333.00	\$ 11,931,823.00
2	30	107	17	\$ 254,590.00	\$ 113,118.00
2	31	107	23	\$ 26,630.00	\$ 22,390.00
2	33	108	23	\$ 10,030,267.00	\$ 6,520,289.00
2	34	108	24	\$ 376,953,236.00	\$ 326,543,021.00
2	35	109	1	\$ 194,846,206.00	\$ 131,785,796.00
2	36	109	9	\$ 89,337,553.00	\$ 67,440,525.00
2	37	109	17	\$ 12,819,035.00	\$ 8,877,209.00
2	38	109	25	\$ 1,287,736.00	\$ 1,238,669.00
2	39	110	1	\$ 7,138,620.00	\$ 4,894,646.00
4	45	113	3	\$ 55,621.00	\$ 43,366.00
4	70	113	28	\$ 814,444.00	\$ 322,347.00
4	75	114	5	\$ 814,389.00	\$ 208,922.00
4	80	114	12	\$ 707,015.00	\$ 40,139.00
5	1	120	1	\$ 1,956,026.00	\$ 1,686,957.00
5	5	120	30	\$ 49,424,759.00	\$ 49,238,777.00
5	6	122	1	\$ 812,500.00	\$ 788,154.00
5	12	124	1	\$ 700,000.00	\$ 674,600.00
5	14	124	19	\$ 383,931.00	\$ 375,950.00
5	20	126	13	\$ 17,000,000.00	\$ 15,887,000.00
5	25	127	8	\$ 2,000,000.00	\$ 1,981,894.00
5	1	127	26	\$ 757,674.00	\$ 647,131.00
5	5	128	16	\$ 20,703,235.00	\$ 17,997,117.00
5	11	129	2	\$ 47,592,340.00	\$ 46,211,466.00
5	14	129	24	\$ 1,799,679.00	\$ 827,300.00
5	14	129	26	\$ 304,679.00	\$ 271,821.00
5	14	129	30	\$ 309,183.00	\$ 164,183.00
5	15	130	12	\$ 539,266.00	\$ 496,154.00
5	15	130	14	\$ 66,322.00	\$ 41,188.00
5	16	130	30	\$ 469,157.00	\$ 225,718.00
5	16	131	1	\$ 379,190.00	\$ 103,914.00
5	17	131	11	\$ 218,018.00	\$ 206,768.00
5	18	131	21	\$ 175,000.00	\$ 94,415.00
5	18	131	24	\$ 178,829.00	\$ 173,732.00

5	18	131	26	\$ 292,528.00	\$ 185,767.00
5	19	132	10	\$ 919,949.00	\$ 588,300.00
5	19	132	13	\$ 91,118.00	\$ 52,831.00
5	21	133	8	\$ 105,724.00	\$ 59,111.00
5	22	133	20	\$ 2,391,688.00	\$ 2,033,029.00
5	25	134	12	\$ 3,035,800.00	\$ 3,029,592.00
5	26	135	13	\$ 509,529.00	\$ 246,756.00
5	26	135	16	\$ 114,600.00	\$ 76,995.00
5	26	135	17	\$ 1,857,549.00	\$ 1,838,914.00
5	49	135	19	\$ 4,455,000.00	\$ 3,640,000.00
5	50	135	29	\$ 55,000,000.00	\$ 50,709,519.00
5	51	136	8	\$ 40,721,264.00	\$ 36,618,117.00
5	52	136	19	\$ 39,238,000.00	\$ 34,676,140.00
5	53	136	30	\$ 48,907,000.00	\$ 48,851,051.00
5	84	138	3	\$ 18,000,000.00	\$ 15,750,000.00
5	1	138	25	\$ 377,007.00	\$ 291,628.00
5	2	139	2	\$ 7,086,553.00	\$ 5,484,725.00
5	5	139	12	\$ 16,328,052.00	\$ 7,956,662.00
5	10	139	30	\$ 102,620.00	\$ 70,133.00
5	10	140	1	\$ 13,964.00	\$ 7,393.00
5	12	140	23	\$ 408,354.00	\$ 384,947.00
5	13	141	2	\$ 1,244,600.00	\$ 1,236,216.00
5	14	141	12	\$ 24,442.00	\$ 21,784.00
5	15	141	25	\$ 71,481.00	\$ 29,303.00
5	15	141	29	\$ 432,384.00	\$ 368,213.00
5	21	143	24	\$ 609,216.00	\$ 446,882.00
5	22	143	26	\$ 2,409,925.00	\$ 2,287,457.00
5	23	144	15	\$ 47,725.00	\$ 36,829.00
5	23	144	16	\$ 249,854.00	\$ 240,116.00
5	23	144	17	\$ 106,000.00	\$ 65,086.00
5	23	144	21	\$ 18,221.00	\$ 11,720.00
5	23	144	22	\$ 79,350.00	\$ 21,104.00
5	23	144	23	\$ 977,202.00	\$ 662,911.00
5	23	144	25	\$ 327,143.00	\$ 297,559.00
5	36	144	27	\$ 2,360,422.00	\$ 2,349,256.00
5	47	145	13	\$ 40,836,944.00	\$ 34,836,944.00
5	48	145	24	\$ 6,643,398.00	\$ 5,805,934.00
5	50	146	2	\$ 8,015,001.00	\$ 5,080,001.00
5	1-1	146	17	\$ 1,962,891.00	\$ 1,628,515.00

5	1-1	147	3	\$	153,750.00	\$	24,791.00
5	1-1	147	4	\$	147,237.00	\$	141,620.00
5	1-1	147	7	\$	39,026.00	\$	38,564.00
5	1-1	147	8	\$	205,525.00	\$	108,505.00
5	1-1	147	9	\$	630,000.00	\$	527,682.00
5	1-3	147	20	\$	5,279,525.00	\$	4,799,546.00
5	1-4	148	2	\$	8,420,826.00	\$	7,689,961.00
5	1-9	148	19	\$	8,283,356.00	\$	7,645,856.00
5	2-174	149	15	\$	7,089,803.00	\$	6,741,635.00
5	4-1	149	24	\$	16,893,500.00	\$	15,519,759.00
5	5-1	150	3	\$	59,279,743.00	\$	55,420,432.00
5	10B	155	24	\$	70,232,823.00	\$	69,974,831.00
5	10G	156	21	\$	774,870.00	\$	492,504.00
6	520	185	4	\$	225,000.00	\$	112,500.00
6	535	185	29	\$	50,000.00	\$	25,000.00
6	600	188	32	\$	1,596,832.00	\$	1,492,444.00
6	605	189	9	\$	16,908,500.00	\$	16,534,259.00
6	1030	212	1	\$	13,154,403.00	\$	12,619,408.00
6	1155	218	18	\$	53,375.00	\$	4,760.00
6	1160	218	25	\$	671,800.00	\$	567,087.00
6	1175	219	17	\$	300,000.00	\$	32,514.00
6	1185	220	16	\$	3,685,400.00	\$	3,671,800.00
6	1205	222	15	\$	171,551.00	\$	141,947.00
6	1240	224	16	\$	27,965,000.00	\$	27,869,539.00
6	1260	225	12	\$	5,000,000.00	\$	7,000.00
6	1265	225	24	\$	125,000.00	\$	10,016.00
6	1285	226	23	\$	36,000,000.00	\$	5,000,000.00
6	1285	226	27	\$	100,798,600.00	\$	80,729,100.00
6	1290	227	1	\$	5,980,800.00	\$	4,322,252.00
6	1360	230	31	\$	1,500,000.00	\$	68,620.00
6	1370	231	15	\$	1,500,000.00	\$	270,819.00
6	1390	232	18	\$	9,525,000.00	\$	9,427,500.00
6	1395	232	30	\$	4,090,987.00	\$	3,441,681.00
6	1410	233	29	\$	183,000.00	\$	182,151.00
Article	Section	Page	Lines	Amou	nt Enacted	Amoun	t Reduced
6	1420	234	25	\$	10,100,000.00	\$	9,729,332.00
6	1440	235	31	\$	7,275,000.00	\$	7,262,500.00
6	1485	238	19	\$	48,000.00	\$	23,503.00
6	1490	238	27	\$	600,000.00	\$	300,000.00

6	1550	241	31	\$ 1,000,000.00	\$ 500,000.00
6	1555	242	8	\$ 400,000.00	\$ 200,000.00
6	1585	243	28	\$ 1.750.000.00	\$ 1.077.765.00

In addition to these specific item and reduction vetoes, I hereby approve all other appropriation items in Senate Bill 1239.

Sincerely, ROD R. BLAGOJEVICH Governor

### MESSAGE FROM THE PRESIDENT

# OFFICE OF THE SENATE PRESIDENT STATE OF ILLINOIS

EMIL JONE, JR. Senate President

327 State Capitol Springfield, Illinois 62706

August 14, 2003

Ms. Linda Hawker Secretary of the Senate Room 403, State House Springfield, IL 62706

Dear Madam Secretary:

Pursuant to Senate Rule 3-3 (a), please be advised that effective today I am creating a special committee that will be called the "Senate Select Committee on Public Pension Investments." The committee will consist of seven members. The four majority caucus members are:

Senator Antonio Munoz, Chairman Senator Denny Jacobs Senator Barack Obama Senator Iris Martinez

> Sincerely, Emil Jones, Jr. Senate President

cc: Senate Minority Leader Frank Watson House Speaker Michael J. Madigan House Minority Leader Tom Cross

#### COMMUNICATION

ILLINOIS STATE SENATE FRANK C. WATSON STATE SENATOR 51<sup>ST</sup> SENATE DISTRICT

August 25, 2003

Ms. Linda Hawker Secretary of the Senate Room 403 State House [October 23, 2003] Springfield, IL 62706

Dear Madam Secretary:

Please be advised that pursuant to Senate Rule 3-3 (a), I am appointing the following members to the Senate Select Committee on Public Pension Investments.

Senator William Brady, Minority Spokesman Assistant Senate Republican Leader William Peterson Senator Christing Radogno

This appointment is effective immediately. If you have any questions, please contact Brian McFadden at 217-782-8184

Sincerely, Frank Watson Senate Republican Leader

cc: Senate President Emil Jones House Speaker Michael Madigan House Minority Leader Tom Cross Jim Harry, Asst. Secretary of the Senate Appointees

# OFFICE OF THE SECRETARY OF STATE **JESSE WHITE** – Secretary of State

September 10, 2003

Honorable Members Illinois State Senate 92<sup>nd</sup> General Assembly Springfield, IL 62706

Dear Members:

I am nominating Mr. Mike Masterson for re-appointment ot the Merit Commission for the office of Secretary of State.

I respectfully ask concurrence in and confirmation of this apointment by your Honorable Body:

COMMISSIONOR OF THE MERIT COMMISSIONS FOR THE OFFICE OF THE SECRETARY OF STATE.

To be Commissioner of the Merit Commission for the office of the Secretary of State for a term ending June 30, 2009.

Mike Masterson \$10,000 (Salary)

If you have any questions, please contact Dale Swinford, Director of Legislative Affairs. Thank you for your consideration.

Sincerely, JESSE WHITE Secretary of State

## MESSAGE FROM THE PRESIDENT

# OFFICE OF THE SENATE PRESIDENT STATE OF ILLINOIS

EMIL JONE, JR. Senate President

327 State Capitol Springfield, Illinois 62706

August 14, 2003

Ms. Linda Hawker Secretary of the Senate Room 403, State House Springfield, IL 62706

Dear Madam Secretary:

Pursuant to Rule 2-10, I am scheduling Wednesday, October 29, 2003 as a perfunctory session day.

Very truly yours,

Emil Jones, Jr. Senate President

cc: Senate Minority Leader Frank Watson
House Speaker Michael J. Madigan
House Minority Leader Tom Cross
All Senators
Joseph B. Handley, Deputy Chief of Staff for Legislative
Affairs, Governor's Office

#### MOTION IN WRITING

Senator Garrett submitted the following Motion in Writing:

I move to accept the specific recommendations of the Governor as to **Senate Bill 902** in manner and form as follows:

# AMENDMENT TO SENATE BILL 902 IN ACCEPTANCE OF GOVERNOR'S RECOMMENDATIONS

Amend Senate Bill 902 as follows:

on page 1, by replacing lines 18 through 26 with the following:

"shall provide that those schools that are on the State Board of Education Early Academic Warning List or the academic watch list under Section 2-3.25d that maintain grades kindergarten through 3 shall be given priority."; and

by replacing lines 30 and 31 on page 1 and lines 1 through 4 on page 2 with the following: "class size of no more than 20 pupils."

The foregoing Motion in writing was filed with the Secretary and referred to the Committee on Rules

#### MOTION WITHDRAWN

ILLINOIS STATE SENATE
WILLIAM E. PETERSON
STATE SENATOR – 26<sup>TH</sup> DISTRICT
ASSISTANT REPUBLICAN LEADER

October 6, 2003

Linda Hawker Secretary of the Senate 403 Statehouse Springfield, Illinois 62706

Dear Linda:

This letter is to advise you that I wish to withdraw my "Motion to Reconsider" regarding Senate Bill 1149. This measure would amend the Illinois Vehicle Code and I do not want to continue with my motion.

Please don't hesitate to contact my office if you need more information.

Thank you for your attention to this matter.

Sincerely,

William E. Peterson

### MESSAGES FROM THE HOUSE

A message from the House by

Mr. Rossi, Clerk:

Mr. President -- I am directed to inform the Senate that the House of Representatives has concurred with the Senate in the passage of a bill of the following title, to-wit:

SENATE BILL NO. 1126

A bill for AN ACT concerning clerks of courts.

Together with the following amendment which is attached, in the adoption of which I am instructed to ask the concurrence of the Senate, to-wit:

House Amendment No. 1 to SENATE BILL NO. 1126

Passed the House, as amended, May 31, 2003.

ANTHONY D. ROSSI, Clerk of the House

## AMENDMENT NO. 1 TO SENATE BILL 1126

AMENDMENT NO. 1. Amend Senate Bill 1126 as follows:

on page 1, line 8, by changing "Illinois Department of Revenue" to "clerk of the circuit court".

Under the rules, the foregoing **Senate Bill No. 1126**, with House Amendment No. 1 was referred to the Secretary's Desk.

A message from the House by

Mr. Rossi, Clerk:

Mr. President -- I am directed to inform the Senate that the House of Representatives has concurred with the Senate in the adoption of their amendment to a bill of the following title, to-wit:

HOUSE BILL NO. 719

A bill for AN ACT in relation to vehicles.

Which amendment is as follows:

Senate Amendment No. 1 to HOUSE BILL NO. 719

I am further directed to inform the Senate that the House of Representatives has refused to concur with the Senate in the adoption of the following amendment:

Senate Amendment No. 2 to HOUSE BILL NO. 719

Action taken by the House, May 31, 2003.

## ANTHONY D. ROSSI, Clerk of the House

Under the rules, the foregoing **House Bill No. 719**, with Senate Amendment No. 2 was referred to the Secretary's Desk.

A message from the House by

Mr. Rossi, Clerk:

Mr. President -- I am directed to inform the Senate that the House of Representatives has concurred with the Senate in the adoption of their amendment to a bill of the following title, to-wit:

HOUSE BILL NO. 741

A bill for AN ACT in relation to alcohol.

Which amendment is as follows:

Senate Amendment No. 1 to HOUSE BILL NO. 741

I am further directed to inform the Senate that the House of Representatives has refused to concur with the Senate in the adoption of the following amendment:

Senate Amendment No. 2 to HOUSE BILL NO. 741

Action taken by the House, May 31, 2003.

ANTHONY D. ROSSI, Clerk of the House

Under the rules, the foregoing **House Bill No. 741**, with Senate Amendment No. 2 was referred to the Secretary's Desk.

A message from the House by

Mr. Rossi, Clerk:

Mr. President -- I am directed to inform the Senate that the House of Representatives has adopted the following joint resolution, in the adoption of which I am instructed to ask the concurrence of the Senate, to-wit:

## HOUSE JOINT RESOLUTION NO. 9

WHEREAS, Throughout history brave Americans have shed their blood during wars and conflicts to preserve, protect, and defend the foundation of the principles of democracy and freedom, and hundreds of thousands have paid the ultimate sacrifice to ensure that future generations enjoy life's liberties; and

WHEREAS, On August 7, 1782, General George Washington established the Military Badge of Merit, which on February 22, 1932 became the present and now the oldest military decoration in the world, the Purple Heart medal; and

WHEREAS, The Purple Heart medal is awarded to all military personnel who are killed or wounded in action against the enemy; therefore, be it

RESOLVED, BY THE HOUSE OF REPRESENTATIVES OF THE NINETY-THIRD GENERAL ASSEMBLY OF THE STATE OF ILLINOIS, THE SENATE CONCURRING HEREIN, that Interstate 72, traversing through the heart of Illinois, be designated as the commemorative "Purple Heart Memorial Highway", to pay tribute to the many thousands of Illinois residents who have been awarded the Purple Heart medal; and be it further

RESOLVED, That the Illinois Department of Transportation is requested to erect at suitable locations, consistent with State and federal regulations, an appropriate plaque or signs giving notice of the name; and be it further

RESOLVED, That a suitable copy of this resolution be delivered to the Secretary of Transportation.

Adopted by the House, May 28, 2003.

ANTHONY D. ROSSI, Clerk of the House

The foregoing message from the House of Representatives, reporting House Joint Resolution No. 9, was referred to the Committee on Rules.

A message from the House by

Mr. Rossi, Clerk:

Mr. President -- I am directed to inform the Senate that the House of Representatives has adopted the following joint resolution, in the adoption of which I am instructed to ask the concurrence of the Senate, to-wit:

## HOUSE JOINT RESOLUTION NO. 15

WHEREAS, A displaced homemaker is an individual who has been providing unpaid services to family members in the home, has been dependent on the income of another family member but is no longer supported by that income, and is currently unemployed or underemployed and experiencing difficulty in obtaining or upgrading employment; and

WHEREAS, Displaced homemakers are often disproportionately poor when compared to the rest of the population; and

WHEREAS, Displaced homemakers must enter the workforce to support themselves and their families, but may face significant barriers to economic self-sufficiency due to a limited education, a lack of work experience, and age discrimination in the hiring process and the workplace; and

WHEREAS, Displaced homemakers may face severe housing problems because they are suddenly left with little or no income and have difficulty finding adequate employment, and because those who are awarded their homes in divorce settlements often cannot afford to maintain the mortgage payments and property taxes; and

WHEREAS, In fiscal year 2002, 1,758 displaced homemakers were enrolled in the Illinois Displaced Homemaker Program, administered by the Illinois Department of Labor, and 75% of these individuals were considered the sole financial support for children under the age of 18; and

WHEREAS, As states respond to federal welfare reform and workforce development legislation, it is likely that Illinois will experience changes in the types of services they currently provide; and

WHEREAS, It is important to continue responding to the needs of displaced homemakers and to expand services to reach more displaced homemakers by moving them out of poverty and beyond reliance on State welfare support and into gainful employment so they may become more active participants in the Illinois economy; therefore be it

RESOLVED, BY THE HOUSE OF REPRESENTATIVES OF THE NINETY-THIRD GENERAL ASSEMBLY OF THE STATE OF ILLINOIS, THE SENATE CONCURRING HEREIN, that a Task Force on Displaced Homemakers, comprised of members of the General Assembly, representatives of the Governor, and concerned members of the public, be created to investigate the services currently offered to displaced homemakers in Illinois and to make recommendations for improvements to those services; and be it further

RESOLVED, That 2 members who are elected members of the House of Representatives or the Senate be appointed by each of the following elected officials: the Speaker of the House, the Minority Leader of the House, the President of the Senate, and the Minority Leader of the Senate; and be it further

RESOLVED, That 2 members be appointed by the Governor to represent State agencies currently offering services and assistance to the displaced homemaker population in Illinois; and be it further

RESOLVED, That the members of the General Assembly appointed to the Task Force appoint 2 chairpersons, who shall be members of the Task Force who are members of the General Assembly, and be it further

RESOLVED, That meetings of the Task Force be held at the call of the chairpersons and that the chairpersons appoint 4 members of the public who have adequate expertise on the issue of displaced homemakers and the services and assistance provided to them; and be it further

RESOLVED, That all appointed members of the Task Force shall be considered to be members with voting rights, that a quorum of the Task Force shall consist of a simple majority of the members of the Task Force, and that all actions and recommendations of the Task Force be approved by a simple majority of the members of the Task Force; and be it further

RESOLVED, That members appointed to the Task Force shall be compensated only for costs that arise out of their official duties with the Task Force and that compensation shall come from an appropriate fund; and be it further

RESOLVED, That a suitable copy of this resolution be delivered to the Governor.

Adopted by the House, May 28, 2003.

ANTHONY D. ROSSI, Clerk of the House

The foregoing message from the House of Representatives, reporting House Joint Resolution No. 15, was referred to the Committee on Rules.

## JOINT ACTION MOTIONS FILED

The following Joint Action Motions to the Senate Bills listed below have been filed with the Secretary and referred to the Committee on Rules:

Motion to Concur in House Amendment 3 to Senate Bill 759 Motion to Concur in House Amendment 1 to Senate Bill 1126

## INTRODUCTION OF BILLS

SENATE BILL NO. 2090. Introduced by Senator Watson, a bill for AN ACT in relation to firearms.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

**SENATE BILL NO. 2091.** Introduced by Senator Welch, a bill for AN ACT concerning crematories.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

SENATE BILL NO. 2092. Introduced by Senator Radogno, a bill for AN ACT concerning municipalities.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules

**SENATE BILL NO. 2093.** Introduced by Senator Radogno, a bill for AN ACT regarding disabled persons.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

SENATE BILL NO. 2094. Introduced by Senator Soden, a bill for AN ACT concerning environmental protection.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

**SENATE BILL NO. 2095.** Introduced by Senators Shadid - Demuzio, a bill for AN ACT concerning taxes.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

SENATE BILL NO. 2096. Introduced by Senators Soden and Radogno, a bill for AN ACT regarding schools.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

SENATE BILL NO. 2097. Introduced by Senator Lauzen, a bill for AN ACT concerning vehicles.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

**SENATE BILL NO. 2098.** Introduced by Senator Sieben, a bill for AN ACT concerning natural resources.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

**SENATE BILL NO. 2099.** Introduced by Senators Watson - Bomke - Righter - J. Jones - Burzynski, Althoff, Brady, Cronin, Dillard, Lauzen, Luechtefeld, Peterson, Petka, Radogno, Risinger, Roskam, Rutherford, Sieben, Soden, D. Sullivan, Syverson, Winkel and Wojcik, a bill for AN ACT concerning finance.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

**SENATE BILL NO. 2100.** Introduced by Senators Wojcik - D. Sullivan - Althoff - Radogno - Burzynski, Cronin, Dillard, Lauzen, Peterson, Roskam, Soden and Syverson, a bill for AN ACT concerning finance.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

SENATE BILL NO. 2101. Introduced by Senators Luechtefeld - Rutherford - Bomke - Righter-Risinger, Watson, Althoff, Brady, Burzynski, Cronin, Dillard, Lauzen, Peterson, Petka, Radogno, Roskam, Sieben, Soden, D. Sullivan, Syverson, Winkel and Wojcik, a bill for AN ACT concerning taxes

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

**SENATE BILL NO. 2102.** Introduced by Senators Watson - J. Jones - Luechtefeld - Bomke - Brady, Burzynski, Cronin, Lauzen, Peterson, Petka, Radogno, Risinger, Roskam, Rutherford, Sieben, Soden, D. Sullivan, Syverson and Winkel, a bill for AN ACT in relation to taxation.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

**SENATE BILL NO. 2103.** Introduced by Senators Radogno - Syverson - Burzynski - Lauzen - Bomke, Watson, Brady, Cronin, Dillard, Geo-Karis, J. Jones, Peterson, Risinger, Roskam, Rutherford, Sieben, Soden, D. Sullivan and Winkel, a bill for AN ACT concerning State finances.

The bill was taken up, read by title a first time, ordered printed and referred to the Committee on Rules.

# PRESENTATION OF RESOLUTIONS

## SENATE RESOLUTION 198

Offered by Senator E. Jones and all Senators: Mourns the death of John Daniels of Chicago.

# **SENATE RESOLUTION 199**

Offered by Senator E. Jones and all Senators: Mourns the death of Dorothy Ruth Davis Curry of Chicago.

## **SENATE RESOLUTION 200**

Offered by Senator E. Jones and all Senators: Mourns the death of Golden Sylvester Tillman of Chicago.

## **SENATE RESOLUTION 201**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of Marshall Micheal Khayat of Wadsworth.

## **SENATE RESOLUTION 202**

Offered by Senator Shadid and all Senators:

Mourns the death of John E. "Jack" Cassidy, Jr. of Peoria.

### **SENATE RESOLUTION 203**

Offered by Senator Shadid and all Senators:

Mourns the death of Sam Stewart, Sr. of Gillespie.

### **SENATE RESOLUTION 204**

Offered by Senator Hunter and all Senators:

Mourns the death of Katherine Viola Bell of Chicago.

## **SENATE RESOLUTION 205**

Offered by Senator Shadid and all Senators:

Mourns the death of Melvin C. "Bud" Hasty of Peoria.

## **SENATE RESOLUTION 206**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of Vicki Smith of Beach Park.

# **SENATE RESOLUTION 207**

Offered by Senator Shadid and all Senators:

Mourns the death of Robert M. Walker of Peoria.

## **SENATE RESOLUTION 208**

Offered by Senator Dillard and all Senators:

Mourns the death of Carleton F. Nadelhoffer, Sr. of Naperville.

### **SENATE RESOLUTION 209**

Offered by Senator Clayborne and all Senators:

Mourns the death of Dominik Antwan Childress of East St. Louis.

## **SENATE RESOLUTION 210**

Offered by Senator Clayborne and all Senators:

Mourns the death of Bruce Hill, Jr. of East St. Louis.

# SENATE RESOLUTION 211

Offered by Senator Shadid and all Senators:

Mourns the death of Robert C. Schmitt of Kickapoo.

## **SENATE RESOLUTION 212**

Offered by Senator Clayborne and all Senators:

Mourns the death of R.C. Nelson of East St. Louis.

# **SENATE RESOLUTION 213**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of Robert Lewis Snook of Waukegan.

## **SENATE RESOLUTION 214**

Offered by Senator Risinger and all Senators:

Mourns the death of Robert E. "Bob" Miller of Galva.

## **SENATE RESOLUTION 215**

Offered by Senator Risinger and all Senators:

Mourns the death of Watts Carey Johnson of Princeville.

## **SENATE RESOLUTION 216**

Offered by Senator Risinger and all Senators:

Mourns the death of Gerald Witt of Walnut.

#### **SENATE RESOLUTION 217**

Offered by Senator Meeks and all Senators:

Mourns the death of Chester Milton De Graff of South Holland.

# **SENATE RESOLUTION 218**

Offered by Senator Clayborne and all Senators:

Mourns the death of Edward Terrell of East St. Louis.

### **SENATE RESOLUTION 219**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of Leland T. "Trig" Watson Jr. of Mundelein.

## **SENATE RESOLUTION 220**

Offered by Senator Dillard and all Senators:

Mourns the death of Donald Lawrence Moore of Naperville.

## SENATE RESOLUTION 221

Offered by Senator Dillard and all Senators:

Mourns the death of Arthur E. Giddings of Willowbrook.

### **SENATE RESOLUTION 222**

Offered by Senator Dillard and all Senators:

Mourns the death of Fred D. Kozarits of Westmont.

# **SENATE RESOLUTION 223**

Offered by Senator Dillard and all Senators:

Mourns the death of Robert A. Koranda of Naperville.

## **SENATE RESOLUTION 224**

Offered by Senator Hunter and all Senators:

Mourns the death of Mary Louise Albey McDowell of Chicago.

# SENATE RESOLUTION 225

Offered by Senator Petka and all Senators:

Mourns the death of First Lt. Timothy Ryan of Aurora.

## **SENATE RESOLUTION 226**

Offered by Senator Petka and all Senators:

Mourns the death of Ruth Brockway of Plainfield.

# **SENATE RESOLUTION 227**

Offered by Senator E. Jones and all Senators:

Mourns the death of Seymour Ronen of Northbrook.

# **SENATE RESOLUTION 228**

Offered by Senator Risinger and all Senators:

Mourns the death of Robert D. Talbot of Walnut.

# **SENATE RESOLUTION 229**

Offered by Senator Brady and all Senators:

Mourns the death of Minor Meyers Jr. of Bloomington.

### **SENATE RESOLUTION 230**

Offered by Senator Link and all Senators:

Mourns the death of Doris B. Bowlin of Sesser.

# **SENATE RESOLUTION 231**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of LeAnna Kay Fischer of Zion.

### SENATE RESOLUTION 232

Offered by Senator Maloney and all Senators: Mourns the death of Mary J. Lynch of Alsip.

### **SENATE RESOLUTION 233**

Offered by Senator Clayborne and all Senators: Mourns the death of Nelma Faye Brown Moody.

### **SENATE RESOLUTION 234**

Offered by Senator Haine and all Senators:

Mourns the death of Junior B. "Lefty" Ayres of Wood River.

## **SENATE RESOLUTION 235**

Offered by Senator Haine and all Senators:

Mourns the death of Buddy Wayne Davis of Wood River.

### **SENATE RESOLUTION 236**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of Helen Frances Luther formerly of Winthrop Harbor.

### **SENATE RESOLUTION 237**

Offered by Senator Shadid and all Senators:

Mourns the death of Kevin P. Schmitt of East Peoria.

## **SENATE RESOLUTION 238**

Offered by Senator Shadid and all Senators:

Mourns the death of Michael A. Perkins, Sr. of Peoria.

## **SENATE RESOLUTION 239**

Offered by Senator Link and all Senators:

Mourns the death of Joseph S. Semasko of North Chicago.

### **SENATE RESOLUTION 240**

Offered by Senator Link and all Senators:

Mourns the death of Bernard "Barney" Loeb of Vernon Hills.

# **SENATE RESOLUTION 241**

Offered by Senator Clayborne and all Senators:

Mourns the death of Jacqueline S. "Jacque" Davis of Columbia.

## **SENATE RESOLUTION 242**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of Walter T. Faro of Zion.

# **SENATE RESOLUTION 243**

Offered by Senator Geo-Karis and all Senators:

Mourns the death of Edo J. Belli of Long Grove.

# **SENATE RESOLUTION 244**

Offered by Senators Watson - Righter and all Senators:

Mourns the death of Spc. Brandon E. Miller formerly of Shelbyville.

## **SENATE RESOLUTION 245**

Offered by Senator Meeks and all Senators:

Mourns the death of Private First Class Brandon Ramsey of Chicago.

## **SENATE RESOLUTION 246**

Offered by Senator Dillard and all Senators:

Mourns the death of John E. "Bud" Widder of Naperville.

## **SENATE RESOLUTION 247**

Offered by Senator Dillard and all Senators:

Mourns the death of Robert Koranda.

# **SENATE RESOLUTION 248**

Offered by Senator Dillard and all Senators:

Mourns the death of Kristen M. Gough of Naperville.

## **SENATE RESOLUTION 249**

Offered by Senator Lightford and all Senators:

Mourns the death of Patrick O'Neal Glenn of Florida.

## **SENATE RESOLUTION 250**

Offered by Senator Link and all Senators:

Mourns the death of William Cramer of Mundelein.

## **SENATE RESOLUTION 251**

Offered by Senator Link and all Senators:

Mourns the death of Gail M. Svendsen of Grayslake.

### **SENATE RESOLUTION 252**

Offered by Senator Wojcik and all Senators:

Mourns the death of Prince Philipp-Ernst of Germany.

### **SENATE RESOLUTION 253**

Offered by Senator Clayborne and all Senators:

Mourns the death of Roy W. "Wesley" Kenney III of East St. Louis.

## **SENATE RESOLUTION 254**

Offered by Senator Hunter and all Senators:

Mourns the death of Rebecca Taylor of Chicago.

## **SENATE RESOLUTION 255**

Offered by Senator Shadid and all Senators:

Mourns the death of Mary Ellen Dougherty Ripper Schmitt of Peoria.

## **SENATE RESOLUTION 256**

Offered by Senator Clayborne and all Senators:

Mourns the death of Derrick Lewis of East St. Louis.

## **SENATE RESOLUTION 257**

Offered by Senator Clayborne and all Senators:

Mourns the death of Frank Clark of East St. Louis.

## **SENATE RESOLUTION 258**

Offered by Senator Lauzen and all Senators:

Mourns the death of James T. Wolf of Aurora.

## **SENATE RESOLUTION 259**

Offered by Senator Lauzen and all Senators:

Mourns the death of George Bennett Peters of Aurora.

# **SENATE RESOLUTION 260**

Offered by Senator Lauzen and all Senators:

Mourns the death of Arthur Boyd, Jr. of Batavia.

## **SENATE RESOLUTION 261**

Offered by Senator Lauzen and all Senators:

Mourns the death of Victor Frantz formerly of Oswego.

#### **SENATE RESOLUTION 262**

Offered by Senator Lauzen and all Senators:

Mourns the death of Raymond D. Heintz of Aurora.

# **SENATE RESOLUTION 263**

Offered by Senator Lauzen and all Senators:

Mourns the death of Janice H. "Ma" Yost of North Aurora.

### **SENATE RESOLUTION 264**

Offered by Senator Lauzen and all Senators:

Mourns the death of Catherine Williams of Oswego.

## SENATE RESOLUTION 265

Offered by Senator Lauzen and all Senators:

Mourns the death of Cathryn S., Rhoades of Aurora.

## **SENATE RESOLUTION 266**

Offered by Senator Lauzen and all Senators:

Mourns the death of Arthur S. Nelson of Geneva.

## **SENATE RESOLUTION 267**

Offered by Senator Lauzen and all Senators:

Mourns the death of Robert Koranda of Naperville.

### **SENATE RESOLUTION 268**

Offered by Senator Lauzen and all Senators: Mourns the death of Donna Diehl of Aurora.

# SENATE RESOLUTION 269

Offered by Senator Lauzen and all Senators:

Mourns the death of M. W. "Bud" Meyer of Aurora.

## SENATE RESOLUTION 270

Offered by Senator Lauzen and all Senators:

Mourns the death of William McKelvey Damisch.

## **SENATE RESOLUTION 271**

Offered by Senator Lauzen and all Senators:

Mourns the death of Sten G. Halfvarson of Aurora.

# **SENATE RESOLUTION 272**

Offered by Senator Schoenberg and all Senators:

Mourns the death of Kathryn "Kay" Katz of Wilmette.

#### SENATE RESOLUTION 273

Offered by Senator Roskam and all Senators:

Mourns the death of Bernadette "Bunny" Cline of Downers Grove.

By unanimous consent, the foregoing resolutions were referred to the Resolutions Consent Calendar.

Senator Watson offered the following Senate Resolution, which was referred to the Committee on Rules:

## **SENATE RESOLUTION NO. 274**

WHEREAS, Pharmacy is one of the oldest of the health professions concerned with the health and well-being of all people; and

WHEREAS, Today, there are over 195,000 pharmacists practicing in the United States providing services to assure the rational and safe use of all medications; and

WHEREAS, The use of medications, as a cost-effective alternative to more expensive medical procedures, is becoming a major force in moderating overall health care costs; and

WHEREAS, Today's powerful medications require greater attention to the manner in which they are used by different patient population groups, both clinically and demographically; and

WHEREAS, It is important that all users of prescriptions and non prescription medications, or their caregivers, be knowledgeable about and share responsibility for their own drug therapy; and

WHEREAS, The American Pharmacists Association, the Illinois Pharmacists Association, and the Joint Commission of Pharmacy Practitioners, have declared the final and complete week of October as National Pharmacy Week with the theme "Know Your Medicines-Know Your Pharmacist"; therefore, be it

RESOLVED, BY THE SENATE OF THE NINETY-THIRD GENERAL ASSEMBLY OF THE STATE OF ILLINOIS, that we proclaim October 19-25, 2003 as National Pharmacy Week and urge all citizens to acknowledge the valuable services of pharmacists to provide safe, affordable, and beneficial pharmaceutical care services and products to all citizens; and be it further

RESOLVED, That a suitable copy of this resolution be presented to the Illinois Pharmacist Association with our regards.

At the hour of 12:23 o'clock p.m., pursuant to the Message from the President dated October 23, 2003, the Chair announced the Senate stand adjourned until Wednesday, October 29, 2003, at 12:00 o'clock noon.